## Handbook of Research on Technological Developments for Cultural Heritage and eTourism Applications.

João M. F. Rodrigues, Célia M. Q. Ramos, Pedro J. S. Cardoso, Cláudia Henriques, Telmo Adão, Dora Agapito, Juan Miguel Alcántara-Pilar, Pau Alonso-Monasterio, Julia Altmman, Tanja Armenski, João Pedro Bernardes, Rui Carvalho, Sergio Casas, Pedro César, Marija Cimbaljevic, Marisol B. Correia, Carlos Costa, J. M. Cotos, Paula Odete Fernandes, Ana Maria Ferreira, Mauro Figueiredo, Marco Freitas, Alexandra Gonçalves y César Gonçalves.

### Cita:

João M. F. Rodrigues, Célia M. Q. Ramos, Pedro J. S. Cardoso, Cláudia Henriques, Telmo Adão, Dora Agapito, Juan Miguel Alcántara-Pilar, Pau Alonso-Monasterio, Julia Altmman, Tanja Armenski, João Pedro Bernardes, Rui Carvalho, Sergio Casas, Pedro César, Marija Cimbaljevic, Marisol B. Correia, Carlos Costa, J. M. Cotos, Paula Odete Fernandes, Ana Maria Ferreira, Mauro Figueiredo, Marco Freitas, Alexandra Gonçalves y César Gonçalves (2018). *Handbook of Research on Technological Developments for Cultural Heritage and eTourism Applications*. Hershey: IGI Global.

Dirección estable: https://www.aacademica.org/del.giorgio.solfa/136

ARK: https://n2t.net/ark:/13683/pa9s/upk



Esta obra está bajo una licencia de Creative Commons. Para ver una copia de esta licencia, visite https://creativecommons.org/licenses/by-sa/4.0/deed.es.

Acta Académica es un proyecto académico sin fines de lucro enmarcado en la iniciativa de acceso abierto. Acta Académica fue creado para facilitar a investigadores de todo el mundo el compartir su producción académica. Para crear un perfil gratuitamente o acceder a otros trabajos visite: https://www.aacademica.org.

# Handbook of Research on Technological Developments for Cultural Heritage and eTourism Applications

João M. F. Rodrigues University of Algarve, Portugal

Célia M. Q. Ramos University of Algarve, Portugal

Pedro J. S. Cardoso University of Algarve, Portugal

Cláudia Henriques University of Algarve, Portugal



Published in the United States of America by

IGI Global Business Science Reference (an imprint of IGI Global) 701 E. Chocolate Avenue Hershey PA, USA 17033

Tel: 717-533-8845 Fax: 717-533-8661

E-mail: cust@igi-global.com Web site: http://www.igi-global.com

Copyright © 2018 by IGI Global. All rights reserved. No part of this publication may be reproduced, stored or distributed in any form or by any means, electronic or mechanical, including photocopying, without written permission from the publisher. Product or company names used in this set are for identification purposes only. Inclusion of the names of the products or companies does not indicate a claim of ownership by IGI Global of the trademark or registered trademark.

Library of Congress Cataloging-in-Publication Data

Names: Rodrigues, Joao M. F., 1971-

Title: Handbook of research on technological developments for cultural heritage and eTourism applications / Joao M. F. Rodrigues, Celia M. Q. Ramos, Pedro J. S. Cardoso, and Claudia Henriques, Editors.

Description: Hershey PA: Business Science Reference, [2018]

Identifiers: LCCN 2017012413| ISBN 9781522529279 (hardcover) | ISBN

9781522529286 (ebook)

Subjects: LCSH: Tourism--Computer network resources. | Culture and

tourism--Computer network resources. | Heritage tourism--Computer network

resources

Classification: LCC G156.5.I5 .T43 2018 | DDC 910.285--dc23 LC record available at https://lccn.loc.gov/2017012413

This book is published in the IGI Global book series Advances in Hospitality, Tourism, and the Services Industry (AHTSI) (ISSN: 2475-6547; eISSN: 2475-6555)

British Cataloguing in Publication Data

A Cataloguing in Publication record for this book is available from the British Library.

All work contributed to this book is new, previously-unpublished material. The views expressed in this book are those of the authors, but not necessarily of the publisher.

For electronic access to this publication, please contact: eresources@igi-global.com.



### Advances in Hospitality, Tourism, and the Services Industry (AHTSI) Book Series

ISSN:2475-6547 EISSN:2475-6555

### Mission

Globally, the hospitality, travel, tourism, and services industries generate a significant percentage of revenue and represent a large portion of the business world. Even in tough economic times, these industries thrive as individuals continue to spend on leisure and recreation activities as well as services.

The Advances in Hospitality, Tourism, and the Services Industry (AHTSI) book series offers diverse publications relating to the management, promotion, and profitability of the leisure, recreation, and services industries. Highlighting current research pertaining to various topics within the realm of hospitality, travel, tourism, and services management, the titles found within the AHTSI book series are pertinent to the research and professional needs of managers, business practitioners, researchers, and upper-level students studying in the field.

### COVERAGE

- Destination Marketing and Management
- Tourism and the Environment
- Food and Beverage Management
- Cruise Marketing and Sales
- Sustainable Tourism
- Service Design
- Service Management
- Health and Wellness Tourism
- Leisure & Business Travel
- Casino Management

IGI Global is currently accepting manuscripts for publication within this series. To submit a proposal for a volume in this series, please contact our Acquisition Editors at Acquisitions@igi-global.com or visit: http://www.igi-global.com/publish/.

The Advances in Hospitality, Tourism, and the Services Industry (AHTSI) Book Series (ISSN 2475-6547) is published by IGI Global, 701 E. Chocolate Avenue, Hershey, PA 17033-1240, USA, www.igi-global.com. This series is composed of titles available for purchase individually; each title is edited to be contextually exclusive from any other title within the series. For pricing and ordering information please visit http://www.igi-global.com/book-series/advances-hospitality-tourism-services-industry/121014. Postmaster: Send all address changes to above address. Copyright © 2018 IGI Global. All rights, including translation in other languages reserved by the publisher. No part of this series may be reproduced or used in any form or by any means – graphics, electronic, or mechanical, including photocopying, recording, taping, or information and retrieval systems – without written permission from the publisher, except for non commercial, educational use, including classroom teaching purposes. The views expressed in this series are those of the authors, but not necessarily of IGI Global.

### Titles in this Series

For a list of additional titles in this series, please visit: www.igi-global.com/book-series

Tourism and Opportunities for Economic Development in Asia

Patricia Ordónez de Pablos (University of Oviedo, Spain) and Zeyar Myo Aung (Ministry of Hotels and Tourism, Myanmar)

Business Science Reference • copyright 2017 • 341pp • H/C (ISBN: 9781522520788) • US \$205.00 (our price)

Handbook of Research on Strategic Alliances and Value Co-Creation in the Service Industry

Shai Rozenes (Afeka Tel-Aviv Academic College of Engineering, Israel) and Yuval Cohen (Afeka Tel-Aviv Academic College of Engineering, Israel)

Business Science Reference • copyright 2017 • 507pp • H/C (ISBN: 9781522520849) • US \$235.00 (our price)

Driving Tourism through Creative Destinations and Activities

Alžbeta Kiráľová (University College of Business in Prague, Czech Republic)

Business Science Reference • copyright 2017 • 375pp • H/C (ISBN: 9781522520160) • US \$200.00 (our price)

Handbook of Research on Holistic Optimization Techniques in the Hospitality, Tourism, and Travel Industry Pandian Vasant (Universiti Teknologi PETRONAS, Malaysia) and Kalaivanthan M. (Rasa Sayang Resort and Spa, Malaysia & Golden Sands Resort, Malaysia)

Business Science Reference • copyright 2017 • 515pp • H/C (ISBN: 9781522510543) • US \$235.00 (our price)

Opportunities and Challenges for Tourism and Hospitality in the BRIC Nations

Mohinder Chand Dhiman (Kurukshetra University, India)

Business Science Reference • copyright 2017 • 426pp • H/C (ISBN: 9781522507086) • US \$205.00 (our price)

Strategic Place Branding Methodologies and Theory for Tourist Attraction

Ahmet Bayraktar (Bozok University, Turkey) and Can Uslay (Rutgers University, USA)

Business Science Reference • copyright 2017 • 394pp • H/C (ISBN: 9781522505792) • US \$205.00 (our price)

Global Place Branding Campaigns across Cities, Regions, and Nations

Ahmet Bayraktar (Bozok University, Turkey) and Can Uslay (Rutgers University, USA)

Business Science Reference • copyright 2017 • 368pp • H/C (ISBN: 9781522505761) • US \$205.00 (our price)

Global Dynamics in Travel, Tourism, and Hospitality

Nikolaos Pappas (The University of West London, UK) and Ilenia Bregoli (University of Lincoln, UK)

Business Science Reference • copyright 2016 • 372pp • H/C (ISBN: 9781522502012) • US \$205.00 (our price)



### Editorial Advisory Board

Jordi Tresserras Juan, University of Barcelona, Spain

João Albino Silva, University of Algarve, Portugal João Barroso, University of Trás-os-Montes e Alto Douro, Portugal Pedro Bittencourt César, University of Caxias do Sul, Brazil Federico Del Giorgio Solfa, National University of La Plata, Argentina Delia Garijo, University of Seville, Spain Beata Godejord, Nord University, Norway Mercedes Gonzalez Bracco, University of Buenos Aires, Argentina Ulrike Gretzel, University of Queensland, Australia Robert Lanquar, University of Paris 1 Panthéon-Sorbonne, France Luiz Mendes Filho, University Federal do Rio Grande do Norte, Brazil Francisco Muñoz-Leiva, University of Granada, Spain Toshiya Naka, Kyoto University, Japan Paula Odete Fernandes, Polytechnic Institute of Bragança, Portugal George A. Papadopoulos, *University of Cyprus*, *Cyprus* José Pérez Ballester, University of València, Spain Cristina Portales, University of València, Spain

### Foreword

Culture matters in sustainable economic development. A symbiotic relationship between culture and tourism should ensure that powerful drivers such as the economy, creativity, innovation, knowledge, take full advantage of the and information and communication technologies (ICTs). ICTs are established as a critical factor for sustainable competitiveness. Technology supports tangible and intangible cultural heritage to play an important role in enhancing authenticity and conservation efforts. It can reinforce engagement with prospective and actual visitors, support community values and identity as well as stakeholder participation.

A range of technologies have been used historically to support tourism (Buhalis, 2003; Egger & Buhalis, 2008; Buhalis & Law, 2008). Technology has supported consumers and the tourism demand, such as enabling tourists to plan visits to destinations more easily and allowing a progressive enrichment of tourists' experiences and knowledge. This has resulted in higher satisfaction levels among these travelers as well as in their engagement through social media and contribution to the image of the destination (Leung et al., 2013; Fotis et al., 2011; Williams et al., 2017; Hays et al., 2012; Neuhofer et al., 2014). ICTs have highlighted tourists, residents, and professionals' roles, enabling a better understanding of how their part in planning, managing, and marketing processes can produce top-quality, long-lasting tourism products.

ICT applications are widely used industry, tourism boards, and regional tourism associations. ICTs are also now applied in cultural and culture-related institutions (e.g., museums, libraries, and archives). This has resulted in significant outputs that have facilitated building the cultural identity and image of host countries, places, and attractions and have enhanced the communication of relevant content. The ever-increasing competition within the travel, tourism, cultural, and creative sectors has made it vital for destinations to address their target groups' needs by redefining marketing policies through these technologies.

Given the increased, widespread impact of new technologies on cultural tourism (Tscheu & Buhalis, 2016), this book seeks to offer a comprehensive discussion of research trends in the study of contemporary ICT systems and applications in cultural heritage and tourism. The chapters enclosed were written by international experts on the design, evaluation, implementation, and use of innovative technologies in this field. This volume's contents combine the different perspectives of cultural tourism management and technological uses in this industry and examine the role of ICTs in enhancing the value of heritage. Each chapter, therefore, connects technological developments with cultural heritage, linking and exploring the relationships between four main topics: (a) ICTs, (b) cultural heritage, (c) eTourism, and (d) tourism and assistive technologies and accessibility.

In this context, this book presents germane discussions exploring the use of ICTs such as augmented reality, multi-dimensional virtual objects, virtual reality, and a mobile system to accurately model and/or reconstruct and visualize real environments. Other applications covered are Wi-Fi information systems, building information modelling, mobile instant messaging, GISs, online community reviews, cloud computing, social media, recommendation systems, intelligent interfaces, and marketing intelligence. Further chapters discuss gamification, tangible user interfaces, 3D printing, beacons, multi-touch, and mobile and cloud computing.

Overall, this volume highlights how ICTs can be adapted to enhance cultural attractions or facilities, including not only monuments, archaeological sites, and museums but also cultural and gastronomic or cultural and religious routes, among other tourism attractions. To highlight the varied ways that cultural heritage can be interconnected with ICTs, many types of tourism are explored, namely, heritage, religious, dark, gastronomic, smart, and rural tourism. Finally, the enclosed chapters highlight the importance of companies such as hotels and restaurants to the travel and tourism industry, as well as emphasizing the role of management boards.

Cultural tourism is a growing tourism segment and experts now recognize its deep interconnection with ICTs. By enhancing the value of regional culture and history, these technologies help their preservation and dissemination, as well as promoting accessibility and reducing social and physical barriers.

Dimitrios Buhalis Bournemouth University, UK

Dimitrios Buhalis is Head of Department Tourism and Hospitality, Director of the eTourism Lab and Deputy Director of the International Centre for Tourism and Hospitality Research, at Bournemouth University in England. He is a Strategic Management and Marketing expert with specialisation in Information Communication Technology applications in the Tourism, Travel, Hospitality and Leisure industries. His research area is cutting across a number of disciplines and is looking into adoption of innovations for creating value for both consumers and organisations. Current research focus includes Smart Tourism, SoCoMo Marketing (Social media Context and Mobile Marketing), Augmented Reality, Experience management and Personalisation, Reputation and Social Media Strategies, Accessible and Special Diet (Alergens) Tourism. Close collaboration with industry, governments and international organisations from around the world ensures that results have relevance and impact in the real world whilst constantly pushing the boundaries of knowledge and academic excellence. Dimitrios is also the immediate past President of the International Federation for Information Technologies in Travel and Tourism (IFITT). Dimitrios is researching, lecturing, advising and facilitating progress in organisations globally, whilst his books and publications are studied widely by both students and professionals around the world. He is frequently delivering keynote presentations, speeches, seminars and workshops in both academic and professional conferences globally.

### REFERENCES

Boes, K., Buhalis, D., & Inversini, A. (2016). Smart tourism destinations: Ecosystems for tourism destination competitiveness. *International Journal of Tourism Cities*, 2(2), 108–124. doi:10.1108/IJTC-12-2015-0032

Buhalis, D. (2003). eTourism: Information technology for strategic tourism management. Pearson.

### **Foreword**

Buhalis, D., & Law, R. (2008). Progress in tourism management: Twenty years on and 10 years after the internet: The state of eTourism research. *Tourism Management*, 29(4), 609–623. doi:10.1016/j. tourman.2008.01.005

Egger, R., & Buhalis, D. (2008). *eTourism Case studies: Management & Marketing issues in eTourism*. Butterworth Heinemann Oxford.

Fotis, J., Buhalis, D., & Rossides, N. (2011). Social media impact on holiday travel: The case of the Russian and the FSU markets. *International Journal of Online Marketing*, *1*(4), 1–19. doi:10.4018/ijom.2011100101

Hays, S., Page, S., & Buhalis, D. (2012). Social media as a destination marketing tool: Its use by national tourism organisations. *Current Issues in Tourism*, 16(3), 211–239. doi:10.1080/13683500.2012.662215

Leung, D., Law, R., van Hoof, H., & Buhalis, D. (2013). Social Media in Tourism and Hospitality: A Literature Review. *Journal of Travel & Tourism Marketing*, 30(1-2), 1–2, 3–22. doi:10.1080/1054840 8.2013.750919

Neuhofer, B., Buhalis, D., & Ladkin, A. (2014). A Typology of Technology-Enhanced Tourism Experiences. *International Journal of Tourism Research*, *16*(4), 340–350. doi:10.1002/jtr.1958

Tscheu, F., & Buhalis, D. (2016). Augmented Reality at Cultural Heritage sites. In A. Inversini & R. Schegg (Eds.), *Information and Communication Technologies in Tourism 2016 ENTER* (pp. 607–619). Springer. doi:10.1007/978-3-319-28231-2\_44

Williams, N., Inversini, A., Buhalis, D., & Ferdinand, N. (2017). Destination eWOM drivers and characteristics. *Annals of Tourism Research*, *64*, 87–101. doi:10.1016/j.annals.2017.02.007

### **Preface**

### INTRODUCTION

Computational devices capable of connecting to the Internet, such as personal computers, smartphones or tablets, are currently available to almost everyone. Those computational devices are used in the most ordinary activities, including recreation, education or travelling, bringing information, culture and visiting experiences to a different level, while possibly helping in the accessibility improvement as well as in the info-exclusion mitigation. Upstream to the referred activities are the Information and Communications Technologies (ICT), often used as an extended synonym for Information Technologies. ICT is a more extensive term that stresses the role of unified communications, enterprise software, middleware, storage, and audio-visual systems, which enable users to access, store, transmit, and manipulate information.

On the other hand, touristic experiences, and in particular Cultural Tourism experiences, can only be understood in a dynamic and evolutionary framework, considering the complexity of the culture conceptualization as well as its role in a sustainable development. Therefore, culture can be considered as a key tourism resource, which contributes and inspires millions of tourists to travel and visit new destinations, several times a year. Cultural Tourism can be a considerable strength for the promotion and safeguarding of the tangible and intangible heritage it relies on, if it is managed and developed in a sustainable manner, while promoting and encouraging the development of creative activities associated to this heritage and the enhancement of the destination profile.

Marrying ICT and Cultural Tourism experiences concepts will enhance touristic destination profiles, providing innovation through the interconnection between culture, knowledge and technologies. The correct conjugation of those will also grant an enduring sustainable development. The profile enhancement creates a main drive for the development of a region where the authenticity, conservation, community, cultural values, and stakeholder's participation play the main role.

This reflection leads us to the Cultural Heritage perspective which should be understood as an expression of the ways of living developed by a community and passed on from generation to generation, including customs, practices, places, objects, artistic expressions, and values. Cultural Heritage is also often expressed as either tangible culture, which includes e.g., buildings, monuments, landscapes, books, works of art, and artefacts, or intangible culture, which includes e.g., folklore, dance, traditions, language, and knowledge. Nevertheless, it can also include natural heritage, such as culturally significant landscapes and biodiversity. In this context, the main assumption is that Cultural Heritage should consider the cultural values of the heritage, processes, community concerns, and administrative practices, conjugated in a way that contributes to a participation in socio-economic benefits for the stakeholders.

Therefore, considering the importance of culture in a sustainable development, it is relevant the role of ICT promoting a balanced approach between the needs, to potentiate the growth of cultural tourism market and better tourism experiences on one side, and the preservation of artifacts, historical sites, and local traditions on the other. A memorable cultural tourism experience should be based on uniqueness, authenticity and differentiation. Having in mind a growing emphasis on creativity, the basis to a more future-oriented creative tourism that values culture, contemporary creativity, technological platforms and contents, innovation, and networks is achievable, financing not only public but also private partnerships.

Associated with the Cultural Heritage experiences, is the eTourism ("electronic travel" or "electronic tourism") which can be defined as the analysis, design, implementation and application of IT and ecommerce solutions in the travel and tourism industry, as well as the analysis of the respective economic processes and market structures and customer relationship management.

Handbook of Research on Technological Developments for Cultural Heritage and eTourism Applications addresses the contemporary ICT technology applied to the disclosure of cultural and natural heritage, including tangible and intangible culture. In parallel, the book aims to reflect on the contemporary ICT relationship between tourism and history, based on the touristic use of the human cultural heritage. Likewise, the book addresses the interconnection with the called smart destinations, as the innovative tourist destination built on an infrastructure of state-of-the-art technology, which guarantees the sustainable development of tourist areas, facilitates the visitor's interaction with the surroundings, increasing the quality of the experience at the destination, but also in the improvements on residents' quality of life. The integration of this technology as special cases of smart cities, will also be an important point of interest.

Essentially, this book unites and explores the relations between three main topics: (a) Information and Communications Technology, (b) Cultural Heritage, and (c) eTourism. The objective of the book is to bring together a comprehensive collection of research trends on contemporary ICT systems and application in Cultural Heritage and Tourism from a set of international experts on the design, evaluation, implementation and use of innovative technologies on the field. In summary, the book emphasizes the connection between contemporary information and communications technologies with cultural heritage and tourism and eTourism.

Policy makers, academicians, researchers, advanced-level students, technology developers or simple curious, that are interested in the new trends in ICT systems and application to Cultural Heritage and Tourism are the target Audience. The book focuses on several different systems and applications, from mobile to "desktop" (three-dimensional) interfaces based on different natural interactions, to augmented reality solutions, and educational applications. Applications dedicate to different ways/types of accessibilities and applications where the Internet-of-Things is the interconnection of uniquely identifiable embedded computing devices, are also presented. All these, are centered in the main subject of Cultural Heritage and eTourism. All actors in this huge area will find this text useful in furthering their research exposure to pertinent topics and assisting in furthering their own research efforts in this field.

### ORGANIZATION OF BOOK

The book is organized as follows:

Chapter 1 proposes a methodological framework for the development of a recommendation system associated with the Religious Tourism Experience Model, with the objective of identifying and select-

ing consumer preferences for the dissemination of cultural heritage in general, and religious heritage in particular.

Chapter 2 discusses the actual panorama of interactive technologies used in museums exhibitions and how these institutions are designing digital installations and utilizing virtual media to enhance the visitors' experience.

In Chapter 3, the authors present a concept of augmented reality, the different aspects of it from other similar concepts, the infrastructure requirements, and analyze the applications examples in the tourism industry in the world especially within the content of destination marketing.

Chapter 4 reviews the state of the art of different techniques to deal with multi-dimensional acquisition of cultural heritage tangible assets. The authors also discuss some representation (e.g., visual, sonic, etc.) and interaction approaches to enhance the value of cultural heritage by means of the technologies of virtual reality and augmented reality, considering also the application of these techniques to tourism. Finally, they also point out some current needs and technological barriers that need to be tackled by the research community regarding the acquisition and dissemination of heritage.

Chapter 5 presents a framework to develop a mobile five human senses augmented reality museum system. This is to be used as a guide in cultural, historical and museum events, complementing or replacing the traditional orientation given by tour guides, directional signs, or maps. The system consists of a smartphone/phablet application and a hardware device to be integrated with smartphones/phablets, in order to explore the five human senses: sight, hearing, touch, smell, and taste.

Chapter 6 focuses on an Interactive Guide of the Portuguese Way of St. James (Camino de Santiago) created by using an Augmented Reality application. The Guide has the geo-information needed for the Pilgrim, in the passage by the municipality of Barcelos to Santiago, and can be accessed through a smartphone.

In Chapter 7, the authors begin with an introduction to the process of virtual reconstruction of archaeological sites, where the several stages that should take place to obtain a faithful virtual representation of an archaeological site and its artifacts are identified. Moreover, each stage is characterized and its main methods and techniques are identified. The chapter ends with the authors' vision about future trends for this field and unveils what could be their contributions to this vision.

Chapter 8 offers an in-depth analysis of archaeological sites and the value they represent to the field of cultural tourism. Specifically, it discusses the suitable means of studying and managing archaeological sites. In order to illustrate this line of research, this analysis will focus on one archaeological site in particular, namely the Archaeological Park of Segóbriga in the province of Cuenca, Spain.

In Chapter 9, the authors mention the problem associated to the accessibility of monuments and archaeological sites by disabled persons and persons with restricted mobility in general, which constitutes a social, financial and political demand. To overcome this problem, the authors presents the project "PROSPELASIS" to counter this challenge by focusing on creating a methodology for facing monuments' accessibility and perceptibility problems for people with disabilities and testing its application at Byzantine monuments of Thessaloniki.

Chapter 10 presents an approach for the preservation and exploration of the archaeological-historical information using the most modern methods of building information modelling together with virtual and augmented reality adapted to archaeology. This three-dimensional model is used to virtually reconstruct the Milreu archeological site, bringing this ancient Roman villa to life again, and supports the virtual reality platform and augmented reality for mobile devices.

Chapter 11 presents an essay involving the relations between the tourist appropriation of an historical architectural attraction – namely Jesuitic Mission of São Miguel – Brazil (legacy of the Guaraní Jesuit Missions - South America) – and the use of digital technology tools.

Chapter 12 discusses the information and communication technology potential to implement a smart approach on mass events. A known music festival was used as an exploratory evidence to support the main idea of the chapter, allowing to present some recommendations for mass event managers.

In Chapter 13, the authors present an exploratory study with the discussion of how creative experiences are being consumed through cultural mediation made possible by technological developments. In addition, the study is complemented with analysis of the creative tourism network website in the light of postmodern cultural mediation and capital construction.

Chapter 14 aims at measuring the image of the tourism brand "España Verde" by using the social media platform Facebook. The ever-increasing competition within the tourism industry makes it vital for destinations to address their target group by tailored marketing measures. Furthermore, social media channels are changing the interaction between Destination Management Organizations and customers. With the aim of exploring España Verde's image, a three components approach: cognitive, affective and conative was applied. Related to the measurement of the cognitive component, the study also focuses on España Verde's touristic potential ranging from a rich and diverse culture as well as a unique natural heritage to a suitable infrastructure for touristic purposes.

In Chapter 15, the cultural tourism consumers are analyzed in a way to understand their decision making process, when use the mobile instant messaging as a communication channel to choice a restaurant, in a social media platform. To achieve the objective was conducted a questionnaire to identify which channels are most used by consumers, when they intend to receive gastronomic recommendations. In the study, the conclusions identified that the channel use is determined by its instantaneousness and permanent information availability and due to a lack of knowledge about the area where the users is.

Chapter 16 analyzes the relationship between mycological tourism and Geographic Information Systems as a tool for tourism planning in rural areas. All this from a participatory approach in which traditional ecological knowledge serves as a source of information and a point of reference to guide policies for economic restructuring of rural areas.

Chapter 17 presents an essay that aims to explain the phenomenon and effect on tourism of the Balinese cemetery in the village of Trunyan, where the dead are not buried. It is a narrative enquiry combined with critical theory largely grounded in the scholarship of dark tourism and communication theory, coupled with content analysis of the online community's reviews from the TripAdvisor website.

In Chapter 18, the adoption of cloud computing in companies in general and in enterprises of the tourism sector in particular is addressed. The chapter includes a literature review to establish the conceptual framework of the technology and of the new economic model that underlies its adoption.

Chapter 19 describes some of the major issues around the cultural semantic web, including issues and range of choices institutions are facing to make data available on the web. Also introduced was Index Rerum, an inventory system based on an object-oriented model, as an example of managing collections and publishing data using the linked open data principles.

Last but not the least, for the book conclusion, one of the main topics that should always be in the center of any technological development and/or travel, the ecological footprint. In this case the minimization of energy consumption and the use of renewable energy. Chapter 20 describes the challenges involved in developing of a decision support platform to help hoteliers monitor their energy consumption, identify which points are consuming more than expected, decide which investments are more cost effective, and

manage their equipment in an optimum way. In this sense, the chapter covers several research fields, including internet of things networks, smart grids, renewable energy, and energy efficiency.

As a final point, it is important to reinforce, that the major contribution of the book, it is the "book as a whole". The "all" is bigger than the sum of the "parts", once the sum of the chapters created a book expected to interest to everyone that works or studies ICT applied to Cultural Heritage and Tourism. Nevertheless, each chapter brings to its respective sub-area the state of the art, showing applications, most of them not yet developed, except by the authors.

João M. F. Rodrigues University of Algarve, Portugal

Célia M. Q. Ramos University of Algarve, Portugal

Pedro J. S. Cardoso University of Algarve, Portugal

Cláudia Henriques University of Algarve, Portugal

### Acknowledgment

The editors would like to acknowledge the help of all the people involved in this project. We especially would like to express our gratitude to the chapter authors' and to the colleagues Ana Mafé Garcia, Ana Renda, António Cunha, Arsénio Reis, Cristiano Cabrita, Cristina Moreira, Delia Garijo, Fernando Perna, Francisco Munoz Lleiva, Gorete Dinis, Isabel Sofia Brito, Janet Hernandez, José Álvarez García, José Figueiredo Santos, Mª Mercedes Rojas Gracia, Muhammad Islam, Özgür Külcü, Pedro Garcês, Raul Guerreiro, Roberto Lam, Silvia Fernandes, Susana Gastal, Tuğçe Ozansoy Çadırcı, Vitor Filipe, Vitor Roque, who took part in the review process.

The editors would also like to acknowledge their research centers, LARSyS (Laboratory of Robotics and Engineering System – FCT: UID/EEA/50009/2013), ISR-Lisbon (Institute for Systems and Robotics, CIAC-UAlg (Center for Research in Communication Sciences and Arts), CEFAGE (Center for Advanced Studies in Management and Economics of the Universidade de Évora - CPEst-C/EGE/UI4007/2013) and CIEO - UAlg (Centre for Spatial and Organizational Dynamics).

### **Table of Contents**

| Foreword XX  | Κ1 |
|--|----|
| Preface xxi  | V  |
| Acknowledgment xxi   | X  |
| Chapter 1  |    |
| Religious Tourism Experience Model (RTEM): A Recommendation Model for Dissemination of       |    |
| Cultural and Religious Heritage  | 1  |
| Célia M. Q. Ramos, University of Algarve, Portugal   |    |
| Cláudia Henriques, University of Algarve, Portugal   |    |
| João M. F. Rodrigues, University of Algarve, Portugal  |    |
| Chapter 2  |    |
| Interactive Technologies in Museums: How Digital Installations and Media Are Enhancing the   |    |
| Visitors' Experience   | 0  |
| Roberto Ivo Fernandes Vaz, University of Porto, Portugal                                     |    |
| Paula Odete Fernandes, Polytechnic Institute of Bragança, Portugal                           |    |
| Ana Cecília Rocha Veiga, Federal University of Minas Gerais, Brazil                          |    |
| Chapter 3  |    |
| Augmented Reality: Applications and Implications for Tourism                                 | 4  |
| Erkan Özdemir, Uludağ University, Turkey   |    |
| Serkan Kılıç, Uludağ University, Turkey  |    |
| Chapter 4  |    |
| Multi-Dimensional Acquisition, Representation, and Interaction of Cultural Heritage Tangible |    |
| Assets: An Insight on Tourism Applications   | 2  |
| Cristina Portalés, Universitat de València, Spain  |    |
| Sergio Casas, Universitat de València, Spain   |    |
| Pau Alonso-Monasterio, Universitat Politècnica de València, Spain                            |    |
| María José Viñals, Universitat Politècnica de València                                       |    |

| Chapter 5   |
|---|
| An Initial Framework to Develop a Mobile Five Human Senses Augmented Reality System for         |
| Museums   |
| João M. F. Rodrigues, University of Algarve, Portugal   |
| Pedro J. S. Cardoso, University of Algarve, Portugal  |
| Joana Lessa, University of Algarve, Portugal  |
| João André Pereira, University of Algarve, Portugal   |
| João D. P. Sardo, University of Algarve, Portugal   |
| Marco de Freitas, University of Algarve, Portugal   |
| Jorge Semião, University of Algarve, Portugal   |
| Jânio Monteiro, University of Algarve, Portugal & INESC-ID, Portugal                            |
| Célia M. Q. Ramos, University of Algarve, Portugal  |
| Roberto Lam, University of Algarve, Portugal  |
| Eduardo Esteves, University of Algarve, Portugal & CCMAR Centre of Marine Sciences,<br>Portugal |
| Mauro Figueiredo, University of Algarve, Portugal   |
| Alexandra R. Gonçalves, University of Algarve, Portugal   |
| Miguel Gomes, SPIC – Creative Solutions, Portugal   |
| Paulo Bica, SPIC – Creative Solutions, Portugal   |
| Chapter 6   |
| Applying an Augmented Reality Tool to the Camino de Santiago in Portugal                        |
| Miguel Nuno Pereira, Minho University, Portugal   |
| Miguel Pazos Otón, University of Santiago de Compostela, Spain                                  |
| José Cotos, University of Santiago de Compostela, Spain   |
| Paula Cristina Remoaldo, Minho University, Portugal   |
| Tama Cristina Itemoarae, Illinio Cinversity, Fortugar   |
| Chapter 7   |
| Reconstructing the Past: Providing an Enhanced Perceptual Experience                            |
| Luís G. Magalhães, Centro Algoritmi, University of Minho, Portugal                              |
| Telmo Adão, University of Trás-os-Montes e Alto Douro, Portugal & INESC TEC, Portugal           |
| Emanuel Peres, University of Trás-os-Montes e Alto Douro, Portugal & INESC TEC,<br>Portugal     |
| Chapter 8   |
| <u>-</u>  |
| Assessing the Use of Archaeological Sites as Cultural Tourism Resources: The Case of Segóbriga  |
| (Spain)   |
| Nuria Huete-Alcocer, University of Castilla-La Mancha, Spain                                    |
| María Pilar Martínez-Ruiz, University of Castilla-La Mancha, Spain                              |
| Víctor Raúl López-Ruiz, University of Castilla-La Mancha, Spain                                 |
| Chapter 9   |
| Accessibility Improvement Interventions at Byzantine Monuments: Use of Technology for           |
| Facilitating Accessibility of Visitors With Sensory Disabilities                                |
| Aristotelis Naniopoulos, Aristotle University of Thessaloniki, Greece                           |
| Panagiotis Tsalis, Aristotle University of Thessaloniki, Greece                                 |
| Tanagions Isans, Arisione University of Thessatoliki, Greece                                    |

| Chapter 10   |     |
|--|-----|
| A Framework Supported by Modeling and Virtual/Augmented Reality for the Preservation a   | and |
| Dynamization of Archeological-Historical Sites   | 215 |
| Mauro Figueiredo, University of Algarve, Portugal  |     |
| João Pedro Bernardes, University of Algarve, Portugal                                    |     |
| José I. J. Rodrigues, University of Algarve, Portugal                                    |     |
| César Gonçalves, University of Algarve, Portugal   |     |
|  |     |
| Chapter 11   |     |
| Architecture, Tourism, and Technological Innovation in the Jesuit Mission of São Miguel, |     |
| Brazil   | 228 |
| Pedro de Alcântara Bittencourt César, Universidade de Caxias do Sul, Brazil              |     |
| Julia Luise Altmman, Universidade de Caxias do Sul, Brazil                               |     |
| Caroline Peccin da Silva, Universidade de Caxias do Sul, Brazil                          |     |
| Bruna Tronca, Universidade de Caxias do Sul, Brazil                                      |     |
| Michele Pinheiro Trentin, Universidade de Caxias do Sul, Brazil                          |     |
| Chapter 12   |     |
| Should Festival Be Smarter? ICT on Mass Events – The Case of the Exit Festival (Novi Sac | 1   |
| Serbia)  |     |
| •  | 243 |
| Uglješa Stankov, University of Novi Sad, Serbia  |     |
| Vanja Pavluković, University of Novi Sad, Serbia   |     |
| Juan Miguel Alcántara-Pilar, University of Granada, Spain                                |     |
| Marija Cimbaljević, University of Novi Sad, Serbia                                       |     |
| Tanja Armenski, University of Novi Sad, Serbia   |     |
| Chapter 13   |     |
| New Cultural Mediators, Cocreation, and the Cultural Consumption of Creative Tourism     |     |
|  | 264 |
| Experiences  | 264 |
| Rui Miguel Ferreira Carvalho, University of Aveiro, Portugal                             |     |
| Carlos Manuel Martins da Costa, University of Aveiro, Portugal                           |     |
| Ana Maria Alves Pedro Ferreira, University of Évora, Portugal                            |     |
| Chapter 14   |     |
| -  | 20/ |
| "España Verde": Tourism Destination Image among German Facebook Users                    | 204 |
| Till Maximilian Schuh, University of Algarve, Portugal                                   |     |
| Dora Agapito, University of Algarve, Portugal  |     |
| Patrícia Pinto, University of Algarve, Portugal  |     |
| Chapter 15   |     |
| Gastronomy in Cultural Tourism: Use of Mobile Instant Messaging to Look for Gastronomi   | c   |
| Recommendations  |     |
| Maria Romero-Charneco, Universidad de Málaga, Spain                                      |     |
|  |     |
| Ana-María Casado-Molina, Universidad de Málaga, Spain                                    |     |
| Pilar Alarcón-Urbistondo, Universidad de Málaga, Spain                                   |     |

| Chapter 16  |
|---|
| Information Mycological Systems and Traditional Ecological Knowledge: The Case of     |
| Mycological Tourism in Central Mexico   |
| Humberto Thomé-Ortiz, Universidad Autónoma del Estado de México, Mexico               |
| Chapter 17  |
| Isle of the Dead: A Study of Trunyan Cemetery (Bali)                                  |
| Bintang Handayani, President University, Indonesia                                    |
| Maximiliano Emanuel Korstanje, University of Palermo, Argentina                       |
| Chapter 18  |
| Perspectives of the Adoption of Cloud Computing in the Tourism Sector                 |
| Pedro R. Palos-Sanchez, University of Extremadura, Spain                              |
| Marisol B. Correia, ESGHT, University of Algarve, Portugal & CEG-IST, Universidade de |
| Lisboa, Portugal  |
| Chapter 19  |
| Semantic Transformation and Use of Cultural Data: Record Once and Publish Many401     |
| Feliz Ribeiro Gouveia, Universidade Fernando Pessoa, Portugal                         |
| Sérgio Lira, Green Lines Institute for Sustainable Development, Portugal              |
| Chapter 20  |
| A Platform for the Promotion of Energy Efficiency and Monitoring in Hotel Units       |
| Sergey Nogin, University of Algarve, Portugal   |
| Jânio Monteiro, University of Algarve, Portugal & INESC-ID, Portugal                  |
| Sergio Gómez Melgar, University of Huelva, Spain & LAR Arquitectura, Spain            |
| José Peyroteo, University of Algarve, Portugal  |
| António Mortal, University of Algarve, Portugal                                       |
| Carlos Miguel A. Santos, University of Algarve, Portugal                              |
| José Livramento, University of Algarve, Portugal                                      |
| Pedro J. S. Cardoso, University of Algarve, Portugal                                  |
| Jorge Semião, University of Algarve, Portugal   |
| Compilation of References   |
|   |
| About the Contributors  |
| Index   |

### **Detailed Table of Contents**

| Foreword   | XXi  |
|--|------|
| Preface  | xxiv |
| Acknowledgment   | xxix |
| Chapter 1  |      |
| Religious Tourism Experience Model (RTEM): A Recommendation Model for Dissemination of |      |
| Cultural and Religious Heritage  | 1    |
| Célia M. Q. Ramos, University of Algarve, Portugal                                     |      |
| Cláudia Henriques, University of Algarve, Portugal                                     |      |
| João M. F. Rodrigues, University of Algarve, Portugal                                  |      |

Information and Communication Technologies (ICT) have established a growing and deeper interconnection alliance with the tourism sector. Few solutions have contributed to the dissemination of cultural and religious heritage of the regions, as a means of promoting it, making known and enriching the culture of its visitors and residents. The development of an intelligent application that allows the dissemination of cultural and religious heritage can bring new experiences and sensations for tourists and for residents in general, and for those with accessibility problems in particular. The present chapter proposes a methodological framework for the development of a recommendation system associated with the Religious Tourism Experience Model (RTEM), with the objective of identifying and selecting consumer preferences for the dissemination of cultural heritage in general, and religious heritage in particular.

### Chapter 2

| Interactive Technologies in Museums: How Digital Installations and Media Are Enhancing the |    |
|--|----|
| Visitors' Experience   | 30 |
| Roberto Ivo Fernandes Vaz, University of Porto, Portugal                                   |    |
| Paula Odete Fernandes, Polytechnic Institute of Bragança, Portugal                         |    |
| Ana Cecília Rocha Veiga, Federal University of Minas Gerais, Brazil                        |    |
|  |    |

With the constant development of digital means of entertainment – that are easily made available to people and, in most cases, can be used anywhere – nowadays, a visit to a museum have to surround publics with unexpected and interactive experiences, in order to capture their attention and make them

want to go to these places, in addition to continue to communicate their collections and promote society education. In this regard, it was discussed in this chapter the actual panorama of interactive technologies used in museums exhibitions worldwide, and there are discussed how these institutions are designing digital installations and utilizing virtual media to enhance the visitors' experience, promoting positive relations between them and their publics. The main conclusion and reflection of the chapter is based on how this new era of technology is allowing increasing physical, cognitive and sensory accessibility, and transforms this kind of experience for disabled publics.

### Chapter 3

Technological advances have had an impact on many industries as well as the tourism industry. Augmented reality applications, one of the emerging new technologies in recent years, have also started to be used in our daily lives. Augmented Reality (AR) is a technology that allows its users to see the real world together with an additional virtual world that is added in real time to the same field of view. The augmented reality applications contribute to the enrichment of tourists' tourism experiences, especially during their visit and result in augmented satisfaction levels. Furthermore, it is one of the effective tools that can be used against the wear and tear of cultural heritage sites caused by overcrowding. In this chapter, the application fields of the augmented reality in the field of tourism have been discussed under the subtitles. As a result of our study, recommendations for the development of AR applications both for the literature and real-life application have been presented.

### Chapter 4

The acquisition, digital representation and interaction of cultural heritage (CH) assets is of high interest for an accurate documentation of our cultural legacy. Detailed studies demand more information on the assets apart from the mere visual appearance or even shape of objects. In this regards, some research works can be found that combine shape with other kind of data, such as radiometric (e.g., IR, thermal radiation, etc.) or semantic information, leading to multi-Dimensional (mD) virtual objects. In this chapter, we review the state of the art of different techniques to deal with mD acquisition of CH tangible assets. We also discuss some representation (e.g., visual, sonic, etc.) and interaction approaches to enhance the value of CH by means of the technologies of virtual reality (VR) and augmented reality (AR), considering also the application of these techniques to tourism. Finally, we also point out some current needs and technological barriers that need to be tackled by the research community regarding the acquisition and dissemination of heritage.

| Cha | pter | 5 |
|-----|------|---|
|-----|------|---|

| An Initial Framework to Develop a Mobile Five Human Senses Augmented Reality System for      |      |
|--|------|
| Museums  | . 96 |
| João M. F. Rodrigues, University of Algarve, Portugal  |      |
| Pedro J. S. Cardoso, University of Algarve, Portugal   |      |
| Joana Lessa, University of Algarve, Portugal   |      |
| João André Pereira, University of Algarve, Portugal  |      |
| João D. P. Sardo, University of Algarve, Portugal  |      |
| Marco de Freitas, University of Algarve, Portugal  |      |
| Jorge Semião, University of Algarve, Portugal  |      |
| Jânio Monteiro, University of Algarve, Portugal & INESC-ID, Portugal                         |      |
| Célia M. Q. Ramos, University of Algarve, Portugal   |      |
| Roberto Lam, University of Algarve, Portugal   |      |
| Eduardo Esteves, University of Algarve, Portugal & CCMAR Centre of Marine Sciences, Portugal |      |
| Mauro Figueiredo, University of Algarve, Portugal  |      |
| Alexandra R. Gonçalves, University of Algarve, Portugal                                      |      |
| Miguel Gomes, SPIC – Creative Solutions, Portugal  |      |
| Paulo Bica, SPIC – Creative Solutions, Portugal  |      |

The Mobile Five Senses Augmented Reality System for Museums (M5SAR) project aims to development an Augmented Reality mobile system for museums. Museums are amazing places, where it is important to sensorial augment as much as possible the visits, permitting to see, ear, touch, smell, and taste all the interesting objects there exist. Also fundamental is that visitors with different profiles (child, adult, expert, etc.) can have different experiences when visiting the same object. In the M5SAR system, the visitor uses its smartphone to select the object(s) to explore, and the user interface adapts on-the-fly to the object(s) and user's profile. Simultaneously, when integrated, a paired hardware device allows the extension of the augmented reality system to the human five senses, complementing the visual and auditory information about the objects. This chapter presents the initial framework to develop a five senses mobile adaptive museum system.

### Chapter 6

This chapter focuses on an Interactive Guide of the Portuguese Way of St. James (Camino de Santiago) created by using an Augmented Reality application. This application was developed specifically for the section of the Portuguese Route to Santiago that crosses the municipality of Barcelos (Northwest of Portugal). The Guide has the geo-information needed for the Pilgrim, in the passage by the municipality of Barcelos to Santiago, and can be accessed through a smartphone. The pilgrim will have at his or her disposal a set of interactive geo-information about Barcelos. At the level of interactivity with the user, each of these points of interest in the Guide will be signaled by a portrait done by a painter from Barcelos (monuments) and universal graphic icons (other geo-information).

### Chapter 7

Accurate modeling/reconstruction and visualization of real environments, particularly archaeological sites, is both a major challenge and a crucial task. This work will address the entire process of the virtual reconstruction of archaeological sites, since the construction of the virtual model until its visualization. The chapter begins with an introduction to the process of virtual reconstruction of archaeological sites, where the several stages that should take place to obtain a faithful virtual representation of an archaeological site and its artifacts are identified. Moreover, each stage is characterized and its main methods and techniques are identified, in dedicated sections. The authors' contribution for the state of the art will be highlighted in each stage. The chapter ends with the authors' vision about future trends for this field and unveils what could be their contributions to this vision.

### Chapter 8

Nuria Huete-Alcocer, University of Castilla-La Mancha, Spain María Pilar Martínez-Ruiz, University of Castilla-La Mancha, Spain Víctor Raúl López-Ruiz, University of Castilla-La Mancha, Spain

This chapter offers an in-depth analysis of archaeological sites and the value they represent to the field of cultural tourism. Specifically, this chapter discusses the suitable means of studying and managing archaeological sites, which represent tangible examples of cultural tourism. Managed properly, such sites may attract a larger number of visitors and thereby contribute to the region's socioeconomic development. In order to illustrate this line of research, this analysis will focus on one archaeological site in particular, namely the Archaeological Park of Segóbriga in the province of Cuenca, Spain. More specifically, the chapter reviews survey data collected over several years by related public institutions. The findings lend support to several measures that may improve the management of such sites in the context of cultural tourism. The most important of these is the dissemination and promotion of information about the site through WOM and eWOM.

### Chapter 9

The accessibility of monuments and archaeological sites by disabled persons and persons with restricted mobility in general, constitutes a social, financial and political demand. The project "PROSPELASIS" attempted to counter this problem by focusing on creating a methodology for facing monuments' accessibility and perceptibility problems for people with disabilities and testing its application at Byzantine monuments of Thessaloniki. In six major monuments included in UNESCO's World Heritage List,

significant accessibility improvements were realized. A Wi-Fi information system was installed in all of them via which an audio and visual information system was created with audio information in three languages, description of visual information, text information, information in Greek and International Sign Language. Additionally, in Rotunda and Heptapyrgion monuments three dimensional models were developed. The successful validation of the proposed methodology constitute the results of this project not only pertinent to Thessaloniki or Greece, but worthy of a wider application.

### Chapter 10

João Pedro Bernardes, University of Algarve, Portugal José I. J. Rodrigues, University of Algarve, Portugal César Gonçalves, University of Algarve, Portugal

This chapter presents an approach for the preservation and exploration of the archaeological-historical information using the most modern methods of Building Information Modelling (BIM) together with virtual and augmented reality adapted to archaeology. It was made a survey of the archaeological site using an unmanned aerial vehicle (UAV) and laser scan to obtain accurate information of existing structures which is stored in a BIM model. By using BIM methodologies, all existing information is organized in one place, shared and preserved for future memory. This 3D model is used to virtually reconstruct the Milreu archeological site, bringing this ancient Roman villa to life again, and supports the virtual reality platform and augmented reality tourism application for mobile devices, used for the promotion of the archaeological site, in an innovative way, in order to ensure a high degree of satisfaction to regular visitors and attract new public, looking for the visits to the archaeological-historical site to become a memorable experience.

### Chapter 11

Pedro de Alcântara Bittencourt César, Universidade de Caxias do Sul, Brazil Julia Luise Altmman, Universidade de Caxias do Sul, Brazil Caroline Peccin da Silva, Universidade de Caxias do Sul, Brazil Bruna Tronca, Universidade de Caxias do Sul, Brazil Michele Pinheiro Trentin, Universidade de Caxias do Sul, Brazil

The contemporaneity demands a better relation between the object and the subject. In this article, it is expected to understand how has been developed this bond of Jesuit architecture and its understanding by the visitor. In this way, it is searched to explore the touristic technological means in Missões Guaraníticas. It questions the use of technology to a better extent of dissemination of local and cultural legacy existing architecture. The methodology presents an exploratory research of technological platforms of information. It is noticed that its use increases tourism, favoring the culture and history of the region, besides promoting accessibility, reducing the social and physical barriers. The technology helps to preserve and disseminate the historical importance of the site. New information media has influenced and has increased the tourism, stimulating the local trade, and also it encourages the preservation of historical heritage. The approach of historical study of the region / country has great potential for increasing interest in historical architecture and culture.

| Cha | 4    | 1 | 1 |
|-----|------|---|---|
| Cha | pter | 1 | Z |

| Should Festival Be Smarter? ICT on Mass Events – The Case of the Exit Festival (Novi Sad, |     |
|---|-----|
| Serbia)   | 245 |
| Uglješa Stankov, University of Novi Sad, Serbia   |     |
| Vanja Pavluković, University of Novi Sad, Serbia  |     |
| Juan Miguel Alcántara-Pilar, University of Granada, Spain                                 |     |
| Marija Cimbaljević, University of Novi Sad, Serbia  |     |
| Tanja Armenski, University of Novi Sad, Serbia  |     |

The trend of information and communication technologies (ICT) employment to enhance transactions and to deliver better experience to visitors of mass events has been growing increasingly popular over the years. The emergence of "smart tourism" agenda which highlights new, more sustainable ways of business management, experience enhancement and destination management also creates new opportunities for ICT employment in mass event. Thus, this chapter discusses existing ICT holding potential for smart approach employment on mass events. An additional case of Exit festival was used as an exploratory evidence to support the main idea of the chapter. Based on comprehensive literary review and additional information on visitors' familiarity with ICT gained from the Exit festival, recommendations for mass event managers are presented.

### Chapter 13

| New Cultural Mediators, Cocreation, and the Cultural Consumption of Creative Tourism |     |
|--|-----|
| Experiences  | 264 |
| Rui Miguel Ferreira Carvalho, University of Aveiro, Portugal                         |     |
| Carlos Manuel Martins da Costa, University of Aveiro, Portugal                       |     |
| Ana Maria Alves Pedro Ferreira, University of Évora, Portugal                        |     |

Creative tourism has been a proficuous ground for the implementation of ICT's (Information and Communications Technology) strategies and the so-called creative industries. New cultural mediators are changing tourism consumption. Postmodern consumers have brought new perceptions to cocreation processes through user-generated content, eWOM (Electronic Word of Mouth), peer-to-peer exchange, collaborative economy, SoLoMo (social-local-mobile) tourists, among others. Booktubers and music fans/consumers in Chile constitute examples of diverse consumption through virtual and physical mediation, as capital construction enables social practices. In this exploratory study, the authors discuss how creative experiences are being consumed through cultural mediation made possible by technological developments. The creative tourism network website is analysed in the light of postmodern cultural mediation and capital construction.

### Chapter 14

This study aims at measuring the image of the tourism brand "España Verde" by using the social media platform Facebook. The ever-increasing competition within the tourism industry makes it vital for destinations to address their target group by tailored marketing measures. Furthermore social media channels are changing the interaction between Destination Management Organizations (DMOs) and

customers. With the aim of exploring España Verde's image, a three components approach—cognitive, affective and conative was applied. Related to the measurement of the cognitive component, the study also focuses on España Verde's touristic potential ranging from a rich and diverse culture as well as a unique natural heritage to a suitable infrastructure for touristic purposes. Although the supply side seems to fit the target group's desires when choosing a vacation destination, the exploratory results reveal that there is currently only little awareness for España Verde on the German market.

### Chapter 15

| Gastronomy in Cultural Tourism: Use of Mobile Instant Messaging to Look for Gastronomic |    |
|---|----|
| Recommendations   | 12 |
| Maria Romero-Charneco, Universidad de Málaga, Spain                                     |    |

Ana-María Casado-Molina, Universidad de Málaga, Spain Pilar Alarcón-Urbistondo, Universidad de Málaga, Spain

Cultural tourism consumers seek advice and recommendations from others before making buying decisions. The mobile instant messaging (MIM) boom could create a channel not only for social communication but also for information about cultural tourism and specifically the choice of restaurant. In this research we study which channels are most used by consumers when asking for gastronomic recommendations and, in particular, the factors behind the use of MIM as the preferred channels when obtaining tourist information. For this, an online survey was carried out with a sample of 393 individuals in Spain and descriptive statistical techniques were applied to the processing and analysis of the data. The results obtained show that: (1) its use is determined by its instantaneousness and permanent availability; and (2) people use it due to a lack of knowledge about the area.

### Chapter 16

| Information Mycological Systems and Traditional Ecological Knowledge: The Case of |    |
|---|----|
| Mycological Tourism in Central Mexico3  | 38 |
| Humberto Thomé-Ortiz, Universidad Autónoma del Estado de México, Mexico           |    |

Wild edible fungi are non-timber forest products that have great relevance for forest communities in central Mexico. Texcaltitlán is a rural community known for its traditional ecological knowledge on the use and identification of wild edible mushrooms. The aim of this work is to link Geographic Information Systems and Traditional Ecological Knowledge, in order to generate Mycological Information Systems. This is a qualitative, quantitative and exploratory research, which seeks to determine the usefulness of Geographic Information Systems (GIS) to systematize and locate mycological resources for use as a tourist attraction. The results show the existence of a wide variety of edible mushrooms in the region, along with a wide mycological traditional knowledge. Both aspects reflect the existence of unique natural and cultural features that can be the basis to build a unique tourism product in central Mexico. It is concluded that GIS are useful tools to build a multifunctional vision of mushrooms.

### Chapter 17

| Isle of the Dead: A Study of Trunyan Cemetery (Bali)            | 354 |
|---|-----|
| Bintang Handayani, President University, Indonesia              |     |
| Maximiliano Emanuel Korstanje, University of Palermo, Argentina |     |

This essay aims to explain the phenomenon and effect on tourism of the Balinese cemetery in the village of Trunyan, where the dead are not buried. It is a narrative enquiry combined with critical theory largely grounded in the scholarship of dark tourism and communication theory, coupled with content

analysis of the online community's reviews from the TripAdvisor website. The study indicates that (1) connectedness to death suggests the existence of spirituality needed by people, at the same time indicating understanding of mortality; (2) social connections developed as a result of visiting Trunyan cemetery not only bring self-awareness and awareness of others, revolving around intrapersonal communication about spirituality and interpersonal communication among members of the online community, but also illustrate the development of dark tourism and conceptualise the role of tourists in building authentic experience as the essence of a death site's brand image.

### Chapter 18

This chapter aims to expose the current situation of the adoption of cloud computing in companies in general and in enterprises of the tourism sector in particular. For this, a review of the literature has been carried out to establish the conceptual framework of technology and of the new economic model that underlies its adoption. Cloud computing is one of the technologies less known to many organizations and especially users, as it is a new technology based on the Internet, through which information is stored on servers, is provided as a service and on clients' demand. Therefore, the main theories of adoption that have been used to explain the different adoption models are presented, as well as, the different solutions that are being used in the tourist industry.

### Chapter 19

Cultural content and cultural assets are increasingly seen as invaluable drivers for education, entertainment, tourism, knowledge, and shared memory initiatives. As such, cultural information, as stored in cultural institutions such as museums, archives, and specialized libraries is increasingly demanded by several actors, from public policy, tourism, app developers and software companies, schools and the citizen at large. To satisfy the needs of such a large spectrum of consumers and end users, the information must be format independent, multi-lingual, multi-purpose, and make use of standards, norms and national and international recommendations. In this is paper we describe our past and current efforts to develop cultural information systems that can satisfy those requirements and those of the specialized public, such as scientists and curators. We describe an inventory and content management system and the conceptual and architectural choices that were made to allow its evolution, standards' compliance and multi-purpose use.

| Chapter | <b>20</b> |
|---------|-----------|
|---------|-----------|

| A Platform for the Promotion of Energy Efficiency and Monitoring in Hotel Units | 420 |
|---|-----|
| Sergey Nogin, University of Algarve, Portugal                                   |     |
| Jânio Monteiro, University of Algarve, Portugal & INESC-ID, Portugal            |     |
| Sergio Gómez Melgar, University of Huelva, Spain & LAR Arquitectura, Spain      |     |
| José Peyroteo, University of Algarve, Portugal                                  |     |
| António Mortal, University of Algarve, Portugal                                 |     |
| Carlos Miguel A. Santos, University of Algarve, Portugal                        |     |
| José Livramento, University of Algarve, Portugal                                |     |
| Pedro J. S. Cardoso, University of Algarve, Portugal                            |     |
| Jorge Semião, University of Algarve, Portugal                                   |     |

Tourists today are more likely to be concerned with the environment and greener lifestyle choices. In this context, a green flagship of some hotels can be an important selection criteria at the time of selecting one. In the near future, buildings should become nearly zero energy, consuming as low as possible and producing almost all the energy they need, using renewable energy sources. To achieve this goal, hotel buildings need to pass through a transformation process that will make them more efficient. In this process, a decision support platform would be important to help hoteliers monitor their energy consumption, identify which points are consuming more than expected, decide which investments are more cost effective and manage their equipment in an optimum way. This chapter describes the challenges involved in developing such a platform, covering several research and development fields, including Internet of Things networks, ICT, Smart Grids, Renewable Energy, Energy efficiency, as well as algorithms for machine learning and optimization.

| Compilation of References | . 449 |
|---------------------------|-------|
| About the Contributors    | . 520 |
| Index                     | . 532 |

### Chapter 1

### Religious Tourism Experience Model (RTEM):

### A Recommendation Model for Dissemination of Cultural and Religious Heritage

**Célia M. Q. Ramos** University of Algarve, Portugal

Cláudia Henriques University of Algarve, Portugal

João M. F. Rodrigues University of Algarve, Portugal

### **ABSTRACT**

Information and Communication Technologies (ICT) have established a growing and deeper interconnection alliance with the tourism sector. Few solutions have contributed to the dissemination of cultural and religious heritage of the regions, as a means of promoting it, making known and enriching the culture of its visitors and residents. The development of an intelligent application that allows the dissemination of cultural and religious heritage can bring new experiences and sensations for tourists and for residents in general, and for those with accessibility problems in particular. The present chapter proposes a methodological framework for the development of a recommendation system associated with the Religious Tourism Experience Model (RTEM), with the objective of identifying and selecting consumer preferences for the dissemination of cultural heritage in general, and religious heritage in particular.

DOI: 10.4018/978-1-5225-2927-9.ch001

### INTRODUCTION

At present, cultural consumption has become increasingly important, constituting a way of experiencing tourist experiences that are different from the daily routine of individuals (Urry, 2002). Richards (1996) and Russo and Borg (2002), have shown that cultural tourism is the main source of income for tourism (Urry, 2002).

Al Subhi, Bell and Lashmar (2015) point out that technologies are needed to enhance / increase the experiences of visitors so as to increase revenues and increase audiences (new visitors). They further note that the cultural heritage sector recognizes the value of presenting different content and different styles of personalization for different types of people (Falk, 2009), and that services can be customized to the different needs and preferences of children, parents and teachers.

In addition, Al Subhi et al. (2015) note that the existing literature on heritage and mobile technology is virtually focused on museums, and that there is a large gap in the use of devices, furniture and technology in large heritage sites such as monuments. Most studies have been carried out on tourist towns, and little attention has been given to integrated solutions for heritage sites and technology that enhance the socio-economic benefits of regions and cities. Lastly, it is beneficial to consider user-centered solutions that allow for a substantial collection of information by non-professional users, in order to create experiences and different values for visitors seeking a cultural aspect.

The structuring of cultural (religious) products for tourist purposes should be based on knowledge of trends in consumer behavior and major tourist market niches. At the same time, it should be "to facilitate knowledge and assist in co-designing sustainable tourism development" (Center for Tourism, Innovation and Culture [TIC], 2015).

The acquisition of knowledge of religious heritage by tourists, residents, and in particular those who have access difficulties (i.e., physical and mental disabilities) is enhanced when it is associated with technological innovations, especially once it is possible to offer new experiences and sensations to all visitors, including ones who have accessibility problems. One of the most important features is the possibility for a tourist to have access to the same information, regardless of their physical limitations. Another relevant potentiality is to reduce the negative impacts of tourism by reducing overcrowding of visitors attempting to access threatened or protected places.

ICT is excellent for contributing to the communication and acquisition of the cultural heritage of a region (Inversini & Cantoni, 2011) and for allowing the development of innovative solutions that meet the interests of economic and touristic partners. In parallel (Falk, 2009), the cultural heritage sector recognizes the value of presenting different content and different styles for different types of people where services can be tailored to the different needs and preferences of children, parents and teachers.

An innovative solution to meet this need is the development of a pioneering mobile smart application to disseminate tourism, cultural and religious heritage content in order to improve the tourist experience, as well as residents' learning.

The development of applications for mobile devices applied to the tourism sector is not new, since there are several prototypes developed in the academic world that combine tourism with Artificial Intelligence (AI) in a recommendation system to the tourist.

In this chapter the authors proposes a framework for the development of a recommendation system in order to identify and select consumer preferences, which can be classified according to the target market, forms of recommendation, use of knowledge or algorithms implementation, with the aim of 27 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/religious-tourism-experience-modelrtem/192758?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

### Related Content

How Do Online Reviews Affect Business Travelers' Accommodation Choices?: The Application of Theory of Reasoned Action (TRA)

Faranak Memarzadeh, Shane Blum and Charlie Adams (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 50-65).* 

www.igi-global.com/article/how-do-online-reviews-affect-business-travelers-accommodation-choices/177122?camid=4v1a

### Key Planning Factors for Deploying Ubiquitous Wi-Fi Networks in Hotels

Galen R. Collins (2017). *International Journal of Tourism and Hospitality Management in the Digital Age* (pp. 18-31).

www.igi-global.com/article/key-planning-factors-for-deploying-ubiquitous-wi-fi-networks-in-hotels/189742?camid=4v1a

### Critical Review of Tourism in India

Neeta Baporikar (2017). *Tourism and Opportunities for Economic Development in Asia (pp. 149-169).* www.igi-global.com/chapter/critical-review-of-tourism-in-india/176296?camid=4v1a

### Framework Oriented Approach to Eco-Tourism

Sanjay Mohapatra and Tripti Naswa (2015). *Hospitality, Travel, and Tourism: Concepts, Methodologies, Tools, and Applications* (pp. 1428-1440).

www.igi-global.com/chapter/framework-oriented-approach-to-eco-tourism/119290?camid=4v1a

## Chapter 2 Interactive Technologies in Museums: How Digital Installations and Media Are Enhancing the Visitors' Experience

Roberto Ivo Fernandes Vaz University of Porto, Portugal

Paula Odete Fernandes
Polytechnic Institute of Bragança, Portugal

Ana Cecília Rocha Veiga Federal University of Minas Gerais, Brazil

### **ABSTRACT**

With the constant development of digital means of entertainment – that are easily made available to people and, in most cases, can be used anywhere – nowadays, a visit to a museum have to surround publics with unexpected and interactive experiences, in order to capture their attention and make them want to go to these places, in addition to continue to communicate their collections and promote society education. In this regard, it was discussed in this chapter the actual panorama of interactive technologies used in museums exhibitions worldwide, and there are discussed how these institutions are designing digital installations and utilizing virtual media to enhance the visitors' experience, promoting positive relations between them and their publics. The main conclusion and reflection of the chapter is based on how this new era of technology is allowing increasing physical, cognitive and sensory accessibility, and transforms this kind of experience for disabled publics.

DOI: 10.4018/978-1-5225-2927-9.ch002

### INTRODUCTION

Because one lives in an era where people spend the most part of their time in front of screens and connected to the internet, museum professionals have to think about how to make collections and exhibitions still relevant nowadays, and how to compete with all the available entertainment ways and technological developments, in order to capture the visitor's attention and encourage their visit to these spaces.

Once this chapter focuses on exploiting digital ways of interactivity in museums, it is crucial, first of all, to understand their role in contemporary: "A Museum is a non-profit, permanent institution in the service of society and its development, open to the public, which acquires, conserves, researches, communicates and exhibits the tangible and intangible heritage of humanity and its environment for the purposes of education, study and enjoyment" (Desvallées & Mairesse, 2010, p. 57). Based on this professional definition, widely recognized and disseminated by ICOM – International Council of Museums – since 2007, it is clear that museums must provide to society the access to education, offering moments of study and leisure, in order to cooperate for its development, committing itself to safeguard and communicate their collections to the public (Desvallées & Mairesse, 2010).

As institutions are increasingly interested in providing outstanding informal learning experiences, there has been an effort to meet the visitor's aspirations, through the adaptation of innovative technology to communicate and encourage the visit to these spaces (Chelini, 2012). As pointed out by Sree Sreenivasan – the Metropolitan Museum of Art's chief digital officer, until June of 2016 – on an interview to Gilbert (2016), museums no longer need to compete with each other, because they are losing their visitors to the omnipresent technologies, games and social media consumed by the modern society; instead, those institutions have to find out ways to embrace the fact that smartphones, tablets, smartwatches and other digital devices are everywhere, and take advantage of the fact that people use them no matter when or where: "People ask me what our biggest competition is (...) It's not the Guggenheim; it's not the Museum of Natural History. It's Netflix. It's Candy Crush" (p. 32) and most recently, Pokémon Go (McCluskey, 2016). Taking these facts into account, it is self-explained why museums are committed, more than ever, in providing new scenarios of interaction and contact with their exhibitions.

On the other hand, it is important to make clear that the use of digital resources would not even be an option to communicate and to interact with the exhibitions if their use offers no contribution to what is being presented: in some cases, technological solutions are seen as essential for facilitating the reproduction of stories and intangible processes, like sound recordings, music, videos and other non-physical heritage; in other instances, they contribute to expand information about what cannot be exposed, like no longer existing artefacts and places, giving access to reconstructions and digital replicas (Chanda, 2013).

About the exhibition of physical samples, the use of digital resources proves to be beneficial because it enables visitors to have a better understanding of objects and ideas – since it allows offering complementary information in a more attractive way for visitors, without overloading the exhibitions environment with excess of information – and allows to increase accessibility and enable opportunities for disabled persons (Freeman et al., 2016; Israel, 2011).

However, professionals should be aware that technology must be used as a medium and not as an end, according to Israel (2011), McMullan (2015) and Olesen (2016), where, ideally, the museum themes should be harmoniously combined with digital content to communicate effectively the exhibitions and, at the same time, provide an incredible experience to visitors while using them: "given the increased emphasis on user experience, it is no longer sufficient to ensure that a system is merely usable" (O'Brien & Toms, 2008, p. 939).

22 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/interactive-technologies-inmuseums/192759?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

### Related Content

### Social Network Sites (SNS) and Digital Culture: Developing the Online Strategy of the Panama Viejo Museum

Ana Luisa Sánchez Laws (2011). Digital Culture and E-Tourism: Technologies, Applications and Management Approaches (pp. 1-9).

www.igi-global.com/chapter/social-network-sites-sns-digital/49616?camid=4v1a

### Key Planning Factors for Deploying Ubiquitous Wi-Fi Networks in Hotels

Galen R. Collins (2017). *International Journal of Tourism and Hospitality Management in the Digital Age* (pp. 18-31).

www.igi-global.com/article/key-planning-factors-for-deploying-ubiquitous-wi-fi-networks-in-hotels/189742?camid=4v1a

### Building Visual Travel Recommender Systems and Tourism Communities for Effective User Experience

Nalin Sharda (2010). Tourism Informatics: Visual Travel Recommender Systems, Social Communities, and User Interface Design (pp. 1-22).

www.igi-global.com/chapter/building-visual-travel-recommender-systems/37756?camid=4v1a

### Studying the Impact of Egyptian Hotels' Websites Marketing on Customers' E-Satisfaction

Nancy Awad and Sherif Saad Soliman (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 53-63).* 

www.igi-global.com/article/studying-the-impact-of-egyptian-hotels-websites-marketing-on-customers-e-satisfaction/189745?camid=4v1a

## Chapter 3 Augmented Reality: Applications and Implications for Tourism

Erkan Özdemir Uludağ University, Turkey

**Serkan Kılıç** Uludağ University, Turkey

### **ABSTRACT**

Technological advances have had an impact on many industries as well as the tourism industry. Augmented reality applications, one of the emerging new technologies in recent years, have also started to be used in our daily lives. Augmented Reality (AR) is a technology that allows its users to see the real world together with an additional virtual world that is added in real time to the same field of view. The augmented reality applications contribute to the enrichment of tourists' tourism experiences, especially during their visit and result in augmented satisfaction levels. Furthermore, it is one of the effective tools that can be used against the wear and tear of cultural heritage sites caused by overcrowding. In this chapter, the application fields of the augmented reality in the field of tourism have been discussed under the subtitles. As a result of our study, recommendations for the development of AR applications both for the literature and real-life application have been presented.

### INTRODUCTION

Today, as it is the case in many industries, there has been an intense competition in the tourism industry as well. In the face of this competition, the organizations in the tourism industry have used new technologies to stay competitive and develop marketing campaigns in various forms (Jung & Han, 2014). One of the techniques developed in this challenging competition is Augmented Reality (AR) applications. The AR technology is a new technology that can offer many opportunities that are difficult to be presented and met by the other technologies (Alkhamisi & Monowar, 2013).

In recent years, there have been many augmented reality applications used in different industries in the world. The augmented reality applications are used in many fields such as industry and military,

DOI: 10.4018/978-1-5225-2927-9.ch003

### Augmented Reality

training and education, travel and tourism, medicine and health care, and retail and marketing (Martínez, Skournetou, Hyppölä, Laukkanen, & Heikkilä, 2014). The tourism industry is also one of the industries where augmented reality applications have a comprehensive area of implementation. The tourism industry is an area in which especially the AR applications have higher potential of use. The availability of the AR applications, especially in mobile devices, causes these applications in the tourism industry to become even more widespread.

Today, the question of protecting cultural heritage has become very important. In addition to natural degradation, due to the negative effects caused by development activities and overcrowding, inheritance conservation efforts are vital in ensuring the sustainability of heritage sites. Just at this point, the advances in digital technologies such as virtual reality (VR) and augmented reality (AR) are the applications that can be valuable in preserving the heritage. These technologies will minimize the impact of overcrowding in the threatened heritage sites, but at the same time they will increase the overall experience of the visitors (Ab Aziz & Siang, 2014). The purpose of this kind of synergy between reality and virtual is to develop and consolidate the place-based economic development, culture, quality and the integration of local resources by preserving their historical identities. The tourism products are developed to include more cultural offerings among the various participating institutional actors, with a balanced dose of competition, co-operation and coordination (Garau, 2014). Augmented reality applications in the tourism industry are dealt with within the scope of smart tourism (Buhalis & Amaranggana, 2013; Koo, Shin, Kim, Kim, & Chung, 2013).

The aim of this chapter is to explicate the concept of augmented reality, the different aspects of it from other similar concepts, the infrastructure requirements, and analyze the applications examples in the tourism industry in the world especially within the content of destination marketing. Hence, in the background section, firstly the relationship between the concept of augmented reality and virtual reality, and the hardware and software infrastructure of the augmented reality is discussed. Afterwards, augmented reality applications in the tourism industry are discussed together with relevant examples. In the conclusion section, technical and social problems of the augmented reality applications are discussed and proposed solutions are presented.

### **BACKGROUND SECTION**

### Augmented Reality

AR is a combination of real-object and computer-generated data where the virtual object has been collated with the real world. This means that it is visible to the user that the virtual field and the real object co-exist in the same place (Azuma, 1997). The AR technology proposes that the real and virtual objects coexist in the same place (Linaza, Gutierrez, & García, 2014, p.498). Today, especially the rapid development of the latest generation of mobile devices now allows anyone with a computer or just a smartphone to be able to make connections in real time and get information from the virtual world as a result of augmented reality through the overlap of information layers (Garau, 2014). The rapid development in the use of smartphones and mobile applications create new ways of connecting with the visitors in many sectors as well as in the tourism industry (Kennedy-Eden & Gretzel, 2012).

16 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage: www.igi-global.com/chapter/augmented-reality/192760?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

The Effect of Room Service, Comfort, and Reservation Process on The Performance Dimension Evaluation: The Case of Grand Mercure Hotel Gajah Mada Jakarta

Adilla Anggraeni and Meyliza Thorina (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 42-52).* 

www.igi-global.com/article/the-effect-of-room-service-comfort-and-reservation-process-on-the-performance-dimension-evaluation/189744?camid=4v1a

Can Educational Tourism Spur Economic Growth in Malaysia?: Evidence from Neoclassical Growth Model

Hylmee Matahir and Chor Foon Tang (2017). *Tourism and Opportunities for Economic Development in Asia (pp. 120-131).* 

www.igi-global.com/chapter/can-educational-tourism-spur-economic-growth-in-malaysia/176294?camid=4v1a

Studying the Impact of Egyptian Hotels' Websites Marketing on Customers' E-Satisfaction Nancy Awad and Sherif Saad Soliman (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 53-63).* 

www.igi-global.com/article/studying-the-impact-of-egyptian-hotels-websites-marketing-on-customers-e-satisfaction/189745?camid=4v1a

Expatriate Selection and Retention: Identifying and Assessing the Other Characteristics beyond Knowledge, Skills, and Abilities

Ben Tran (2015). *Handbook of Research on Global Hospitality and Tourism Management (pp. 468-492).* www.igi-global.com/chapter/expatriate-selection-and-retention/135821?camid=4v1a

# Multi-Dimensional Acquisition, Representation, and Interaction of Cultural Heritage Tangible Assets: An Insight on Tourism Applications

# Cristina Portalés

Universitat de València, Spain

### Sergio Casas

Universitat de València, Spain

# Pau Alonso-Monasterio

Universitat Politècnica de València, Spain

### María José Viñals

Universitat Politècnica de València

# **ABSTRACT**

The acquisition, digital representation and interaction of cultural heritage (CH) assets is of high interest for an accurate documentation of our cultural legacy. Detailed studies demand more information on the assets apart from the mere visual appearance or even shape of objects. In this regards, some research works can be found that combine shape with other kind of data, such as radiometric (e.g., IR, thermal radiation, etc.) or semantic information, leading to multi-Dimensional (mD) virtual objects. In this chapter, we review the state of the art of different techniques to deal with mD acquisition of CH tangible assets. We also discuss some representation (e.g., visual, sonic, etc.) and interaction approaches to enhance the value of CH by means of the technologies of virtual reality (VR) and augmented reality (AR), considering also the application of these techniques to tourism. Finally, we also point out some current needs and technological barriers that need to be tackled by the research community regarding the acquisition and dissemination of heritage.

# INTRODUCTION

Cultural Heritage (CH) is a fundamental expression of the richness and diversity of our culture and therefore, its conservation, documentation and dissemination is considered of the utmost importance. Beyond the well-established methods that have been traditionally employed in the field of CH up to the present moment, technological tools have become increasingly popular for the heritage enhancement, tourism experience development and dissemination. This is because they have the ability to engender fascination far beyond a tourist brochure or any other printed material.

The digital documentation and dissemination of CH tangible assets through current technologies usually involves several steps, such as the visual appearance and shape (or three-dimensional, 3D) acquisition of the assets, their virtual reconstruction, storage (e.g. repositories), digital representation or reproduction (mainly visual), and interaction techniques, among others. This chapter focuses on the acquisition, representation and interaction techniques with emphasis in multi-Dimensional (mD) data.

The acquisition step is fundamental, as it deals directly with digitizing the real characteristics of the assets. Thus, it plays an important role in the accurate documentation and safeguard of our cultural legacy. However, not all the acquisition technologies have been exploited in depth so far for the CH sector. In this regards, some research works can be found that combine shape with other kind of data, such as radiometric (e.g. IR, thermal radiation, etc.) or semantic information, leading to the mD virtual objects and opening the avenues for new dissemination and understanding of heritage. One of the aims of this chapter is to provide a review of these acquisition technologies, in order to point out the benefits that they might have for CH and to discuss current needs, technological barriers and/or new trends that might be of interest for the sector.

Additionally, this chapter also aims to give a review on the current representation (not only visual, but also sonic, haptic, etc.) and interaction solutions related to touristic applications, focusing on the technologies of virtual and augmented reality that are increasingly being used in museums, exhibitions, heritage sites, etc., being a vehicle for the access and dissemination of CH to a number of stakeholders and the wide public. Indeed, the dissemination of CH is essential to promote its conservation, so communicating the values and the significance of our heritage, especially to the general public, is a key issue for rising awareness. The thematic interpretation is the main tool to achieve proactive behaviours in general public towards conservation, and it is based upon the following sequence: "Through interpretation, understanding; through understanding, appreciation; through appreciation, protection" (Tilden, 1957, p. 38). The scientific literature regarding the role of new technologies in the dissemination of heritage is extensive. However, there is nothing from the methodological point of view in relation to the specific techniques analysed in this work. Therefore, to evidence this type of applications linked to the dissemination of heritage, various examples developed by the authors are provided through the chapter.

This chapter is organized as follows. In a first section, a review on the state of the art of different available techniques and the related methodologies to deal with mD acquisition for the virtual reconstruction of CH tangible assets are presented and discussed through different examples. In the next section, a review on various visualization and interaction approaches (based on virtual and augmented reality) to enhance the value of CH by means of these virtually reconstructed objects is given. Finally, the discussion section points out some current needs and technological barriers that need to be tackled by the research community in order to further advance on the given topic, which might be relevant for the CH sector, as argumented in the previous sections.

22 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/multi-dimensional-acquisition-representation-and-interaction-of-cultural-heritage-tangible-assets/192761?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

Local Community Attitude and Support Towards Tourism Development at Saint Martin Island, Bangladesh: Local Community Attitude and Support

Md. Abdul Hai and Md. Badsha Alamgir (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 32-41).* 

www.igi-global.com/article/local-community-attitude-and-support-towards-tourism-development-at-saint-martin-island-bangladesh/189743?camid=4v1a

#### Role of ICT And Tourism in Economic Growth of India

Surender Kumar, Karuna Chauhan and Abhay Kumar Srivastava (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-16).* 

www.igi-global.com/article/role-of-ict-and-tourism-in-economic-growth-of-india/177119?camid=4v1a

# Marketing Your City's Industries to the World: Building and Retaining Export Oriented Clusters through Strategic ICT Investments

Nicholas P. Robinson and Prescott C. Ensign (2009). *Information Communication Technologies and City Marketing: Digital Opportunities for Cities Around the World (pp. 203-220).* 

www.igi-global.com/chapter/marketing-your-city-industries-world/22580?camid=4v1a

# Middle East Tourists' Behaviour Towards Malaysian Gastronomy and Dining Experience in Malaysia

Quee-Ling Leong, Shahrim Ab Karim, Bee-Lia Chua and Kallayanee Tengpongsathon (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 64-79).* 

www.igi-global.com/article/middle-east-tourists-behaviour-towards-malaysian-gastronomy-and-dining-experience-in-malaysia/189746?camid=4v1a

# An Initial Framework to Develop a Mobile Five Human Senses Augmented Reality System for Museums

João M. F. Rodrigues

University of Algarve, Portugal

Pedro J. S. Cardoso

University of Algarve, Portugal

Joana Lessa

University of Algarve, Portugal

João André Pereira

University of Algarve, Portugal

João D. P. Sardo

University of Algarve, Portugal

Marco de Freitas

University of Algarve, Portugal

Jorge Semião

University of Algarve, Portugal

Jânio Monteiro

University of Algarve, Portugal & INESC-ID, Portugal

Célia M. Q. Ramos

University of Algarve, Portugal

Roberto Lam

University of Algarve, Portugal

**Eduardo Esteves** 

University of Algarve, Portugal & CCMAR Centre of Marine Sciences, Portugal

Mauro Figueiredo

University of Algarve, Portugal

Alexandra R. Gonçalves

University of Algarve, Portugal

**Miguel Gomes** 

SPIC - Creative Solutions, Portugal

Paulo Bica

SPIC - Creative Solutions, Portugal

# **ABSTRACT**

The Mobile Five Senses Augmented Reality System for Museums (M5SAR) project aims to development an Augmented Reality mobile system for museums. Museums are amazing places, where it is important to sensorial augment as much as possible the visits, permitting to see, ear, touch, smell, and taste all the interesting objects there exist. Also fundamental is that visitors with different profiles (child, adult, expert, etc.) can have different experiences when visiting the same object. In the M5SAR system, the visitor uses its smartphone to select the object(s) to explore, and the user interface adapts on-the-fly to the object(s) and user's profile. Simultaneously, when integrated, a paired hardware device allows the extension of the augmented reality system to the human five senses, complementing the visual and auditory information about the objects. This chapter presents the initial framework to develop a five senses mobile adaptive museum system.

# INTRODUCTION

The Mobile Five Senses Augmented Reality System for Museums (M5SAR) project aims at the development of an enhanced Augmented Reality (AR) system, to be a guide in cultural, historical and museum events, complementing or replacing the traditional orientation given by tour guides, directional signs, or maps. The system consists of a (i) smartphone/phablet application (APP) and a (ii) hardware device (referenced as HDevice) to be integrated with smartphones/phablets, in order to explore the five human senses: sight, hearing, touch, smell, and taste. Both components, (i) and (ii), can work in stand-alone or integrated fashion.

Nowadays, many personal and context-aware tourism and cultural experiences are constructed based on mobile APPs (Jung, Chung, & Leue, 2015), including the ones that use AR. Those APPs are increasing in number due to the popularity of built-in cameras, global positioning systems, and with the massive availability of Internet connections. On the other hand, most of the present User Interfaces (UI) still traditionally follow a one-size-fits-all model, typically ignoring the needs, abilities, and preferences of individual users. However, past research pointed out that visualization performance could be improved by adapting some of its aspects to the individual user (Steichen, Conati & Carenini, 2014). Conati, Carenini, Toker, and Lallé (2015) state that intelligent adaptive user interfaces (AUI) and/or visualizations, that can adapt on-the-fly to the specific needs and abilities of each individual user, are a long-term research goal. This is due to two main difficulties: (a) the extraction of information about the users' needs and abilities and (b) the implementation of UIs that can adapt/change "themselves" on-the-fly.

Reinecke and Bernstein (2013) refer that a modular UI that allows a flexible composition of various interface elements increases the number of variations of the interface element to the power of the number of adaptable elements. Thus, instead of designing each interface from scratch, a modular user interface approach is possibly a better solution, once it allows achieving many more versions with less design effort. Therefore, the requirement changes to the creation of different designs for all those parts of the interface that are subject to user or cultural preferences. Equal importance should be given to the UI adaptation to users with different impairments. Unfortunately, because of the great variety of existing impairments, it is expectable that manually and, probably, modular designing interfaces, for each one of those impairments, is impractical and not scalable (Gajos, Wobbrock, & Weld, 2008; Rodrigues, Lessa, Gregório, Ramos, & Cardoso, 2016). Nevertheless, the modular and/or adaptive generation of

22 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage: <a href="https://www.igi-global.com/chapter/an-initial-framework-to-develop-a-mobile-five-human-senses-augmented-reality-system-for-museums/192762?camid=4v1">www.igi-global.com/chapter/an-initial-framework-to-develop-a-mobile-five-human-senses-augmented-reality-system-for-museums/192762?camid=4v1</a>

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian: www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

# Systems Approach as a Creative Driving Force for a Tourism Destination

Tadeja Jere-Jakulin (2017). *Driving Tourism through Creative Destinations and Activities (pp. 1-19).* www.igi-global.com/chapter/systems-approach-as-a-creative-driving-force-for-a-tourism-destination/174514?camid=4v1a

Local Community Attitude and Support Towards Tourism Development at Saint Martin Island, Bangladesh: Local Community Attitude and Support

Md. Abdul Hai and Md. Badsha Alamgir (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 32-41).* 

www.igi-global.com/article/local-community-attitude-and-support-towards-tourism-development-at-saint-martin-island-bangladesh/189743?camid=4v1a

The Effect of Room Service, Comfort, and Reservation Process on The Performance Dimension Evaluation: The Case of Grand Mercure Hotel Gajah Mada Jakarta

Adilla Anggraeni and Meyliza Thorina (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 42-52).* 

www.igi-global.com/article/the-effect-of-room-service-comfort-and-reservation-process-on-the-performance-dimension-evaluation/189744?camid=4v1a

Understanding the Importance of Property Management Practices in Indian Hotel Industry Mohinder Chand Dhiman and Manish Anand (2015). *International Tourism and Hospitality in the Digital Age* (pp. 205-219).

www.igi-global.com/chapter/understanding-the-importance-of-property-management-practices-in-indian-hotel-industry/130103?camid=4v1a

# Applying an Augmented Reality Tool to the Camino de Santiago in Portugal

# Miguel Nuno Pereira

Minho University, Portugal

# Miguel Pazos Otón

University of Santiago de Compostela, Spain

### José Cotos

University of Santiago de Compostela, Spain

#### Paula Cristina Remoaldo

Minho University, Portugal

# **ABSTRACT**

This chapter focuses on an Interactive Guide of the Portuguese Way of St. James (Camino de Santiago) created by using an Augmented Reality application. This application was developed specifically for the section of the Portuguese Route to Santiago that crosses the municipality of Barcelos (Northwest of Portugal). The Guide has the geo-information needed for the Pilgrim, in the passage by the municipality of Barcelos to Santiago, and can be accessed through a smartphone. The pilgrim will have at his or her disposal a set of interactive geo-information about Barcelos. At the level of interactivity with the user, each of these points of interest in the Guide will be signaled by a portrait done by a painter from Barcelos (monuments) and universal graphic icons (other geo-information).

### INTRODUCTION

In global world where speed and availability of information are increasingly important values, new technologies play a key role. This scenario is due to the significant growth in the number of inventions that have emerged throughout our most recent history, from the invention of the telephone, radio, tele-

vision to the Internet. Information must become a global asset of rapid spread and low cost of access. Geographic information is a particular case that is worth studying.

The new trends present a perfect integration between geographic information and virtual reality. In its turn, the usage of Virtual Reality in tourism is also currently increasing (Butler, 2016; Graham, 2016). Presently, the usage of geographic information tools that use Virtual Reality have become more common, more specifically one of its variations - the Augmented Reality (Milgram et al., 1994; Azuma, 1997; Ajanki et al., 2010; Charara, 2016).

Tourism and its value chain, as well as the territories themselves, undergo constant changes as a result of the technological advances, some of them being perceived in real time. According to recent data from the World Tourism Organization (UNWTO, 2014), approximately 60% of current tourists consult and decide their holiday destinations through the Internet, of which around 40% make their reservations via the Internet. These facts lead us to conclude that an important change in the habits of the tourists has happened. Increasingly, travel agencies (intermediaries in the tourism value chain) are losing importance, since the tourist directly book their flight through the reservation center of an airline and their accommodation through the reservation center of any hotel chain. Thus, we are in a new era, the one that Manuel Castells (2002) calls "informationalism", which in turn, originates to what he also calls a new form of capitalism, "informational capitalism."

However, in a postmodern and transforming society, where technology and distance communication prevails, the concept of travel has also undergone transformations. Therefore, the journey began to be made by many from the Internet, in real time, often transcending geographical boundaries, social distances and the very concept of physical travel (Dennis & Urry, 2009).

Based on these assumptions, we chose to develop a research between 2010 and 2014 focused on the following objectives: to explore the topic of Augmented Reality and its importance in territorial marketing, promotion, dissemination and access to tourism; and create a guide for mobile devices. The latter is an interactive guide with Augmented Reality, which supports the pilgrim during his or her visit to the municipality of Barcelos.

The intention was to relate the tools and new technologies accessible to individuals and institutions with the availability, sharing and usage of tourism information. Also demonstrate how to promote a territory, using new technologies in a simple and objective way, trying to attract new tourists and new investments, and contributing to improve the quality of the visit. It's a matter of integrating and making available to the citizen and institutions all the tourist information of a territory in an interactive and accessible way.

The selected territory corresponded to the municipality of Barcelos located in the Northwest of Continental Portugal. It's a territory linked to the Jacobean Pilgrimages, from the beginnings of the Middle Ages. The Legend of the Rooster is the ultimate exponent of this secular connection. The Portuguese Way of Santiago crosses the municipality of Barcelos in a direction of south to north, in an extension of 32 kilometers. It is estimated that about 17,500 pilgrims pass through Barcelos every year, and in Holy Years they can exceed 25,000 pilgrims (Barcelos Tourist Office, 2014).

In order to reach the objectives considered, after collecting the data and the field work carried out including meetings with all stakeholders and agents of the territory, a Geographic Information System applied to municipal tourism planning and management was created and also an interactive guide on The Portuguese Way of Santiago in Barcelos.

This application was the first author application made in Portugal about the Portuguese Way of Santiago in Barcelos. Worldwide it was the second to be implemented, since in 2010 Banco Bilbao Viscaya sponsored and published a guide of the French Way of Santiago, during the last Holy Year.

18 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/applying-an-augmented-reality-tool-to-the-camino-de-santiago-in-portugal/192763?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian: www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

How Do Online Reviews Affect Business Travelers' Accommodation Choices?: The Application of Theory of Reasoned Action (TRA)

Faranak Memarzadeh, Shane Blum and Charlie Adams (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 50-65).* 

www.igi-global.com/article/how-do-online-reviews-affect-business-travelers-accommodation-choices/177122?camid=4v1a

#### "España Verde": Tourism Destination Image among German Facebook Users

Till Maximilian Schuh, Dora Agapito and Patrícia Pinto (2018). *Handbook of Research on Technological Developments for Cultural Heritage and eTourism Applications (pp. 284-311).*www.igi-global.com/chapter/espaa-verde/192771?camid=4v1a

Studying the Impact of Egyptian Hotels' Websites Marketing on Customers' E-Satisfaction Nancy Awad and Sherif Saad Soliman (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 53-63).* 

www.igi-global.com/article/studying-the-impact-of-egyptian-hotels-websites-marketing-on-customers-e-satisfaction/189745?camid=4v1a

# E-Marketing and the Hotel Industry: Calculating Web Presence Index (WPI) for the Hospitality Sector

Roli Bansal, Manisha Lamba, Shirin Alavi and Vandana Ahuja (2015). *Hospitality, Travel, and Tourism: Concepts, Methodologies, Tools, and Applications (pp. 303-313).* 

www.igi-global.com/chapter/e-marketing-and-the-hotel-industry/119222?camid=4v1a

# Chapter 7 Reconstructing the Past: Providing an Enhanced Perceptual Experience

# Luís G. Magalhães

Centro Algoritmi, University of Minho, Portugal

#### Telmo Adão

University of Trás-os-Montes e Alto Douro, Portugal & INESC TEC, Portugal

#### **Emanuel Peres**

University of Trás-os-Montes e Alto Douro, Portugal & INESC TEC, Portugal

### **ABSTRACT**

Accurate modeling/reconstruction and visualization of real environments, particularly archaeological sites, is both a major challenge and a crucial task. This work will address the entire process of the virtual reconstruction of archaeological sites, since the construction of the virtual model until its visualization. The chapter begins with an introduction to the process of virtual reconstruction of archaeological sites, where the several stages that should take place to obtain a faithful virtual representation of an archaeological site and its artifacts are identified. Moreover, each stage is characterized and its main methods and techniques are identified, in dedicated sections. The authors' contribution for the state of the art will be highlighted in each stage. The chapter ends with the authors' vision about future trends for this field and unveils what could be their contributions to this vision.

# INTRODUCTION

Accurate modeling/reconstruction and visualization of real environments, particularly archaeological sites, is simultaneously a major challenge and a crucial task. It allows the digital preservation and the visualization of not only current cultural assets, but also of those who no longer exist, resulting in a true glimpse of the past. This enables experts to study and to interact with their objects of interest, while allowing the general public to visualize such virtual environments, fostering cultural, social and scientific participation. By using mobile devices, such as smartphones or tablets, virtual models can be visualized

#### Reconstructing the Past

in loco, giving the experts a better understanding of how objects of interest integrate in their original environment and providing the general public an enhanced perceptual experience.

Despite sometimes being used as synonyms, in the context of this chapter "modeling" and "reconstruction" are considered as similar concepts, but with different working objects. The modeling task aims to build a model of an existent object while the reconstruction task is concerned with building models of no longer existent objects or model (reconstruct) the original shape of damaged objects, based on some information.

Highly accurate modeling/reconstruction is a very complex and time consuming task. To achieve a faithful representation of the object of interest all stages of the reconstruction and visualization processes must be carefully executed. These stages include constructing the object's virtual model, mimic the ancient original illumination conditions, to render and visualize the model, under the correct context.

This work will address the entire process of the virtual reconstruction of archaeological sites, since the construction of the virtual model until its visualization. The chapter begins with an introduction to the process of virtual reconstruction of archaeological sites, where the several stages that should take place to obtain a faithful virtual representation of an archaeological site and its artifacts are identified. Moreover, each stage is characterized and its main methods and techniques are identified, in dedicated sections. The authors' contribution for the state of the art will be highlighted in each stage. The chapter ends with the authors' vision about future trends for this field and unveils what could be their contributions to this vision.

#### VIRTUAL RECONSTRUCTION PROCESS

The use of computers in Archaeology dates back to the late 1950s and early 1960s, essentially for the statistical treatment of information collected during excavations (Reilly & Rahtz, 1992). At that time, the use of computers was confined to a very limited set of archaeologists. In the early 1970s, a group of enthusiasts and experts in the use of computers in Archaeology joined in and started a cycle of international conferences designated as "Computer Applications in Archaeology", which still takes place with the title "Computer Applications and Quantitative Methods in Archaeology". In the following years, the use of computers in Archaeology has evolved to the use of computer aided design (CAD) systems, database management systems (DBMS) and geographic information systems (GIS). The first, besides allowing to digitally store the topographic survey carried out during the field work also enabled, based on the acquired data, some earlier works on the 3D reconstruction of both the terrain and the archaeological site's existing structures. While both DBMS and GIS allowed storing data acquired during the excavation, GIS enabled storing, manipulating and visualizing data with a spatial nature. The term "Virtual Archaeology" as a reference to the use of Virtual Reality in Archaeology was first mentioned by Reilly (1990), but the creation of virtual reconstructions in the field of archaeology began in the 1980s.

Since then, the use of computers, particularly the virtual reconstruction as a support to research and promotion of cultural heritage, has gained an increasing interest. Presently there are several international conferences dedicated to this topic. In the last years there has been a significant increase in the number of projects involving some sort of historical reconstruction as well as its web promotion, which causes a greater impact and consequently affects more people. This increase can be related to two factors: (i) the technological development, which allows to obtain these reconstructions more easily; (ii) the increased

31 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage: www.igi-global.com/chapter/reconstructing-the-past/192764?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

# Key Planning Factors for Deploying Ubiquitous Wi-Fi Networks in Hotels

Galen R. Collins (2017). *International Journal of Tourism and Hospitality Management in the Digital Age* (pp. 18-31).

www.igi-global.com/article/key-planning-factors-for-deploying-ubiquitous-wi-fi-networks-in-hotels/189742?camid=4v1a

The Effect of Room Service, Comfort, and Reservation Process on The Performance Dimension Evaluation: The Case of Grand Mercure Hotel Gajah Mada Jakarta

Adilla Anggraeni and Meyliza Thorina (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 42-52).* 

www.igi-global.com/article/the-effect-of-room-service-comfort-and-reservation-process-on-the-performance-dimension-evaluation/189744?camid=4v1a

### An Overview of Reproductive Tourism

(2018). Legal and Economic Considerations Surrounding Reproductive Tourism: Emerging Research and Opportunities (pp. 1-17).

www.igi-global.com/chapter/an-overview-of-reproductive-tourism/192646?camid=4v1a

Local Community Attitude and Support Towards Tourism Development at Saint Martin Island, Bangladesh: Local Community Attitude and Support

Md. Abdul Hai and Md. Badsha Alamgir (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 32-41).* 

www.igi-global.com/article/local-community-attitude-and-support-towards-tourism-development-at-saint-martin-island-bangladesh/189743?camid=4v1a

# Assessing the Use of Archaeological Sites as Cultural Tourism Resources: The Case of Segóbriga (Spain)

#### Nuria Huete-Alcocer

University of Castilla-La Mancha, Spain

#### María Pilar Martínez-Ruiz

University of Castilla-La Mancha, Spain

# Víctor Raúl López-Ruiz

University of Castilla-La Mancha, Spain

### **ABSTRACT**

This chapter offers an in-depth analysis of archaeological sites and the value they represent to the field of cultural tourism. Specifically, this chapter discusses the suitable means of studying and managing archaeological sites, which represent tangible examples of cultural tourism. Managed properly, such sites may attract a larger number of visitors and thereby contribute to the region's socioeconomic development. In order to illustrate this line of research, this analysis will focus on one archaeological site in particular, namely the Archaeological Park of Segóbriga in the province of Cuenca, Spain. More specifically, the chapter reviews survey data collected over several years by related public institutions. The findings lend support to several measures that may improve the management of such sites in the context of cultural tourism. The most important of these is the dissemination and promotion of information about the site through WOM and eWOM.

# INTRODUCTION

As one of the main economic activities in the world, tourism has become an important source of income for many countries (Su & Lin, 2014), as well as a creator of jobs in various service industries (Yang, Lin, & Han, 2010). This is particularly true in Spain, which has become one of the world's most important tourism powers—placing third just behind France and the United States (World Tourism Organization, 2016). Spain not only possesses many attractions, but also rivals China and Italy in terms of the number of goods it produces (UNESCO, 2015). Meanwhile, Spain's position on the World Heritage list (a marketing tool used by national tourism campaigns to attract a large number of visitors; Li, Wu, & Cai, 2008) speaks to the richness of its historical and archaeological heritage (Boto, 2016). As cultural tourism has gradually become pivotal to a destination's attractiveness (McKercher, Ho, & Du Cros, 2005), Spain has outpaced many other countries in leveraging its cultural resources, becoming the second-most important country for this type of tourism (Santos & Meléndez, 2016). Case in point: In 2015, the total volume of visitors to Spain exceeded 68 million foreigners, a year-on-year increase of 4.9% (INE, 2016).

In short, culture constitutes an important part of tourism, as it increases the attractiveness and competitiveness of tourist destinations (Hennessey, Yun, & MacDonald, 2014; Pal, 2015). Indeed, several studies have shown that tourists are particularly drawn to destinations with a richer cultural or natural heritage (e.g., Bille & Schulze, 2006; Cooke & Lazzaretti, 2008; Su & Lin, 2014; Yang et al., 2010). Through cultural tourism, visitors gain insight into the history, culture and way of life of other regions of the world (Pal, 2015). As cultural tourism gains ever more importance (Chen & Chen, 2010), it is worth noting that the fastest-growing places are those that offer tourists a glimpse into both the tangible and intangible sides of cultural experience(Datta et al., 2015).

Considering the above, this work places special emphasis on a cultural destination – Segóbriga – that possesses a unique archaeological heritage. Located in the province of Cuenca, Segóbriga is one of the five archaeological parks of Castilla - La Mancha. Having ascended as a cultural destination in recent years, this area serves as a useful case study in how tourism management can leverage archaeological sites as tangible cultural resources. Through proper management, such sites may attract more visitors and thereby contribute to the socio-economic development of the areas involved. In order to illustrate this line of research, this chapter will explore how the archaeological park of Segóbriga can act as an important historical resource for cultural tourism in that area.

# BACKGROUND

In order to accomplish the above goal, we need to first establish some conceptual distinctions regarding cultural tourism and, more specifically, archaeological heritage.

Cultural tourism emerged in the early 1990s, originally for the benefit of elite clientele, but was eventually generalized as an alternative to mass tourism that values the culture of the visited community (Jovicic, 2016). Silberberg (1995) offered one of the first and most enduring definitions of cultural tourism: namely, as "visits by people outside the host community motivated in their entirety or partly by interest in the historical, artistic, scientific or lifestyle offerings, heritage of a community, region, group or institution" (p. 361). Similarly, Wall and Mathieson (2006) defined cultural tourism in terms of "tourists who experience and have contact with a host population and its cultural expressions, experiencing the uniqueness of culture, Heritage and the characters of its place and people" (p. 261; see also Lynch

16 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/assessing-the-use-of-archaeological-sites-ascultural-tourism-resources/192765?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian: www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

The Effect of Room Service, Comfort, and Reservation Process on The Performance Dimension Evaluation: The Case of Grand Mercure Hotel Gajah Mada Jakarta

Adilla Anggraeni and Meyliza Thorina (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 42-52).* 

www.igi-global.com/article/the-effect-of-room-service-comfort-and-reservation-process-on-the-performance-dimension-evaluation/189744?camid=4v1a

# Middle East Tourists' Behaviour Towards Malaysian Gastronomy and Dining Experience in Malaysia

Quee-Ling Leong, Shahrim Ab Karim, Bee-Lia Chua and Kallayanee Tengpongsathon (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 64-79).* 

www.igi-global.com/article/middle-east-tourists-behaviour-towards-malaysian-gastronomy-and-dining-experience-in-malaysia/189746?camid=4v1a

# Experience Marketing and Tourism: An Application of Service-Dominant Logic to Addiopizzo Travel

llenia Bregoli (2016). *Global Dynamics in Travel, Tourism, and Hospitality (pp. 58-69).* www.igi-global.com/chapter/experience-marketing-and-tourism/156749?camid=4v1a

# Service Design: New Methods for Innovating Digital User Experiences for Leisure

Satu Miettinen (2011). Digital Culture and E-Tourism: Technologies, Applications and Management Approaches (pp. 36-47).

www.igi-global.com/chapter/service-design-new-methods-innovating/49618?camid=4v1a

# Accessibility Improvement Interventions at Byzantine Monuments:

Use of Technology for Facilitating Accessibility of Visitors With Sensory Disabilities

**Aristotelis Naniopoulos** 

Aristotle University of Thessaloniki, Greece

**Panagiotis Tsalis** 

Aristotle University of Thessaloniki, Greece

# **ABSTRACT**

The accessibility of monuments and archaeological sites by disabled persons and persons with restricted mobility in general, constitutes a social, financial and political demand. The project "PROSPELASIS" attempted to counter this problem by focusing on creating a methodology for facing monuments' accessibility and perceptibility problems for people with disabilities and testing its application at Byzantine monuments of Thessaloniki. In six major monuments included in UNESCO's World Heritage List, significant accessibility improvements were realized. A Wi-Fi information system was installed in all of them via which an audio and visual information system was created with audio information in three languages, description of visual information, text information, information in Greek and International Sign Language. Additionally, in Rotunda and Heptapyrgion monuments three dimensional models were developed. The successful validation of the proposed methodology constitute the results of this project not only pertinent to Thessaloniki or Greece, but worthy of a wider application.

# INTRODUCTION

A visit to an archaeological site or monument is a unique experience that also constitutes a main tourist attraction. A visit to a monument allows the composition of all stimuli into a unique experience, consisting of the following:

- Visual Stimuli: Shapes, colours, textures, images, sense of the real scale of the site.
- **Tactile Stimuli:** The floor's texture and the way in which it is perceived (e.g. gravel or stone) or the texture of the seats (e.g. wood, marble, metal).
- Auditory Stimuli: The soundscape of the surrounding area, either natural or artificial, the sound
  of interaction between man and the environment, walking, the swish of plants when walking
  through them.
- **Olfactory Stimuli:** Natural scents (depending on the season, time, weather) and artificial scents (e.g. scents of diverse materials or deliberately diffused scents and incenses).
- **Gustatory Stimuli:** Tastes referring to the site and underlining its uniqueness, such as fruits from the trees of a garden, the water of a fountain, the treat in a monastery.

In Thessaloniki, a city with rich heritage, monuments constitute isolated "islets", with the modern cityscape "drowning" them, allowing no visual connection between them, making their identification, their understanding, and physical access to them difficult for visitors in general and persons with restricted mobility in particular. This problem is exacerbated by the fact that the current city's ground level is often four to five meters above the level, where the entrance to the monuments is. Thus, any attempt to "socialize" the cultural heritage of the city has to pursue the following two objectives: a) to provide easy access to monuments for all visitors, including persons with disability and restricted mobility, and b) to help visitors through the provision of adequate information and the indication of ways of moving from one monument to another, connecting the" pieces of the puzzle" (Naniopoulos et al., 2011b.).

The present chapter describes the process of improvement of accessibility in selected Byzantine monuments of Thessaloniki using a methodology created in the frame of the "PROSPELASIS" project, financed by a grant from Iceland, Liechtenstein and Norway through the EEA Financial Mechanism 2004 – 2009 (50%) and from the public investments programme of the Hellenic Republic (50%). The project was realized by the cooperation of the Aristotle University of Thessaloniki (project leader) and the 9th Ephorate of Byzantine Antiquities. The chapter describes the background and the needs that lead to the development of the methodology, the methodology itself, its implementation and the accessibility improvement interventions carried out at selected Byzantine monuments of Thessaloniki. The authors aspire that the adoption of the proposed methodology can lead to further actions that will improve monuments' and archaeological sites' accessibility at national and international level.

#### BACKGROUND

Addressing, at an international level, the problems of protection, conservation, restoration and management of historical centres, archaeological sites, architectural aggregations, monuments and works of art, has been a major issue. Nowadays, we are not only interested in the monuments' past; we are also interested in their preservation, restoration and role in direct relation to their present and future environ-

22 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/accessibility-improvement-interventions-at-byzantine-monuments/192766?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian: www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

# Middle East Tourists' Behaviour Towards Malaysian Gastronomy and Dining Experience in Malaysia

Quee-Ling Leong, Shahrim Ab Karim, Bee-Lia Chua and Kallayanee Tengpongsathon (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 64-79).* 

www.igi-global.com/article/middle-east-tourists-behaviour-towards-malaysian-gastronomy-and-dining-experience-in-malaysia/189746?camid=4v1a

#### Key Planning Factors for Deploying Ubiquitous Wi-Fi Networks in Hotels

Galen R. Collins (2017). *International Journal of Tourism and Hospitality Management in the Digital Age* (pp. 18-31).

www.igi-global.com/article/key-planning-factors-for-deploying-ubiquitous-wi-fi-networks-in-hotels/189742?camid=4v1a

# Middle East Tourists' Behaviour Towards Malaysian Gastronomy and Dining Experience in Malaysia

Quee-Ling Leong, Shahrim Ab Karim, Bee-Lia Chua and Kallayanee Tengpongsathon (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 64-79).* 

www.igi-global.com/article/middle-east-tourists-behaviour-towards-malaysian-gastronomy-and-dining-experience-in-malaysia/189746?camid=4v1a

# The Impact of Social Network on Italian Users' Behavioural Intention for the Choice of a Medical Tourist Destination

Francesca Di Virgilio, Angelo A. Camillo and Isabell C. Camillo (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 36-49).* 

www.igi-global.com/article/the-impact-of-social-network-on-italian-users-behavioural-intention-for-the-choice-of-a-medical-tourist-destination/177121?camid=4v1a

# A Framework Supported by Modeling and Virtual/Augmented Reality for the Preservation and Dynamization of Archeological-Historical Sites

# Mauro Figueiredo

University of Algarve, Portugal

#### João Pedro Bernardes

University of Algarve, Portugal

### José I. J. Rodrigues

University of Algarve, Portugal

# César Gonçalves

University of Algarve, Portugal

# **ABSTRACT**

This chapter presents an approach for the preservation and exploration of the archaeological-historical information using the most modern methods of Building Information Modelling (BIM) together with virtual and augmented reality adapted to archaeology. It was made a survey of the archaeological site using an unmanned aerial vehicle (UAV) and laser scan to obtain accurate information of existing structures which is stored in a BIM model. By using BIM methodologies, all existing information is organized in one place, shared and preserved for future memory. This 3D model is used to virtually reconstruct the Milreu archeological site, bringing this ancient Roman villa to life again, and supports the virtual reality platform and augmented reality tourism application for mobile devices, used for the promotion of the archaeological site, in an innovative way, in order to ensure a high degree of satisfaction to regular visitors and attract new public, looking for the visits to the archaeological-historical site to become a memorable experience.

### INTRODUCTION

The ruins of Milreu are a good example of a Roman villa of great archaeological importance (Santos, 1972), that is not immediately visible to the visitor without professional or specialized knowledge to directly get a correct idea of what that place would be like at the time.

This chapter presents an approach that enhances the preservation of information in a centralized system that allows exploration and valorization of all archaeological information.

This approach is supported by Building Information Modeling (BIM) methodologies allowing the integration of different types of archaeological information into a shared database, constituting a source of 2D and 3D information that is shared and exploited by visitors and / or specialists.

An unmanned aerial vehicle (UAV) conducts a topographical survey of the ruins of Milreu and thus obtain accurate information of the existing constructions, which is the basis that will allow the recording of all the existing information about Milreu. Such records include information about the interventions, interpretations and models of the different epochs.

Virtual and augmented reality technologies are integrated in the tourism application is used to improve the interpretation and the understanding of Milreu to the visitors, using mobile devices as an artifact mediator in the discovery of this archaeological site. Virtual 3D models of some spaces where historical-archaeological information exists were built to support the virtual and augmented reality platform for mobile devices. In this way, the visitor is able to explore a three-dimensional virtual environment on the web and on a smartphone or a tablet that helps him to interpret the archaeological site.

The creation of the virtual space of Milreu allows the visitor the interactive and immersive exploration in a 2D / 3D environment presenting the history not visible directly from the archaeological site. In this way, the three-dimensional virtual reconstruction of the archaeological heritage reveals the importance of these places at the time to the general public by allowing, in an easy and interactive way, the visualization, understanding and exploration of these spaces through Internet.

On the other hand, when using augmented reality, the visitor obtains information superimposed on reality, when he is physically visiting the ruins of Milreu. Which helps him to read and interpret the different spaces, vestiges, objects and existing buildings and, in this way, to transform the visit into a memorable experience.

# **MILREU**

The ruins of Milreu, located in Estoi, 7 Km from Faro, correspond to a Roman villa with a long period of occupation between the 1st century and 5th century, being one of the largest and best preserved Roman villa in the southwest of the Iberian Peninsula. It is classified as a national monument since 1910, after the excavations of the 19th century, carried out by archaeologist Estácio da Veiga, have revealed the archaeological importance of the site (Santos, 1972). From then, becomes a reference among the Portuguese archaeological sites, beginning a wide-ranging programme of excavations from de1971, coordinated by Theodor Hauschild. It is then that is definitely in sight the residential and richest part of the villa, the *pars urbana*, which was acquired by the State and is valued and converted into a visitable archaeological site. In this part, are to highlight the mosaics with marine motifs, the temple, the baths and the residential area organized around a peristyle and an atrium (Hauschild & Teichner, 2002; Lancha & Oliveira, 2013). The *pars rustica*, i.e. the part corresponding to the agricultural dependencies, stables

11 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/a-framework-supported-by-modeling-andvirtualaugmented-reality-for-the-preservation-and-dynamization-ofarcheological-historical-sites/192767?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian: www.igi-global.com/e-resources/library-recommendation/?id=132

# **Related Content**

Sustainable Entrepreneurship Development in the Accommodation Sector: The Case of the "Albergo Diffuso" (Diffused Hotel) in Italy

Angelo Presenza, Angelo A. Camillo and Isabell C. Camillo (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-17).* 

www.igi-global.com/article/sustainable-entrepreneurship-development-in-the-accommodation-sector/189741?camid=4v1a

Green Digital Enterprise and Chaos Theory in Tourism Development: The Case of Eco-Tourism in South Africa

Ezendu Ariwa and Carsten Martin Syvertsen (2015). *Hospitality, Travel, and Tourism: Concepts, Methodologies, Tools, and Applications (pp. 1474-1488).* 

www.igi-global.com/chapter/green-digital-enterprise-and-chaos-theory-in-tourism-development/119292?camid=4v1a

## Medical Tourism Service Quality: An Indian Perspective

Bikash Ranjan Debata, S. S. Mahapatra and Bhaswati Patnaik (2015). *New Business Opportunities in the Growing E-Tourism Industry (pp. 149-180).* 

www.igi-global.com/chapter/medical-tourism-service-quality/133031?camid=4v1a

The Effect of Room Service, Comfort, and Reservation Process on The Performance Dimension Evaluation: The Case of Grand Mercure Hotel Gajah Mada Jakarta

Adilla Anggraeni and Meyliza Thorina (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 42-52).* 

www.igi-global.com/article/the-effect-of-room-service-comfort-and-reservation-process-on-the-performance-dimension-evaluation/189744?camid=4v1a

# Architecture, Tourism, and Technological Innovation in the Jesuit Mission of São Miguel, Brazil

Pedro de Alcântara Bittencourt César

Universidade de Caxias do Sul, Brazil

Julia Luise Altmman

Universidade de Caxias do Sul, Brazil

Caroline Peccin da Silva

Universidade de Caxias do Sul, Brazil

**Bruna Tronca** 

Universidade de Caxias do Sul, Brazil

Michele Pinheiro Trentin

Universidade de Caxias do Sul, Brazil

# **ABSTRACT**

The contemporaneity demands a better relation between the object and the subject. In this article, it is expected to understand how has been developed this bond of Jesuit architecture and its understanding by the visitor. In this way, it is searched to explore the touristic technological means in Missões Guaraníticas. It questions the use of technology to a better extent of dissemination of local and cultural legacy - existing architecture. The methodology presents an exploratory research of technological platforms of information. It is noticed that its use increases tourism, favoring the culture and history of the region, besides promoting accessibility, reducing the social and physical barriers. The technology helps to preserve and disseminate the historical importance of the site. New information media has influenced and has increased the tourism, stimulating the local trade, and also it encourages the preservation of historical heritage. The approach of historical study of the region / country has great potential for increasing interest in historical architecture and culture.

# INTRODUCTION

Before presenting some determinants results of the relationship between technology and tourism as a feature of the architectural tourist attraction, an overview is introduced of this current environment. In it, ideas of a society as a holder of a traditional modernity has been surpassed by diverse values that conditions to a new world context. The subjects, men and women, distinctly, have transformed technological props in support of their daily lives.

Another aspect of this activity portrays Lash and Urry (1998, p.16) when referring to mobility. These social scientists are directed to the capitalist economy in postmodernity, characterizing its currently striking aspects. In this condition, this social factor is determined as the structure of objects and subjects. This value not only configures the recognition of products that are formed by movements between project, components and ideas which overcome local and national physical barriers, from its origin to the final use (consumption); but also of subjects who are increasingly in different transits and forming global networks of flows of people.

In this issue, points are established with more interest in being accessed, which in Critical Geography can be denominated as luminous points of world capitalism (Santos, 2008). It is known that for various interests often they stand out for having their appeal associated with major world themes, reworking values, such as cultural, by structures based on information and communication. "Thus, places are redefined: as points of interest of long and short-haul, global and locals, manifested by a range of classifications that are expanding and changing" (Santos, 2008, p.18).

The present moment, called by Lash and Urry (1998) of disorganized capitalism, presupposes "a predominance of the culture, global, local and a concern for the environment, all aspects that characterize the trip and the hospitality in contemporary era" (Lash & Urry, 1998, p.348). Previously these displacements were directed to instructions of specialized guides, diverse printed matter and with little universal access. Nowadays, the traveler hopes to absorb an enormous amount of artifact and information, cultures and sense systems (Lash & Urry, 1998, p.345). In this condition, he (the traveler) uses various technologies. Corroborates in this context, the positioning of the localities, often predetermined as a place of tourist interest, not by determinism but by capitalist reproduction's logic, and in many ways are positioned, like those qualified as a World Heritage Site by Unesco, offering a landscape among other expectations to the future visitor.

It is also important to note that "the study of the relationship between the digital and the social tends to be characterized by technological determinism or by indeterminacy" (Sassen, 2010, p.192). In these conditions, the use of technology can be seen in several ways, as an independent relationship to a performance relationship. However,

The digital and the social can shape and condition each other, but each one is, and remains, specific and distinct. And this interaction can occur in short or long chains, where a social result contributes with a new technical element and so on (Sassen, 2010, p.192).

It is also observed that the digital space confronts the dominant idea previously referred to it. Not always, necessarily, but in it adds an imminent value of contradiction in the places' affirmation. This panorama can be easily observed with the reading of Augé (1994). This anthropologist and researcher highlight the necessity to determine a status of the concept of non-place by associating it with various connotations.

15 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/architecture-tourism-and-technological-innovation-in-the-jesuit-mission-of-so-miguel-brazil/192768?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

The Role of the UNWTO in Visa Facilitation: The Diverse Impacts on Tourism Industries of China, Russia and Turkey

Ebru Tekin Bilbil (2017). *International Journal of Tourism and Hospitality Management in the Digital Age* (pp. 17-35).

www.igi-global.com/article/the-role-of-the-unwto-in-visa-facilitation/177120?camid=4v1a

An Assessment Methodology for Hotel Websites: Application to the Top 10 Cities Destination in the World

F. Javier Miranda, Sergio Rubio and Antonio Chamorro (2015). *Hospitality, Travel, and Tourism: Concepts, Methodologies, Tools, and Applications (pp. 284-302).* 

www.igi-global.com/chapter/an-assessment-methodology-for-hotel-websites/119221?camid=4v1a

Sex Tourism and the Internet: Information, Amplification, and Moral Panics

Jerry Eades (2007). Information and Communication Technologies in Support of the Tourism Industry (pp. 260-284).

www.igi-global.com/chapter/sex-tourism-internet/22533?camid=4v1a

Consumer Preferences and Key Aspects of Tourism and Hospitality Marketing on Island Destinations

Nikolaos Pappa (2015). Hospitality, Travel, and Tourism: Concepts, Methodologies, Tools, and Applications (pp. 904-923).

www.igi-global.com/chapter/consumer-preferences-and-key-aspects-of-tourism-and-hospitality-marketing-on-island-destinations/119257?camid=4v1a

# Chapter 12 Should Festival Be Smarter? ICT on Mass Events – The Case of the Exit Festival (Novi Sad, Serbia)

# Uglješa Stankov

University of Novi Sad, Serbia

# Vanja Pavluković

University of Novi Sad, Serbia

# Juan Miguel Alcántara-Pilar

University of Granada, Spain

# Marija Cimbaljević

University of Novi Sad, Serbia

# Tanja Armenski

University of Novi Sad, Serbia

# **ABSTRACT**

The trend of information and communication technologies (ICT) employment to enhance transactions and to deliver better experience to visitors of mass events has been growing increasingly popular over the years. The emergence of "smart tourism" agenda which highlights new, more sustainable ways of business management, experience enhancement and destination management also creates new opportunities for ICT employment in mass event. Thus, this chapter discusses existing ICT holding potential for smart approach employment on mass events. An additional case of Exit festival was used as an exploratory evidence to support the main idea of the chapter. Based on comprehensive literary review and additional information on visitors' familiarity with ICT gained from the Exit festival, recommendations for mass event managers are presented.

# INTRODUCTION

Various mass events are becoming increasingly prolific as communities and destinations attempt to differentiate themselves from others and appeal to visitors. With increased competition for visitors, mass events are constantly striving for innovation by diversifying their programming. A wide range of information and communication technologies (ICT) can be used to enhance the event experience (Halpenny, Salenieks, Song, Van Winkle, & MacKay, 2013). Research in event tourism is fundamentally changing due to an explosive growth of ICTs (Rubinstein, 2015; Lee, Boshnakova, & Goldblatt, 2016). The event industry is also being transformed by digital proliferation and business innovations where ICT plays a crucial role enhancing tourist experiences on the mass events and prove more opportunities for their management. In this chapter, the use of ICTs on mass events is examined to help planners, managers and operators understand the current state of technology employment. Particularly, event organizers, with their constrained budgets and limited human resources capacity, must make choices about embracing these technologies smartly. The emergence of "smart tourism" agenda which highlights new, more sustainable ways of business management, experience enhancement and destination management (Cannon Hunter, Chung, Gretzel, & Koo, 2015) also creates new opportunities for ICT employment in mass event. These choices must be made with full knowledge of what each event is about, who their audience is, and what resources are required to support the implementation of ICT (Halpenny, Salenieks, Song, Van Winkle, & MacKay, 2013).

Thus, the purpose of this chapter is to review the current state of ICT implementation on the mass events and to advocate for the smart tourism approach employment. The proof-a-concept approach of this paper is based on an extensive literary review. To gain access to users' familiarity with ICT, the chapter examines the additional representative study of Exit festival, one of the leading music festivals in Europe and in the world. Based on the comprehensive literary review and results of the exploratory case, basic recommendations for smart approach employment for mass event organizers are given.

The chapter is structured as follows. Section one points to basic characteristics of smart approach in ICT design with the focus on travel industry that, among first disciplines, openly employed this concept. Section two examines some of the existing and forthcoming examples of ICT on mass events. Section three presents the exploratory case of Exit festival. The final section identifies main recommendations, contribution and limitations of presented research.

# ICT ON MASS EVENTS: TOWARDS SMART TOURISM APPROACH

The review of the relevant literature showed that all ICT applications in this field can be roughly distinguished to two basic types according to its main purpose – visitors-centric applications and managerial ones (Catherwood & Van Kirk, 1992; Lee, Boshnakova, & Goldblatt, 2016). Visitor-centric applications are usually intended to serve various pre- and on-festival experiences enabling visitors' easier interaction, identification, transaction and enhancing the experience thought co-creation. For example, Tomorrowland's (Boom, Belgium) tickets come in the form of radio-frequency identification (RFID) wristbands that employ personal ticket information and facilitate a "cashless" environment. At the same time, wristbands allow wearers to connect with one another on social media sites and embedded LED lights can be remotely triggered by the festival, producing light shows from the crowd (Rubinstein, 2015). Managerial applications are usually concerned with visitors' management in the preparation phase, event

17 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage: www.igi-global.com/chapter/should-festival-be-smarter/192769?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

The Impact of Social Network on Italian Users' Behavioural Intention for the Choice of a Medical Tourist Destination

Francesca Di Virgilio, Angelo A. Camillo and Isabell C. Camillo (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 36-49).* 

www.igi-global.com/article/the-impact-of-social-network-on-italian-users-behavioural-intention-for-the-choice-of-a-medical-tourist-destination/177121?camid=4v1a

Impact of Advertising and Public Relations on Tourism Development in Da Nang, Vietnam

Xuan Tran, Minh Nguyen, Ha Kieu Tan Luu, Ny Ngo, My Tran, Ngoc Bich Nguyen and Hai Tran (2017). *Tourism and Opportunities for Economic Development in Asia (pp. 1-16).* 

www.igi-global.com/chapter/impact-of-advertising-and-public-relations-on-tourism-development-in-da-nang-vietnam/176287?camid=4v1a

How Do Online Reviews Affect Business Travelers' Accommodation Choices?: The Application of Theory of Reasoned Action (TRA)

Faranak Memarzadeh, Shane Blum and Charlie Adams (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 50-65).* 

www.igi-global.com/article/how-do-online-reviews-affect-business-travelers-accommodation-choices/177122?camid=4v1a

The Role of the UNWTO in Visa Facilitation: The Diverse Impacts on Tourism Industries of China, Russia and Turkey

Ebru Tekin Bilbil (2017). International Journal of Tourism and Hospitality Management in the Digital Age (pp. 17-35).

www.igi-global.com/article/the-role-of-the-unwto-in-visa-facilitation/177120?camid=4v1a

# Chapter 13 New Cultural Mediators, Cocreation, and the Cultural Consumption of Creative

Rui Miguel Ferreira Carvalho University of Aveiro, Portugal

**Tourism Experiences** 

Carlos Manuel Martins da Costa University of Aveiro, Portugal

Ana Maria Alves Pedro Ferreira University of Évora, Portugal

### **ABSTRACT**

Creative tourism has been a proficuous ground for the implementation of ICT's (Information and Communications Technology) strategies and the so-called creative industries. New cultural mediators are changing tourism consumption. Postmodern consumers have brought new perceptions to cocreation processes through user-generated content, eWOM (Electronic Word of Mouth), peer-to-peer exchange, collaborative economy, SoLoMo (social-local-mobile) tourists, among others. Booktubers and music fans/consumers in Chile constitute examples of diverse consumption through virtual and physical mediation, as capital construction enables social practices. In this exploratory study, the authors discuss how creative experiences are being consumed through cultural mediation made possible by technological developments. The creative tourism network website is analysed in the light of postmodern cultural mediation and capital construction.

# INTRODUCTION

Creative tourism can be understood as a postmodern type of tourism (Molina, 2016) where tourists engage with locals directly and are given the chance to develop their cultural capital (Bourdieu, 1984, 1986, 2010). ICT's and new technologies play an important role where new consumers act as cultural mediators influencing cultural and tourism consumption. New cultural mediators (Arriagada & Cruz, 2014; Jeffman, 2014), present themselves as being part of a specialized group of consumers which proactively influence consumption choices in several fields. In this exploratory study, two examples are analyzed in order to connect new approaches to cultural mediation in the sociology of culture and tourism. Booktubers and music consumer fans represent two important examples where cultural mediation exceeds virtual purposes. New ways of cultural capital creation are present as the establishment of new virtual and physical networks crucial for tourism and culture fruition, boosts social phenomena. The aim of this chapter is to discuss how creative tourism experiences are being consumed by a new cultural mediation made possible by technological developments. The creative tourism network website is analyzed (http://www.creativetourismnetwork.org) using content analysis (Carvalho, Costa, & Ferreira, 2015; Ratz, 2016). This exploratory research seeks to perceive how cultural mediation occurs in relation to this special niche tourism (Richards, 2016), how the website allows for social engagement and how creative destinations are been promoted in terms of cultural mediation processes. The chapter is organized in several sections. The Background is divided in three theoretical areas in which the paper is based. In the next section the main focus of the chapter treats in detail the creative tourism network website, followed by future research directions and conclusions sections.

# **BACKGROUND**

# Specialised Cultural Consumption in Postmodern Tourism

I am someone who can adapt to any cultural way-of-life, any fashion style, or any cultural good as I flexibly accumulate a broad range of diverse experiences. (Thompson, 2000, p. 130)

The statement above made by Thompson (2000), illustrates the versatile character of postmodern consumers where cultural fruition, mediation and symbolic capital are present. Postmodernism consists of the dissolution of borders, not only between high and low culture, but also between different cultural forms, such as tourism, art, education, photography, television, music, sports, shopping and architecture (Urry, 2002). "Architecture, the representational arts, and all branches of the humanities and the social fields have been energized (...) by the turn to things postmodern" (Thompson, 2000, p. 118). In short, the mass communication era has transformed the tourist gaze and many of the features of "postmodernity" have already prefigured in part on existing tourism practices (Urry, 2002). This also influences tourism outcomes as the new tourist is motivated by higher-level motivations, driven by post-materialistic values and consciously seeks the experiences important for him (McLeod, 2006).

The concept and its meaning has been discussed in different ways, for example for Giddens, (2002) the word "post-modernity" is often used as a synonym for "postmodernism" or post-industrial society". For the author, postmodernism is appropriated to refer to styles or movements in literature, painting and

18 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage: <a href="https://www.igi-global.com/chapter/new-cultural-mediators-cocreation-and-the-cultural-consumption-of-creative-tourism-experiences/192770?camid=4v1">www.igi-global.com/chapter/new-cultural-mediators-cocreation-and-the-cultural-consumption-of-creative-tourism-experiences/192770?camid=4v1</a>

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

Promoting America: How Do College-Age Millennial Travelers Perceive Terms for Branding the USA?

Lisa T. Fall and Charles A. Lubbers (2017). *Global Place Branding Campaigns across Cities, Regions, and Nations (pp. 270-286).* 

www.igi-global.com/chapter/promoting-america/160551?camid=4v1a

Studying the Impact of Egyptian Hotels' Websites Marketing on Customers' E-Satisfaction Nancy Awad and Sherif Saad Soliman (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 53-63).* 

www.igi-global.com/article/studying-the-impact-of-egyptian-hotels-websites-marketing-on-customers-e-satisfaction/189745?camid=4v1a

# From the Museums of Objects to the Virtual Museums: An Opportunity for Local Tourism Development1

Sonia Ferrari and Monica Gilli (2015). *New Business Opportunities in the Growing E-Tourism Industry (pp. 255-272).* 

www.igi-global.com/chapter/from-the-museums-of-objects-to-the-virtual-museums/133036?camid=4v1a

# Sustainable Entrepreneurship Development in the Accommodation Sector: The Case of the "Albergo Diffuso" (Diffused Hotel) in Italy

Angelo Presenza, Angelo A. Camillo and Isabell C. Camillo (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-17).* 

www.igi-global.com/article/sustainable-entrepreneurship-development-in-the-accommodation-sector/189741?camid=4v1a

# Chapter 14 "España Verde": Fourism Destination Image amo

# Tourism Destination Image among German Facebook Users

#### **Till Maximilian Schuh**

University of Algarve, Portugal

# **Dora Agapito**

University of Algarve, Portugal

#### Patrícia Pinto

University of Algarve, Portugal

#### **ABSTRACT**

This study aims at measuring the image of the tourism brand "España Verde" by using the social media platform Facebook. The ever-increasing competition within the tourism industry makes it vital for destinations to address their target group by tailored marketing measures. Furthermore social media channels are changing the interaction between Destination Management Organizations (DMOs) and customers. With the aim of exploring España Verde's image, a three components approach—cognitive, affective and conative was applied. Related to the measurement of the cognitive component, the study also focuses on España Verde's touristic potential ranging from a rich and diverse culture as well as a unique natural heritage to a suitable infrastructure for touristic purposes. Although the supply side seems to fit the target group's desires when choosing a vacation destination, the exploratory results reveal that there is currently only little awareness for España Verde on the German market.

# INTRODUCTION

The tourism brand España Verde was founded in 1989 and unites four Spanish autonomous communities - Galicia, Asturias, Cantabria and the Basque Country. The area ranges from the French border, along the northern Spanish coast until the western end of the Iberian Peninsula. The destination is characterized by a broad touristic supply and represents an alternative to the typical "sun and beach" vacations. The brand was founded in order to establish a strong brand image which clearly represents the characteristics

of the participating communities (Cano Fuentes, 2006). These autonomous communities consider tourism as an industry that supports mutually the economic development as well as the natural and cultural heritage of the region (Ivars Baidal, 2004).

Tourism destinations like España Verde are competing in an environment where it is vital to be differentiated from competition and to focus on unique selling points (Echtner & Ritchie, 1991). It is of increasing importance to attract the tourist's attention in order to be considered as a possible holiday location (Gartner, 1993). Hence, the destination image held by target markets is essential in order to build and manage a strong tourism brand (Bigné, Sánchez & Blas, 2009; Martín & Bosque, 2008). Additionally, new Information and Communication Technologies (ICTs) are deeply affecting the tourism industry and changing the interaction between destinations and tourists (Buhalis & Law, 2008). New ICTs, as for instance Social Media, provide a broad range of possibilities for destinations such as countries, regions and cities to address customers and to influence their decision - making - process by tailored marketing measures (Stepaniuk, 2015).

In the annual journal "España Hoy" (Spain Today) – published by the Ministry of Presidency – the Spanish State Secretariat for Communication gives a current overview of Spain including descriptions of the measures and decisions undertaken by the government such as the new policy for promoting tourism abroad as committed by the Spanish General Secretariat of Tourism TURESPAÑA. In Spain Today's version of 2015, the first measure of this policy aims to apply new ICTs as well as Social Media, due to the technological advances, which are also affecting the tourism industry. The policy regarding mature markets, as for instance Germany, is to increase the tourist's loyalty and to attract new demand segments demanding products towards art and culture, urban tourism, local gastronomy and nature (Ministry of the Presidency, 2015).

According to the ranking of international tourist arrivals by country of residence, as presented by the National Institute of Statistics (Instituto Nacional de Estadística, 2016), the incoming tourism from Germany to entire Spain amounts to 10,293,964 arrivals or about 15% of the total inbound tourism in 2015. Considering those numbers, Germany is ranked on the third position. Hence, the German market can be considered as a mature market. In view of the tourism supply of España Verde, as outlined at a later stage of this chapter, it becomes evident that its supply side is suitable for the attraction of new demand segments and the diversification of the array of products. Statistics, achieved from the tourism promotion board of Cantabria (CanTur – Sociedad Regional Cántabra de Promoción Turística) show that Germany was ranked on the fourth position regarding the number of tourists who visited España Verde in 2015 between January and September (IET Instituto de Estudios Turísticos, 2015)<sup>1</sup>, and can, therefore, be considered as a strategic market for this Spanish destination.

This study aims at analyzing the destination image of the tourism brand España Verde amongst the German market by using a Social Media channel. The results provide the groundwork for the development and application of suitable marketing measures for connecting the destination with its target audience.

This chapter follows the following structure: a) in the background section, a literature review on the destination image construct and related concepts is provided, as well as on the relationship between destination image and social media, which is followed by a summarized presentation of the tourism supply of España Verde; b) the section related to the methodology, presents the objectives and the research questions, followed by the design of the questionnaire, the process for data collection, and the summary of instruments for data analysis; c) in the section data analysis, the results for each of the research questions are depicted; d) in the discussion and conclusion section, the findings are discussed and the main contributions of this scientific work are presented.

26 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/espaa-verde/192771?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

# Related Content

# Sales Force Technology for the Hospitality Industry

Ruth Rios-Morales and John C. Crotts (2012). *Global Hospitality and Tourism Management Technologies* (pp. 136-150).

www.igi-global.com/chapter/sales-force-technology-hospitality-industry/56438?camid=4v1a

Sustainable Entrepreneurship Development in the Accommodation Sector: The Case of the "Albergo Diffuso" (Diffused Hotel) in Italy

Angelo Presenza, Angelo A. Camillo and Isabell C. Camillo (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-17).* 

www.igi-global.com/article/sustainable-entrepreneurship-development-in-the-accommodation-sector/189741?camid=4v1a

# Minimizing the Cost of Capital in Hotel Investments

Prashant Das and Gabrielle Bodenmann (2017). *Handbook of Research on Holistic Optimization Techniques in the Hospitality, Tourism, and Travel Industry (pp. 191-207).* 

www.igi-global.com/chapter/minimizing-the-cost-of-capital-in-hotel-investments/170958?camid=4v1a

Sustainable Entrepreneurship Development in the Accommodation Sector: The Case of the "Albergo Diffuso" (Diffused Hotel) in Italy

Angelo Presenza, Angelo A. Camillo and Isabell C. Camillo (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-17).* 

www.igi-global.com/article/sustainable-entrepreneurship-development-in-the-accommodation-sector/189741?camid=4v1a

# Gastronomy in Cultural Tourism:

# Use of Mobile Instant Messaging to Look for Gastronomic Recommendations

Maria Romero-Charneco Universidad de Málaga, Spain

Ana-María Casado-Molina Universidad de Málaga, Spain

**Pilar Alarcón-Urbistondo** Universidad de Málaga, Spain

# **ABSTRACT**

Cultural tourism consumers seek advice and recommendations from others before making buying decisions. The mobile instant messaging (MIM) boom could create a channel not only for social communication but also for information about cultural tourism and specifically the choice of restaurant. In this research we study which channels are most used by consumers when asking for gastronomic recommendations and, in particular, the factors behind the use of MIM as the preferred channels when obtaining tourist information. For this, an online survey was carried out with a sample of 393 individuals in Spain and descriptive statistical techniques were applied to the processing and analysis of the data. The results obtained show that: (1) its use is determined by its instantaneousness and permanent availability; and (2) people use it due to a lack of knowledge about the area.

#### INTRODUCTION

Tourism is undergoing major changes. One of these changes is that tourists are seeking destinations that allow them to enjoy different experiences (De la Rosa, 2003; Cracolici & Nijkamp, 2008). At the same time, cultural tourism has been transformed, adapting to the new demands of travellers (Guzmán-López & Jesús, 2011). This has resulted in it undergoing an evolution influenced by the conceptual changes

#### Gastronomy in Cultural Tourism

that have taken place in the area of cultural heritage, social changes and consequently by changes in demand, looking for new forms of cultural consumption (Ibarra, 2001).

Cultural tourism can still be defined as that which includes all aspects of the trip, allowing the tourist to learn about the history, heritage or current ways of living and thinking (Richards, 2001). However, the way this is experienced has changed. Cultural tourism had always prioritised the visual sense over the rest of the senses, turning the tourist into a mere observer of reality, a passive tourist. However, tourists have evolved, changing from a mere collector of photos to wanting to seek out experiences. This search by the tourist for trips offering new experiences entails the greater involvement of the other senses in order to understand the customs of the place (McCabe, 2014; Ricky & Scott, 2017).

Thus, today's traveller wants to have a full sensory experience and this includes taste. Through food, the cultural tourist obtains a greater understanding of the place they are visiting, beyond the simple observation mentioned previously (Cohen & Avieli 2004; Kivela & Crotts, 2006; Molz, 2007; Mak et al., 2012; Oliveira, 2007). In addition, it should be noted that gastronomy forms part of the cultural, social, environmental and economic history of places and their inhabitants, reflecting a particular lifestyle and being deeply rooted in the culture and tradition (Mitchell & Hall, 2006). In fact, one of the elements of cultural tourism is gastronomy, with strong cultural links between gastronomy and tourism (Ritzer, 1995; Richards, 2002; Roden, 2003; Dos Santos & Antonini, 2004; López-Guzmán & Sánchez-Cañizares, 2008; Sánchez-Cañizares & López-Guzmán, 2011). In addition, it is a factor that has traditionally been used by destinations to improve their image and create a differentiated positioning (Hjalager, 2010).

It is in this context that wine tourism, culinary tourism and olive oil tourism are becoming more important, with the gastronomic factor being either the main motivation or a key part of the cultural trip, rather than simply a necessary activity at the destination (Flavián & Fandos, 2011).

Therefore, gastronomy and culture are closely related. When tourists travel they do not just go to restaurants to satisfy their need for food. Fields (2002) identifies four types of gastronomic motivation: physical, cultural, interpersonal, and status and prestige. In this study we focus on cultural motivations. In these, the importance of food in the culture of a country is taken into account. According to Vinha (2004), food has always been a key element of the culture of each society and tourists increasingly perceive gastronomy as an important factor in better understanding the culture of a place. In this sense, gastronomic tourism is understood not only as the act of going to a restaurant and tasting the typical dishes, but also being aware of the rituals and habits associated with the local gastronomy and of the option to visit museums and other attractions on this topic.

In cultural tourism, and specifically in gastronomic tourism, the Internet plays a very important role (O'Connor 2011). In the Web 2.0 environment, users become the main channels for the dissemination of online information, generated in turn by them and allowing for electronic word of mouth dissemination, or e-word of mouth as it is known (*e-WOM*). This type of information is called User-Generated Content (UGC) (Burgess et al., 2009).

In addition, the buying decisions of tourists are influenced by the ratings and comments made by other gastronomic tourism consumers. Several authors (Borja et al., 2002; Borja & Gomis, 2009; O'Connor, 2011) argue that this type of tourist in particular is influenced during the pre-purchase stage, purchase stage and post-purchase stage, not only by offline recommendations from friends/family and traditional channels but also by the online environment: *word of mouth* of the blogs, websites such as *TripAdvisor* or social networks like *Facebook*. So when it comes to making a decision, the recommendations taken from the social media become a key element influencing the evaluation of the information on the tour-

24 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/gastronomy-in-culturaltourism/192772?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

## Related Content

## Role of ICT And Tourism in Economic Growth of India

Surender Kumar, Karuna Chauhan and Abhay Kumar Srivastava (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-16).* 

www.igi-global.com/article/role-of-ict-and-tourism-in-economic-growth-of-india/177119?camid=4v1a

## **Defining Place Image**

Candi Clouse and Ashutosh Dixit (2017). Strategic Place Branding Methodologies and Theory for Tourist Attraction (pp. 1-20).

www.igi-global.com/chapter/defining-place-image/163008?camid=4v1a

Local Community Attitude and Support Towards Tourism Development at Saint Martin Island, Bangladesh: Local Community Attitude and Support

Md. Abdul Hai and Md. Badsha Alamgir (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 32-41).* 

www.igi-global.com/article/local-community-attitude-and-support-towards-tourism-development-at-saint-martin-island-bangladesh/189743?camid=4v1a

## Key Planning Factors for Deploying Ubiquitous Wi-Fi Networks in Hotels

Galen R. Collins (2017). *International Journal of Tourism and Hospitality Management in the Digital Age* (pp. 18-31).

www.igi-global.com/article/key-planning-factors-for-deploying-ubiquitous-wi-fi-networks-in-hotels/189742?camid=4v1a

# Information Mycological Systems and Traditional Ecological Knowledge: The Case of Mycological Tourism in Central Mexico

## **Humberto Thomé-Ortiz**

Universidad Autónoma del Estado de México, Mexico

## **ABSTRACT**

Wild edible fungi are non-timber forest products that have great relevance for forest communities in central Mexico. Texcaltitlán is a rural community known for its traditional ecological knowledge on the use and identification of wild edible mushrooms. The aim of this work is to link Geographic Information Systems and Traditional Ecological Knowledge, in order to generate Mycological Information Systems. This is a qualitative, quantitative and exploratory research, which seeks to determine the usefulness of Geographic Information Systems (GIS) to systematize and locate mycological resources for use as a tourist attraction. The results show the existence of a wide variety of edible mushrooms in the region, along with a wide mycological traditional knowledge. Both aspects reflect the existence of unique natural and cultural features that can be the basis to build a unique tourism product in central Mexico. It is concluded that GIS are useful tools to build a multifunctional vision of mushrooms.

## INTRODUCTION

This chapter is part of a basic scientific research entitled "Evaluation of the recreational dimension of wild edible mushrooms, their socioeconomic interest and their perspectives of rural development", supported by the National Council of Science and Technology of Mexico. This project contains three central components: (i) a participatory approach integrating local stakeholders as promoters of mycological tourism; (ii) generating mycological information systems based on traditional ecological knowledge

DOI: 10.4018/978-1-5225-2927-9.ch016

and scientific knowledge; and (iii) proposing a strategy for the tourism management of mycological resources in the central Mexican plateau.

Mycological tourism consists of experimenting with the natural and cultural dimensions of wild edible mushrooms, through an articulated offer of goods and services that allow a recreational experience to be lived. Its main activities are the identification, harvesting and tasting of the mushrooms in close contact with the nature and mushroom picking communities. This tourist modality is part of the new trends in rural tourism that are characterized by their high specialization, focused on a product anchored to the territory.

Mycological tourism is an example of the valorization of forests as a tourist resource in contemporary societies. The growing importance of forest spaces as tourist scenarios is associated with the specific characteristics of these ecosystems. This value can be increased by modifying forest management practices; For example, by maintaining the abundance of certain local resources of special interest (Bostedt & Mattsson, 1995). However, tourism exploitation of forests also generates environmental, social and cultural risks (Kuvan, 2005). Aspect by which planning is a central theme.

In the case of Geographic Information Systems, the interconnections between tourism and new technologies not only reveal the location of tourist attractions in a specific territory. Rather, they allow the mobilization of local resources, converted into cultural goods, as a distinctive sign of contemporary tourist leisure (Hannam, Butler & Morris, 2014).

The objective of this chapter is to analyze the relationship between mycological tourism and Geographic Information Systems as a tool for tourism planning in rural areas. All this from a participatory approach in which traditional ecological knowledge serves as a source of information and a point of reference to guide policies for economic restructuring of rural areas.

For this the text is divided into six parts. After this introductory section the relationship between Geographic Information Systems and Mycological Tourism is presented. Later, the use of GIS in the planning of mycological tourism is explored as a tool for generating data that can be analyzed from an integral and participatory perspective for the determination of the tourist potential of the territory and its adequate spatial projection. Later, the case study in a forest community in central Mexico is presented, where the characteristics of the observation unit and the methodological design of the research are discussed. Next, the application of the Mycological Information System is presented, through the evaluation of local mycological resources and a proposal of paths for mycological tourism. Finally, the conclusions and some considerations for the future of the research are presented.

## **BACKGROUND**

## Geographic Information Systems Applied to Tourism

The use of local resources for tourism can produce the opportunity to generate processes of socio-economic transformation that help to improve the quality of life of rural inhabitants. However, the introduction of new non-agricultural rural activities such as tourism requires the development of systematized information platforms for the success of development strategies.

In the case of nature-based tourism, many of the resources in which the activity is focused can be documented through spatial information sources that indicate the quantity and characteristics of tourism resources, as well as the socio-economic impacts that can derive from their use as tourist capital.

14 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/information-mycological-systems-and-traditionalecological-knowledge/192773?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

## Related Content

Ways of Applying Destination Management Concepts to the Development of Internal Tourism in Uzbekistan

Bakhtiyor Navruz-Zoda and Nutfillo Ibragimov (2015). *International Tourism and Hospitality in the Digital Age (pp. 172-190).* 

www.igi-global.com/chapter/ways-of-applying-destination-management-concepts-to-the-development-of-internal-tourism-in-uzbekistan/130101?camid=4v1a

Sustainable Entrepreneurship Development in the Accommodation Sector: The Case of the "Albergo Diffuso" (Diffused Hotel) in Italy

Angelo Presenza, Angelo A. Camillo and Isabell C. Camillo (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-17).* 

www.igi-global.com/article/sustainable-entrepreneurship-development-in-the-accommodation-sector/189741?camid=4v1a

Gap Analysis and Infrastructure Requirement for Tourism Development in the State of West Bengal: Evidence from Bishnupur, West Bengal, India

Dillip Kumar Das and Nilanjan Ray (2015). *Hospitality, Travel, and Tourism: Concepts, Methodologies, Tools, and Applications (pp. 186-197).* 

www.igi-global.com/chapter/gap-analysis-and-infrastructure-requirement-for-tourism-development-in-the-state-of-west-bengal/119214?camid=4v1a

The Role of the UNWTO in Visa Facilitation: The Diverse Impacts on Tourism Industries of China, Russia and Turkey

Ebru Tekin Bilbil (2017). International Journal of Tourism and Hospitality Management in the Digital Age (pp. 17-35).

www.iqi-global.com/article/the-role-of-the-unwto-in-visa-facilitation/177120?camid=4v1a

## Chapter 17 Isle of the Dead: A Study of Trunyan Cemetery (Bali)

## Bintang Handayani

President University, Indonesia

## Maximiliano Emanuel Korstanje

University of Palermo, Argentina

## **ABSTRACT**

This essay aims to explain the phenomenon and effect on tourism of the Balinese cemetery in the village of Trunyan, where the dead are not buried. It is a narrative enquiry combined with critical theory largely grounded in the scholarship of dark tourism and communication theory, coupled with content analysis of the online community's reviews from the TripAdvisor website. The study indicates that (1) connectedness to death suggests the existence of spirituality needed by people, at the same time indicating understanding of mortality; (2) social connections developed as a result of visiting Trunyan cemetery not only bring self-awareness and awareness of others, revolving around intrapersonal communication about spirituality and interpersonal communication among members of the online community, but also illustrate the development of dark tourism and conceptualise the role of tourists in building authentic experience as the essence of a death site's brand image.

## INTRODUCTION

The expression "Lonely in a crowd" indicates how the quality of relationships matters, and signifies a much deeper existence from postmodern perspectives where people seek for feelings shared by others. This aspect represents the roots of reciprocity as it was studied by ethnographers in non-western cultures (Malinowski, 1926). In this context, feeling of this calibre are driven by the so called dark tourism sites (Biran, Poria, & Oren, 2011; Stone & Sharpley, 2008). Although gaining in popularity, dark sites wake up a profound ethical discussion in public opinion (Tzanelli 2016). While a family of theories understand that dark sites express an emotional arousal which links community with pastime, addressing some of the existential questions of human existence (Lennon & Foley 2000; Raine 2013, Biran & Hyde, 2013; Cohen 2011; Podoshen 2013; Stone 2011); others claim beyond the quest for death, underlies a

DOI: 10.4018/978-1-5225-2927-9.ch017

sentiment of morbid voyeurism which is politically manipulated to perpetuate the legitimacy of elite, even in contexts of disasters and trauma (Bowman & Pezzullo, 2009; Hartmann 2014; Tzanelli 2016; Korstanje 2016). Likely, the Trunyan cemetery in Bali offers a fertile ground to overcome the limitations dark tourism literature shows. Despite this cemetery attracts both types of segments, overseas and domestic tourists, it is difficult to frame this space into the model of Seaton and Lennon (2004), As P. Stone (2006) puts it, dark tourism takes a wider spectrum which oscillates from battlefronts memorials which draws much attention from media, to cemeteries, where any intrusion is seen as a clear violation to private life. In this respect, B. Heidelberg (2015) dissects the case of Amityville US to understand many factors are involved at time of adopting dark tourism as the first option. Many communities, even, are far from the needs of reminding the event, and opted to silence it in the dust of oblivion. Korstanje has gathered substantial evidence during his fieldwork in Cromañón, Buenos Aires (Argentina) to confirm some communities see in tourism as an activity that very well may corrupt their memory of deads (Korstanie 2015), while in other cases, dark sites include the colloquially named perilous places, houses of horror, fields of fatality, tours of torment, and themed Thanatos, are theoretically on the dark tourism spectrum (Sharpley & Stone, 2009; Stone, 2006; Bowman & Pezzullo, 2009). Death tourism, which is defined as activities visiting unique and macabre sites such as cemeteries, appears not only to encourage human contemplation on death of the self (Stone, 2012; Korstanje and George, 2015), but also to revolve around how the death sites (those visited by tourists) serve as a distraction from one's own mortality (Korstanje & George, 2015). Coupled with the conversation on social media (Sigala, Christou & Gretzel, 2012; Nikiforova, 2013) from a postmodern perspective, i.e. the phenomenon of an online community generating e-comments and e-word of mouth (e-WOM) (e.g. see Gretzel & Yoo, 2008; Yoo & Gretzel, 2008, 2010, 2011; Xiang and Gretzel, 2010), this phenomenon of death sites in particular and the surprisingly popular dark tourism in general, leads to question whether it is tourist-demand or attraction-driven (Stone, 2006; Wilkinson, 2010).

As this backdrop, the online comments into death sites in the online community not only indicate how the role of tourists in building shared values may (or may *not*) shape the brand image of death sites, but also signifies the intertwined relationships of spirituality and specific tourist typology from the postmodern perspective. This strengthens the conceptualisation of "secular tourists" that Korstanje and George (2015) coined in their study on philosophical issues in dark tourism. The secular tourist is considered as one who may seek for authentic life experience of social connectedness and the need for otherness, which in other domains of study is identified as human needs of spirituality (e.g. Krentzman, 2013). It also indicates the magnitude of good relationships which revolve around human interactions with death and lead to contemplation of self-awareness (i.e. to intrapersonal communication) and how the conversation about death in the online community (i.e. interpersonal communication) validate the sense of spirituality and its philosophical perspectives. For example, Krentzman (2013) examines the intertwined relationship of spirituality and religion, leading to how attributes such as belief, comfort, reflection, ethics, and awe fuel the content of the shared values under discussion. Arguably, it keeps people alive and gives validation to their lives, as well as serving self-awareness (e.g. Krentzman, 2013; Holt-Lunstad, Smith & Layton, 2010).

The structure of this essay is as follows. First, the literature of dark tourism, from the tourist's and postmodern perspectives, and the purpose of life revolving around spirituality and religion are reviewed. Next, the research gap, the research methodology and an overview of the study context are outlined. The findings of the content analysis revolve around the presentation of death and its communication patterns; and how the presentation of death and religiosity emerge as commodification and affect tour-

21 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/isle-of-the-dead/192774?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

## Related Content

The Effect of Room Service, Comfort, and Reservation Process on The Performance Dimension Evaluation: The Case of Grand Mercure Hotel Gajah Mada Jakarta

Adilla Anggraeni and Meyliza Thorina (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 42-52).* 

www.igi-global.com/article/the-effect-of-room-service-comfort-and-reservation-process-on-the-performance-dimension-evaluation/189744?camid=4v1a

The Effect of Room Service, Comfort, and Reservation Process on The Performance Dimension Evaluation: The Case of Grand Mercure Hotel Gajah Mada Jakarta

Adilla Anggraeni and Meyliza Thorina (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 42-52).* 

www.igi-global.com/article/the-effect-of-room-service-comfort-and-reservation-process-on-the-performance-dimension-evaluation/189744?camid=4v1a

Museum Communication: Towards a Framework for Resource-Holder Relations Management Alfonso Siano, Mario Siglioccolo, Carmela Tuccillo and Francesca Conte (2015). *Hospitality, Travel, and Tourism: Concepts, Methodologies, Tools, and Applications (pp. 1172-1200).*www.igi-global.com/chapter/museum-communication/119271?camid=4v1a

Delivering Hospitality through Franchising: The Case of Monginis, India

Christo F. V. Fernandes, Babu George and Ajit Mishra (2016). *Strategic Tools and Methods for Promoting Hospitality and Tourism Services (pp. 139-148).* 

www.igi-global.com/chapter/delivering-hospitality-through-franchising/143405?camid=4v1a

## Chapter 18

## Perspectives of the Adoption of Cloud Computing in the Tourism Sector

## Pedro R. Palos-Sanchez

University of Extremadura, Spain

## Marisol B. Correia

ESGHT, University of Algarve, Portugal & CEG-IST, Universidade de Lisboa, Portugal

## **ABSTRACT**

This chapter aims to expose the current situation of the adoption of cloud computing in companies in general and in enterprises of the tourism sector in particular. For this, a review of the literature has been carried out to establish the conceptual framework of technology and of the new economic model that underlies its adoption. Cloud computing is one of the technologies less known to many organizations and especially users, as it is a new technology based on the Internet, through which information is stored on servers, is provided as a service and on clients' demand. Therefore, the main theories of adoption that have been used to explain the different adoption models are presented, as well as, the different solutions that are being used in the tourist industry.

## INTRODUCTION

Information and Communication Technologies, also known as Information Technology (IT) have been widely adopted by the society, but especially by the enterprises. These technologies have been widely adopted under the concept of solutions that are provided through IT services. This solution provides effective low-cost communication tools for customers (Tan & Teo, 1998). This is something to be considered for the small and medium-sized enterprise (SME), which adopts IT in a gradual and precise way, not having the need to make large investments, distributing the effort for years.

The adoption of Internet-based IT has made it possible to develop the paradigm of everything as a service, where the user pays only for the use made of the service. These services can range from software applications (Application Service Provider or ASP and Software as a Services or SaaS) to systems

DOI: 10.4018/978-1-5225-2927-9.ch018

infrastructure (Infrastructure as a Service or IaaS). This formula is intuited as a good solution for small and medium enterprises SME, in the line of jumping the barriers that prevented them to take advantage of IT. In the following section, as well as throughout the whole chapter, this concept will be expanded.

Heart and Pliskin (2002) defined this concept as "eRent" of Information Systems (IS), made through the Internet and thanks to an ASP. They say that for SMEs, the IS through "eRental" could be an attractive solution to the expensive and complex acquisitions and implementations of traditional IT.

Johansson (2004) also states that the main reason for SMEs to adopt IT by contracting an ASP is that they allow full control cost and a lower cost when adopting and maintaining them. In spite of this, it assures that if this decision is examined in detail, it is observed that the customers do not emphasize these reasons, and it induces to think that the perspective of the cost is secondary for the SME.

In this way it identifies the three main reasons why ASP clients hire their services. The main reason is that the company is looking to outsource everything that is not basic competencies of its business, which most usually meet the IS. The other two reasons are the lack of qualified personnel and the overall strategy of the organization.

The adoption of IT in the tourism sector is evolving with changes of habits of the travelers, driven by the adaptation of IT as the Internet of Things (IoT), which consists of an automated connection through the Internet of the information generated by the devices and systems. IoT can be defined as a set of interconnected things over the Internet, which have the ability to measure, communicate and act all over the world. The key idea of the IoT is to obtain information about the environment to understand and control and act on it (Díaz, Martín, & Rubio, 2016). According to these authors, the cloud computing and IoT integration, known as Cloud of Things, solves such problems as IoT's limitations, data access, computing, data analysis, and can create new opportunities.

Another adoption of IT in the tourism sector is SoLoMo (Social, Local and Mobile), which reflects consumers' preferences for sharing content and habits through the social web (social purchase, recommendations, etc.), only to say some examples (Junta de Andalucía, 2012).

Thus, travelers or tourists increasingly use different devices connected to the internet to find information, locate offers, book all services related to travel or make purchases at destination, which has resulted in the explosion of travel agencies (Online Travel Agencies), internet portals that intermediate in the booking of tourism products at worldwide level, the sale of second-hand travel through Internet, offering huge discounts through online outlet portals, or the emergence of new typologies of solutions to optimize the management of these channels and the presence based on solutions such as the Central Reservation System, Channel Managers (Channel Manager System) or online communication systems (Online Reputation, Social Networks, etc.).

The aim of this chapter is to present the different models of adoption of cloud computing by companies and organizations and, in particular, by enterprises of the tourism sector. The description and features of cloud computing, the benefits and obstacles for the companies, the added value for adopting the cloud, as well as, some examples of open source solutions are presented. Finally, the characteristics of the solutions adopted by the enterprises of the tourism sector are explained and discussed.

This chapter is structured as follows. After this introduction, a brief review of the state of the art of the literature of cloud computing is developed in Section Theoretical Background. Then, in Section Cloud and Enterprise, the basic principles of adoption of cloud computing by enterprises in general are presented, while in Section Smart Tourism Destinations and Cloud Computing in the Tourism Sector, the adoption of cloud computing by the different subsectors of activity of the tourism sector is explained.

22 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/perspectives-of-the-adoption-of-cloud-computing-in-the-tourism-sector/192775?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

## Related Content

## Customer-Perceived Value of Medical Tourism

Eunhee Sung and Wilson Ozuem (2015). Current Issues and Emerging Trends in Medical Tourism (pp. 58-78).

www.igi-global.com/chapter/customer-perceived-value-of-medical-tourism/133637?camid=4v1a

## An Assessment Methodology for Hotel Websites: Application to the Top 10 Cities Destination in the World

F. Javier Miranda, Sergio Rubio and Antonio Chamorro (2015). *Hospitality, Travel, and Tourism: Concepts, Methodologies, Tools, and Applications (pp. 284-302).* 

www.igi-global.com/chapter/an-assessment-methodology-for-hotel-websites/119221?camid=4v1a

## Key Planning Factors for Deploying Ubiquitous Wi-Fi Networks in Hotels

Galen R. Collins (2017). *International Journal of Tourism and Hospitality Management in the Digital Age* (pp. 18-31).

www.igi-global.com/article/key-planning-factors-for-deploying-ubiquitous-wi-fi-networks-in-hotels/189742?camid=4v1a

## Role of ICT And Tourism in Economic Growth of India

Surender Kumar, Karuna Chauhan and Abhay Kumar Srivastava (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-16).* 

www.igi-global.com/article/role-of-ict-and-tourism-in-economic-growth-of-india/177119?camid=4v1a

# Chapter 19 Semantic Transformation and Use of Cultural Data: Record Once and Publish Many

## Feliz Ribeiro Gouveia

Universidade Fernando Pessoa, Portugal

## Sérgio Lira

Green Lines Institute for Sustainable Development, Portugal

## **ABSTRACT**

Cultural content and cultural assets are increasingly seen as invaluable drivers for education, entertainment, tourism, knowledge, and shared memory initiatives. As such, cultural information, as stored in cultural institutions such as museums, archives, and specialized libraries is increasingly demanded by several actors, from public policy, tourism, app developers and software companies, schools and the citizen at large. To satisfy the needs of such a large spectrum of consumers and end users, the information must be format independent, multi-lingual, multi-purpose, and make use of standards, norms and national and international recommendations. In this is paper we describe our past and current efforts to develop cultural information systems that can satisfy those requirements and those of the specialized public, such as scientists and curators. We describe an inventory and content management system and the conceptual and architectural choices that were made to allow its evolution, standards' compliance and multi-purpose use.

## INTRODUCTION

Cultural content and cultural assets are increasingly seen as invaluable drivers for education, entertainment, tourism, knowledge, and shared memory initiatives. As such, cultural information, as stored in cultural institutions such as museums, archives, and specialized libraries is increasingly demanded by several actors, from public policy, tourism, mobile app developers and software companies, schools and the citizen at large. To satisfy the needs of such a large spectrum of consumers and end users, the information must, as much as possible, be format independent, multi-lingual, multi-purpose, and make

DOI: 10.4018/978-1-5225-2927-9.ch019

use of standards, norms and national and international recommendations. As Libraries, Archives and Museums are arguably the oldest examples of knowledge based organization (Hedstrom & King, 2003) it is natural that they are at the forefront of the web evolution. Cultural institutions have been using several inventory management systems, and investing in cleaning, formatting and adapting data to be shown on public portals and mobile applications. Some institutions have been exporting their data, encoded in some metadata format, to aggregators with national or international dimension such as Europeana.

After data is published on the web there is a growing need for linking it from different places, themes, languages, and culture so that it can be part of the linked open data world, and be used to its full extent. It becomes virtually impossible for an institution to cope with so many requests for information, in different formats, encoding standards, languages and level of detail. Requests can come under the form of sharing in social networking sites, supplying data to content aggregators, providing programming interfaces to app developers and other end users, participating in joint portals, and virtual exhibits, for example. Additionally, published data should be correctly found by search engines, and made accessible to the general public. The use of right metadata, together with search engine optimization techniques can help disseminate a cultural institution's content. The web is an interlinked collection of resources, identified by what is called a URI (Uniform Resource Identifier). A kind of URI is what people generally type in a web browser to access a given site. To be able to make available their data so that it is found, either by searching or by looking from other data, institutions need to take a systematic approach to web publishing by providing as much context information as possible.

The so called semantic web is a web with context information, organized in a way to make search and navigation easy and accurate by providing only relevant information. The semantic web is designed to make search and discovery more efficient, and to navigate from one resource to the other as one would normally do in a web page. For example, when searching for the painter Picasso there would be irrelevant results, like those related to a car model with that same name, software packages or generic pages mentioning Picasso. Discovery is also an important behavior, as people would normally move from one page to the other by following links. The semantic web should keep and encourage that same behavior, mainly if computer programs are browsing the data as this is the only way they can move from one page to another.

The linked data movement proposed the Resource Description Framework (RDF) syntax to represent information to be exchanged (Bizer, Heath & Berners-Lee, 2009; W3C, 2014). There were several RDF vocabularies available then, and their number and availability continues to increase. Anyone can define and publish a RDF vocabulary, making it available such that other people can reuse it (see for example http://lov.okfn.org/dataset/lov/ for a database of more than 600 available RDF vocabularies).

Tim Berners-Lee (2006) summarized and defined the following characteristics for the semantic web, a web of linked data:

- Use URIs as names for things.
- Use HTTP URIs so that people can look up those names.
- When someone looks up a URI, provide useful information, using the standards (RDF, SPARQL).
- Include links to other URIs so that they can discover more things.

In 2010, Berners-Lee updated the semantic web document and defined a 5 star system to rate linked open data (LOD) projects on the web. Linked data is defined as above, and open means open license (Berners-Lee, 2006; http://5stardata.info/en/); item numbering is the number of stars:

17 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/semantic-transformation-and-use-of-cultural-data/192776?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian: www.igi-global.com/e-resources/library-recommendation/?id=132

## Related Content

## Role of ICT And Tourism in Economic Growth of India

Surender Kumar, Karuna Chauhan and Abhay Kumar Srivastava (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-16).* 

www.igi-global.com/article/role-of-ict-and-tourism-in-economic-growth-of-india/177119?camid=4v1a

Business Opportunity and Social Responsibility: Growing Importance of Accessible E-Tourism for Senior Tourists and Persons with Disabilities

Aleksandar M. Ivanovi (2015). New Business Opportunities in the Growing E-Tourism Industry (pp. 303-336).

www.igi-global.com/chapter/business-opportunity-and-social-responsibility/133041?camid=4v1a

## Sex Tourism and the Internet: Information, Amplification, and Moral Panics

Jerry Eades (2007). Information and Communication Technologies in Support of the Tourism Industry (pp. 260-284).

www.igi-global.com/chapter/sex-tourism-internet/22533?camid=4v1a

Sustainable Entrepreneurship Development in the Accommodation Sector: The Case of the "Albergo Diffuso" (Diffused Hotel) in Italy

Angelo Presenza, Angelo A. Camillo and Isabell C. Camillo (2017). *International Journal of Tourism and Hospitality Management in the Digital Age (pp. 1-17).* 

www.igi-global.com/article/sustainable-entrepreneurship-development-in-the-accommodation-sector/189741?camid=4v1a

## Chapter 20

# A Platform for the Promotion of Energy Efficiency and Monitoring in Hotel Units

## **Sergey Nogin**

University of Algarve, Portugal

## Jânio Monteiro

University of Algarve, Portugal & INESC-ID, Portugal

## Sergio Gómez Melgar

University of Huelva, Spain & LAR Arquitectura, Spain

## José Peyroteo

University of Algarve, Portugal

## António Mortal

University of Algarve, Portugal

## Carlos Miguel A. Santos

University of Algarve, Portugal

## José Livramento

University of Algarve, Portugal

## Pedro J. S. Cardoso

University of Algarve, Portugal

## Jorge Semião

University of Algarve, Portugal

## **ABSTRACT**

Tourists today are more likely to be concerned with the environment and greener lifestyle choices. In this context, a green flagship of some hotels can be an important selection criteria at the time of selecting one. In the near future, buildings should become nearly zero energy, consuming as low as possible

DOI: 10.4018/978-1-5225-2927-9.ch020

## A Platform for the Promotion of Energy Efficiency and Monitoring in Hotel Units

and producing almost all the energy they need, using renewable energy sources. To achieve this goal, hotel buildings need to pass through a transformation process that will make them more efficient. In this process, a decision support platform would be important to help hoteliers monitor their energy consumption, identify which points are consuming more than expected, decide which investments are more cost effective and manage their equipment in an optimum way. This chapter describes the challenges involved in developing such a platform, covering several research and development fields, including Internet of Things networks, ICT, Smart Grids, Renewable Energy, Energy efficiency, as well as algorithms for machine learning and optimization.

## INTRODUCTION

There has been a gradual shift in the expectations and demands of tourists, regarding hotels. Today's typical hotel guests are more likely to be concerned about environmental issues comprising greener lifestyle choices, such as recycling, fuel-efficient vehicles or organic food. While the location of hotels is still one of the most important features to select them, it is also common to find many hotels in the same region or area. In these cases, the green flagship of some of these hotels can be an important differentiating factor that make them more attractive to some tourists.

To achieve that green flagship, a better energy efficiency of hotels is required. In this field, the Directive 2010/31 of the European Union (European Parliament, council on the energy performance of buildings, 2010) makes mandatory for all new public buildings after January 2020 and all other new buildings from January 2021 onward to become nearly zero energy. This means that, they will have to consume as low as possible and at the same time will have to produce almost all the energy they need, in their own building or in their vicinity.

As tourism is one of the priorities for the economic activities of several European regions, the implications of the European Directive 2010/31 are of high importance for touristic companies and in particular for hotel units, affecting their competitiveness and image. Monitoring energy consumption to achieve the zero energy goal also helps to accomplish the sustainability of the region where the hotel is. By conveying the achievements obtained with these measures in future eTourism Applications, the image of the region as a whole can be improved.

In order for them to achieve the goals of zero energy building, a complete set of decision support tools is required that, besides monitoring consumption, could support the development of innovative methods to improve energy efficiency. Their aim is to give hoteliers and technicians a set of tools that include recommendations on building insulation, amount of renewable energy needed, automated detection and identification of excessive consumption levels, and management of consuming devices.

In some cases, as for instance in the detection and automatic identification of abnormal consumption levels, machine learning algorithms are required to compare the current level of consumption with the expected one, obtained from the consumption history of the building and weighted by several factors such as occupancy levels and environmental variables. Machine learning algorithms are thus used to perform a real time and continuous analysis, selecting the relevant information between huge amounts of data generated, e.g., by a network of IoT (Internet of Things) sensors and SCADA (Supervisory Control and Data Acquisition) devices.

27 more pages are available in the full version of this document, which may be purchased using the "Add to Cart" button on the product's webpage:

www.igi-global.com/chapter/a-platform-for-the-promotion-of-energy-efficiency-and-monitoring-in-hotel-units/192777?camid=4v1

This title is available in Advances in Hospitality, Tourism, and the Services Industry, InfoSci-Books, InfoSci-Business and Management, Business, Administration, and Management, InfoSci-Computer Science and Information Technology, InfoSci-Government and Law, Science, Engineering, and Information Technology. Recommend this product to your librarian:

www.igi-global.com/e-resources/library-recommendation/?id=132

## Related Content

The Role of the UNWTO in Visa Facilitation: The Diverse Impacts on Tourism Industries of China, Russia and Turkey

Ebru Tekin Bilbil (2017). *International Journal of Tourism and Hospitality Management in the Digital Age* (pp. 17-35).

www.igi-global.com/article/the-role-of-the-unwto-in-visa-facilitation/177120?camid=4v1a

Institutional Changes and Dynamics in the European Aviation Sector: Implications for Tourism Marina Efthymiou, Pavlos Arvanitis and Andreas Papatheodorou (2016). *Global Dynamics in Travel, Tourism, and Hospitality (pp. 41-57).* 

www.igi-global.com/chapter/institutional-changes-and-dynamics-in-the-european-aviation-sector/156748?camid=4v1a

Human Resource Management in Indian Hospitality Industry: Emerging Theoretical and Practical Aspects

Mohinder Chand Dhiman and Amit Kumar (2015). *International Tourism and Hospitality in the Digital Age* (pp. 259-275).

www.igi-global.com/chapter/human-resource-management-in-indian-hospitality-industry/130106?camid=4v1a

### Tourism Demand Forecasting Based on a Neuro-Fuzzy Model

George Atsalakis, Eleni Chnarogiannaki and Consantinos Zopounidis (2015). *Hospitality, Travel, and Tourism: Concepts, Methodologies, Tools, and Applications (pp. 106-115).* 

www.igi-global.com/chapter/tourism-demand-forecasting-based-on-a-neuro-fuzzy-model/119209?camid=4v1a

## **About the Contributors**

**João Rodrigues** graduated in Electrical Engineering in 1993, he got his M.Sc. in Computer Systems Engineering in 1998 and Ph.D. Electronics and Computer Engineering in 2008 from University of the Algarve, Portugal. He is Adjunct Professor at Instituto Superior de Engenharia, also in the University of the Algarve, where he lectures Computer Science and Computer Vision since 1994. He is member of associative laboratory LARSyS (ISR-Lisbon), CIAC and the Associations APRP, IAPR and ARTECH. He participated in 15 financed scientific projects, and he is co-author more than 150 scientific publications. His major research interests lies on computer and human vision, assistive technologies and human-computer interaction.

**Célia M. Q. Ramos** graduated in Computer Engineering from the University of Coimbra, obtained her Master in Electrical and Computers Engineering from the Higher Technical Institute, Lisbon University, and the PhD in Econometrics in the University of the Algarve (UALG), Faculty of Economics, Portugal. She is Adjunct Professor at School for Management, Hospitality and Tourism, also in the UALG, where she lectures computer science. Areas of research and special interest include conception and development of information systems, tourism information systems, big data, etourism, econometric modeling and panel-data models. Célia Ramos has published in the fields of information systems and tourism, namely, she has authored a book, two book chapters, conference papers and journal articles. At the level of applied research, she has participated in several funded projects.

**Pedro Cardoso** holds a PhD in the field of Operational Research from the University of Seville (Spain), a Master in Computational Mathematics from the University of Minho (Portugal) and a Degree in Mathematics - Computer Science from the University of Coimbra (Portugal). He teaches Computer Science and Mathematics at the Instituto Superior de Engenharia of the Universidade do Algarve (UAlg) and is member of LARSys/UAlg. He has high knowledge in the fields of databases, algorithms and data structures, and Operational Research. Over the past few years has been involved in 7 national and international scientific and development projects and is the co-author of about 50 scientific publications.

Cláudia Henriques has a PhD in Economics (Algarve University, 2002), thesis under the subject of "Management and Planning of Sustainable Tourism in cities and historic city quarters. The case of Lisbon". She is lecturer of Tourism Economics, Cultural and Creative Sector and Tourism Planning and Managing at Algarve University (ESGHT\_UALG) from 1993 until today. Her Research interests are in Cultural Tourism, Tourism Planning and Management; Cultural and Tourism Policies and Cultural heritage (Historic Centres). She is member of CIEO (Centre for Spatial and Organizational Dynamics).

\* \* \*

**Telmo Adão** attained his PhD in Computer Science at the University of Trás-os-Montes and Alto Douro, where he developed a procedural modelling methodology for the production of coarse traversable buildings composed by arbitrary shapes. During his journey as a researcher, he developed work not only in computer graphics but also in context-aware advertising, accessibility for blind people, usability for children, computer vision applied to the elderly, augmented reality in cultural heritage and human-computer interfaces for persons with disability. Currently, he is working in hyperspectral remote sensing for disease detection in agroforestry. Machine learning is also in his range of interests.

**Dora Agapito** holds a PhD in Tourism, a master in Marketing, and a bachelor degree in Communication Sciences. She is Assistant Professor at the Faculty of Economics of University of Algarve, Portugal, and an active member of the scientific committees for the Master of Tourist Organizations Management and the Master of Communication Design for Tourism and Culture in the same institution. Her research and publication interests are in tourism marketing, sensory tourist experiences, rural tourist experiences, sensory marketing, and social marketing. She is a member of the Research Centre for Spatial and Organizational Dynamics (CIEO).

**Juan Miguel Alcántara-Pilar** is Associate Professor of Marketing at University of Granada (Spain) at the Faculty of Education, Economy and Technology of Ceuta. His areas of specialization are marketing online, cross-cultural marketing, language effect on consumer behavior and behaviors in multi-racial communities. He has published several books as co-author on these matters (Springer, IGI Global). He has also published various peer-reviewed papers in prestigious journals such as Cross Cultural Management: An International Journal, International Journal of Advertising, European Research on Management and Business Economics and Computers in Human Behavior, among others. He has participated in prestigious international congress (EMAC, ICORIA, Academy of Marketing).

Pau Alonso-Monasterio is PhD by the Universitat Politècnica de València, where she has been developing her career as a researcher in tourism management of natural and cultural heritage sites, through more than 20 research and development international projects (ENPI, Interreg, Euromed Heritage, Azahar, etc.). She also has been combining her role as researcher with teaching in grade and postgraduate courses at university. Dr. Alonso-Monasterio is specialized in designing maps for visitors in areas with natural and cultural heritage, having published, among others, maps in World Heritage Sites such as Petra Archaeological Park (Jordan) and the Maya Site of Copán (Honduras), and in natural protected areas such as the Albufera Natural Park (Valencia, Spain), as well as thematic maps like the edited for Ramsar Convention 'Mediterranean Wetlands Cultural Landscapes'. She has contributed in Ramsar COP exhibitions, and has been member of the design team in the Visitors Center of Tous Dam (Turia River, Spain), the Visitor Center of Gandia Marsh (Spain) and the Visitor Center of Ghar el Melh (Tunis).

**Julia Altmman** is an Architecture Student at the University of Caxias do Sul. Scholarships student FAPERGS.

**Tanja Armenski**, PhD, is an Assistant Professor at the Department of Geography, Tourism and Hotel Management, Faculty of Sciences, University of Novi Sad, Serbia. Her research areas are destination competitiveness, consumer behaviour, cross-cultural studies and applied statistics. She has been a Departmental coordinator for international student mobility as well as the participant in the several projects of national and international importance. Tanja is currently in the Editorial Board of the Academica Turistica Journal and a member of the Canadian Association of Geographers and the Statistical Society of Canada.

João Pedro Bernardes is graduate in History (Variant in Archaeology) by the University of Coimbra (1985) and post-graduate (M Phil in1996, PhD in 2002), with a dissertation about the Roman Settlement in Leiria region (Portugal). He teached disciplines of Archaeology and Ancient History at the University of Azores. Since 1997 he teaches at University of Algarve disciplines as History of the Classical Culture and Classical Archaeology, where he is associate professor. He was director of the Department of History, Archaeology and Cultural Heritage of the University of Algarve. He directed archaeological excavations in the country, but over all in the region of Leiria and Algarve. He made missions of research in the German Archaeological Institute of Rome, in the Center Pierre Paris of the University of Bordéus III, or in the archaeological Institute of Southampton University. He was member of the National commission of evaluation of the research projects submitted for the National Plan of Archaeological Works in the year of 2005. He was member of History and archaeology board of Phd and Pos-doc grants evaluation for the Fundação para a Ciência e Tecnologia (Portuguese Foundation for the Science ant Technology - FCT). Currently, he is a member of the Panel for the evaluation of Multiannual Projects of Archaeology submitted to the Portuguese Ministry of culture. He has published papers in national and international journals, proceedings of scientific meetings and books, most of all on Roman Archaeology. He has developed and coordinated National Research Projects financed by the Portuguese Institutions programs and he was the project leader (coordinator) of the international project "Roman Mosaics of Andalusia and Algarve - MOSUDHIS" (program INTERREG III), and he was the principal investigator of the Portuguese partner of the projects "El mundo funerario romano en el sudoeste europeo - MORSUDE (i2Tep Programme) and "CROSUDHIS: Ciudades romanas del suroeste de Hispania" (Poctep Programme). Recently he coordinated a study entitled "Cultural Heritage and Territory" about the Portuguese Cultural Heritage Classified ordered by Government of Portugal (Secretary of State for Culture), under Portuguese Studies plan for the culture of the European strategic framework 2014-2020.

**Rui Carvalho** is a PhD Tourism Student at the University of Aveiro, Portugal. He completed his degree in Cultural and Tourism Management and MSc degree in Development of Cultural Tourism Products at the Polytechnic Institute of Tomar, Portugal. He is also a Member of the research unit on Competitiveness, Governance and Public Policies (GOVCOPP) of the University of Aveiro and is a Tourism Teacher at the Vocational School of Torres Novas, Portugal. His research interests include cultural and creative tourism, creative experiences, co-creation, tourism events, tourism consumption in postmodernity, sociology and anthropology in tourism and epistemology of tourism.

**Sergio Casas** has a degree in Computer Science and also a degree in Telecommunications Engineering - Telematics Specialty. He received his PhD in Computational Mathematics at the University of Valencia in 2014. He works as a post-doctoral researcher in the Robotics Institute of the University of Valencia, where he is also a part-time professor at the School of Engineering (ETSE).

Pedro César is an Architect and urban planner, Prof. Dr. at the University of Caxias do Sul - PPGTURH.

Marija Cimbaljević is PhD candidate at the Faculty of Science, University of Novi Sad. She received her MSc degree in tourism from the University of Novi Sad in 2014. Her research interests lie in the area of tourism and particularly area of smart tourism. Further, her research interests include information technology and geographic information systems (GIS). She is currently engaged as an teaching assistant at the Faculty of Sciences, Department of Geography, Tourism and Hotel Management in Novi Sad, Serbia.

Marisol B. Correia is a professor in the Information Technologies and Systems scientific area of the School of Management, Hospitality and Tourism of the University of Algarve and she is a member collaborator of the Centre for Management Studies of IST of the University of Lisbon. She holds a PhD in Electronics and Computer Engineering from the University of the Algarve, a masters in Electronics and Computer Engineering from the University of Lisbon and 5-year undergraduate degree in Informatics Engineering from the University of Coimbra. She is a scientific reviewer for some international journals and conferences and she participated in I&DT projects. She has presented papers at international conferences and has published scientific papers in several journals. Her current research interests include Information Technologies applied to Hospitality, Management, Tourism and Marketing, Cloud Computing, Websites Evaluation, Web Semantic, Business Intelligence and Evolutionary Computation.

Carlos Costa, PhD, is Full Professor at the University of Aveiro. He holds a Doctorate in Tourism from the University of Surrey (UK) and a degree in Regional and Urban Planning from the University of Aveiro. He is the Director of the Department of Tourism, Editor of the Tourism & Development Magazine, and Head of the Tourism research group of the Research Unit on Governance, Competitiveness and Public Policy (GOVCOPP). He is also the Technical-Scientific Director of the spin-off tourism company 'idtour-unique solutions'.

**J. M. Cotos** is Professor at Santiago de Compostela University (Spain) Researcher at IT Research Center (CITIUS, http://citius.usc.es).

Paula Odete Fernandes graduated in Business Administration, received MSc Degree in Management and a PhD in Economics and Management (2005). She is currently a Coordinator Professor of the Department of Economics and Management at the Polytechnic Institute of Bragança (IPB) -Portugal. She teaches several courses in the area of Management, since 1992. She is researcher and Scientific Coordinator of UNIAG (Applied Management Research Unit), and researcher of NECE (UBI). She participated in 8 international projects I&D and has more than 160 publications in peer-reviewed journals, book chapters and proceedings, and has several communications at international scientific conferences. She serves as a member of Program Board and Organizing Committees for some Scientific Committees of International Conferences and has collaborated as reviewer with several journals.

Ana Maria Ferreira, PhD, Associate Professor at the University of Évora, Portugal. Received his doctorate in tourism from the University of Aveiro. She has a degree in History and a MSc Degree in Art History from the Universidade Nova de Lisboa. She is a member of the Interdisciplinary Center of History, Culture and Societies (CIDEHUS), University of Évora. In addition to being a member of several scientific boards of scientific journals, she is a regional coordinator of the CREATOUR Research Project - Developing creative tourism destinations in small towns and rural areas. Among her research interests are: Tourism, Culture and Urban and Rural Regeneration, Cultural districts and clusters of tourist attractions. Colocation; Cultural and creative industries. Creative tourism companies, Cultural Tourism and Creative Tourism, Creative tourism experiences, Co-creation, Interorganizational Networks and Copetition.

Mauro Figueiredo has a PhD in computer science from the University of Salford, Manchester, in 2005. He was teacher at University of Coimbra from 1989 until 1996. Since 1996 he is at the Algarve University where he is adjunct professor. His research interests are in the use of information technologies for education, e-learning, b-learning, games and augmented reality. His PhD students are currently working with ebooks and augmented reality tools for e-learning. He is author of more than seventy international journal and conferences articles, book chapters and books and he collaborated and participated in different National projects. He is the international coordinator of the Erasmus+ project MILAGE: Interactive Mathematics by implementing a Blended-Learning model with Augmented Reality and Game books. He has several papers best awards. He already organized several international conferences. Most of the research is conducted at Univ. Algarve, at CIMA (Center of Marine and Environmental Research) and CIAC (Center for the Arts and Communication Research). CV: http://w3.ualg.pt/~mfiguei/cvmfuk.pdf ORCID: http://orcid.org/0000-0001-9394-4868.

Marco Freitas is a detail-oriented professional with more than five years of experience as a software developer since he graduated at the University of the Algarve in 2008, in Electric and Electronic Engineering. Skilled at programming in C# and operating in Visual Studio he also has two programming certifications in Java 7. He is capable of explaining complex software issues in easy-to-understand terms and has a problem solving creative approach. Follows standards, adheres to procedures, calculates risks and weighs all sides of a problem. As a team member, he is loyal and committed to his organization. open-minded, thorough and dependable person. He is introverted but friendly, relaxed, easygoing and funny when he feels comfortable around others. Likes to help, listen to music, read books and go to the cinema. Enjoys learning, traveling and trying new foods.

**Alexandra Gonçalves** is Regional Director of Culture of the Algarve since December 2013. Adjoint Professor at the University of the Algarve/School of Management, Hospitality and Tourism since 1997. Master in Cultural and Heritage Management and Doctor in Tourism.

**César Gonçalves** has a degree in Mechanical Engineering, Master in Industrial Engineering, with research focused on reliability and failure rate analysis and a PhD in Industrial Engineering with research in the maintenance performance measurement. He is a professor in mechanical engineering course at the Instituto Superior de Engenharia of Universidade do Algarve, in the subjects of Maintenance Management, Reliability, Drawing, Mechanical Technology, CAD and CAM. His research interests are maintenance management, BIM and augmented reality. He was a speaker in congresses and conferences focusing in maintenance issues namely connected with buildings. He is also author of international journals papers.

**Feliz Gouveia** graduated in Electrotechnical Engineering from the University of Porto in 1986. Earned a PhD in Computer Systems from the Université de Technologie de Compiègne in 1992. Participated in research and development projects for the European Space Agency and several european companies in the areas of knowledge management and object-oriented databases. From 1996 is associate professor at University Fernando Pessoa, Porto, where he participated in European projects in the area of elearning and presence research.

**Bintang Handayani**, SIK., M.Si., MBA., Ph.D., is a Senior Researcher at President University, Indonesia. Her work scrutinises personification and impression management of a nation and its market offerings in consumer cultures. It tackles nation-brand images (NBI), Tourism and Hospitality attributes, Dark Sites phenomenon, Nation's Positioning, Place Attachment from Socio-cultural and Post-modern perspectives, coupled with Communication & Media studies.

**Nuria Huete Alcocer** is an Assistant Professor of Econometrics since October 2013 in the area of Spanish and International Economics, Econometrics, History, and Economic Institutions at the Faculty of Economics and Business, Albacete, University of Castilla-La Mancha. Research interests include: Tourism Marketing and Quantitative Methods in Economics and Business Tourism.

**Serkan Kılıç**, born in 1981, graduated in Business Administration in Bursa, Turkey. He received the Master degree and the PhD degree from the Uludağ University. He is currently working as an Assistant Prof. in the Department of Business Administration at Uludağ University in Turkey. He teaches marketing and innovation related courses at Uludağ University. His research interests are marketing, retailing, branding, consumer behavior and innovation. Dr. Kılıç has written three books named as Private labels in Retailing, Innovation Management and New Product Innovation. He has many published articles in different journals. He has also worked on internationalization and generation related projects as an academician and project manager.

Maximiliano Korstanje is editor in chief of International Journal of Safety and Security in Tourism (UP Argentina) and International Journal of Cyber Warfare and Terrorism (IGI-GlobalUS). Besides being Senior Researchers in the Department of Economics at University of Palermo, Argentina, he is a global affiliate of Tourism Crisis Management Institute (University of Florida US), Centre for Ethnicity and Racism Studies (University of Leeds), The Forge (University of Lancaster and University of Leeds UK) and The International Society for Philosopher, hosted in Sheffield UK. With more than 700 published papers and 25 books, Korstanje was awarded as Outstanding Reviewer 2012. International Journal of Disaster Resilience in the Built Environment. University of Salford, UK, Outstanding Reviewer 2013, Journal of Place Management and Development, Institute of Place, Manchester Metropolitan University, UK and Reviewer Certificate of Acknowledgement 2014, International Journal of Contemporary Hospitality Management (IJCHM), University of Central Florida, US. Now he co-edits almost 10 specialized journals in such themes as human rights, mobility, tourism and terrorism. Korstanje is subject to biographical records for Marquis Who's Who in the World since 2009. He had nominated to 5 honorary doctorates for his contribution in the study of the effects of terrorism in tourism. In 2015 he was awarded as Visiting Research Fellow at School of Sociology and Social Policy, University of Leeds, UK and the University of La Habana Cuba. In 2017 is elected as Foreign Faculty Member of AMIT, Mexican Academy in the study of Tourism, which is the most prominent institutions dedicated to tourism research in Mexico.

**Víctor-Raúl López Ruiz** is Professor of Econometrics at Castilla-La Mancha University. His fields of research are Intellectual Capital Measurement, International Growth, Regional Economy and Models Applied to Management. He is the author of several working papers, books and documents on these topics. He has taught international courses on Intellectual Capital and has also been awarded several research prizes. Further information available at http://victorraullopez.blogspot.com/.

Luís Gonzaga Magalhães has a BSc and MSc in Informatics from the University of Minho and a PhD in Computer Science from the University of Trás-os-Montes e Alto Douro. Currently, he is an Assistant Professor with Habilitation at University of Minho. Recently, he assumed the role of Scientific Coordinator of the Computer Vision, Interaction and Graphics unit at the Computer Graphics Centre. He is also a Senior Researcher at ALGORITMI Centre. He is author or co-author of more than 70 scientific publications on international journals and conference proceedings. His research interests include the use of Computer Vision, Augmented Reality, Computer Graphics, and more recently on the use of procedural modeling techniques. He is or was a research member in projects related to Expeditious Modeling of Virtual Environments, Immersive 3D Environments, Virtual Environments for Education, Virtual Heritage and High Dynamic Range Imaging. More recently was the scientific leader of a I&DT project with the industry which aimed to develop a mixed reality system for cultural and entertainment industries.

María Pilar Martínez-Ruiz, PhD, is Associate Professor of Marketing at Marketing Department, University of Castilla-La Mancha (Spain). Visiting Professor at University of Edinburgh (UK) and University of Plymouth (UK). Her research interests fall in the areas of consumer behaviour, retailing, marketing communications, tourism, innovation and ICT. Professor Martínez-Ruiz has published several articles in different high standing international Journals, such as Tourism Management, Service Industries Journal, Service Business, European Journal of Marketing, Journal of Retailing and Consumer Services, Journal of the Operational Research Society, International Journal of Market Research, Total Quality Management and Business Excellence, Decision Support Systems and Electronic Commerce, European Journal of Operational Research, among others. She was co-track chair of the Pricing and Retailing Track at the 2012 AMA Summer Marketing Educators' Conference. She is author of several chapters and books on these areas, and serve as a reviewer and associated editor of several international journals.

**Sergio Gómez Melgar** is a researcher and associate professor at the University of Huelva, specializing in energy efficiency in buildings. Sergio is the responsable researcher of the Energy Efficiency project at TEP192 Control and Robotics Research Group in the University of Huelva. This research pertains to instrumentation and data analysis for improving energy efficiency in buildings and cities, with specific application in passive design for energy efficiency, monitoring and control, and integration of renewable energy in buildings and cities. Sergio Gomez Melgar is also LAR Architecture's CEO. In his leadership position at the firm, he has designed more than 600,000 sq. feet of buildings of diverse typologies including residential, educational, social, healthcare, administrative and physical rehabilitation facilities for both private and publics sector clients. Some of his constructed works such as Zaranda House and Engineers University School in Huelva, have received international energy efficiency awards (Viena 2012, Gijon 2013 and Dublin 2014).

Jânio Monteiro graduated in Electrical and Computers Engineering in 1995 from the University of Porto, and later concluded a Master and PhD degrees respectively in 2003 and 2010, also in Electrical and Computer Engineering from Instituto Superior Técnico, Technical University of Lisbon. Since 2003, he is a member INESC-ID Lisbon. He participated in four European R&TD projects funded by the European Commission, and six national projects, with local companies, coordinating two of them. He is one of the editors of one book and of more than four dozen publications, including journal papers, publications in scientific conferences, book chapters and national patents. His main areas of interest involve communication networks, smart grids, sensor networks, human-machine interfaces and Internet of Things. He is currently an Assistant Professor of the Electrical Engineering Department of Instituto Superior de Engenharia, University of the Algarve where he teaches several courses related with "Computer Networks", "Sensor Networks" and "Smart Grids".

António Mortal is Assistant Professor in Mechanical Engineering Department of University of Algarve / ISE. Since 1992 he has taught subjects related to the energy, renewable energy and energy efficiency in undergraduate and graduate level. He also develops consultant and research activities in this area. Currently his main line of activity is related with energy in buildings, HVAC systems, energy audits, indoor air quality and with the application of the EPBD directive in Portugal. He also has experience using modelling tools for energy simulation in buildings as DesignBuilder software. He has experience in organizational coordination and leadership, actually he is the head of the Engineering School (ISE/UALG). He has more than 20 articles in scientific conferences.

Aristotelis Naniopoulos is a professor in Transport Systems at the Faculty of Engineering, Aristotle University of Thessaloniki (AUTh). He is currently Director of the Transport Systems Research Group (www.tsrg.gr) and the Laboratory of Urban and Regional Planning and Development (UREDLab) of AUTh. He graduated the same school from both the departments of Surveying Engineering (1976) and the Architecture Engineering (1981) from where he holds the two corresponding degrees. He pursued postgraduate studies in Operational Research at Cranfield Institute of Technology of U.K (1976), where he obtained the M.Sc. Degree. He also obtained the Ph.D. degree in Transport (1988). Prof. Naniopoulos has participated in more than 50 national and international projects, having acted as a coordinator in many of them.

**Sergey Nogin** completed the B.Sc. in Electrical and Electronics Engineering in 2015, from ISE, University of Algarve. He is now a student of the M.Sc. degree of the same institution.

Miguel Otón is a Professor at Santiago de Compostela University.

**Erkan Özdemir** received the Master degree and the PhD degree from the Uludağ University in Turkey. He is currently working as an Assoc. Prof. in the Department of Business Administration at Uludağ University in Turkey. He teaches marketing management, sales management, marketing strategies courses at Uludağ University. His research interests are marketing, consumer behavior, high-tech marketing and innovation. Dr. Özdemir has written three books named as Marketing Strategies to Women Consumers,

Marketing Ethics, Marketing of High-Tech Products and Services. He has many published articles in different journals. He also served as an assistant researcher at Uludağ University Technology Transfer Office Project supported by The Scientific and Technological Research Council of Turkey (TÜBİTAK) between 2014 and 2015 and has been in charge of business development and entrepreneurship department at Uludağ University Technology Transfer Office.

**Pedro Palos** has a PhD in Business Economics from the University of Seville, Bachelor in Information Systems Engineering and Bachelor in Research and Market Techniques. More than 20 years working in companies of the sector of Information Technology in different management positions. He is Associate Professor in the Department of Business Management and Sociology (Marketing and Research and Markets Area), teaching at the Faculty of Business Studies and Tourism of the University of Extremadura (Spain). He has participated with various lectures, courses and articles in various congresses and programs on digital economy, entrepreneurship and management both nationally and internationally, being his research lines Digital Marketing and Information Systems. He is a member of the UEx MARKETUR Research Group Research Group.

Vanja Pavluković is Associate Professor of Tourism at the University of Novi Sad Faculty of Sciences, Department of Geography, Tourism and Hotel Management. Her research interests focus on destination competitiveness, business travel and tourism, travel agency and tour operator management and event management. She has also published various peer-reviewed papers in international journals such as Current Issues in Tourism, International Journal of Tourism Research, Acta Oeconomica, European Journal of Tourism Research and Tourism and Hospitality Management.

**João Pereira** received his Bachelor's Degree in Electrical and Electronic Engineering in 2012 and is now finishing his Master also in Electrical and Electronic Engineering at University of the Algarve. He is a Researcher in the Associative Laboratory LARSyS (ISR-LX) and in the University of the Algarve. His major interests lie on programming, learning and exploring new frameworks.

**Emanuel Peres** is an assistant professor at Trás-os-Montes e Alto Douro University (UTAD) in Portugal. He has been lecturing in areas such as web development, electronics, power electronics, telecommunications and computer networks since 2003. He is an integrated researcher at the Technology and Science Associate Laboratory (INESC-TEC) and his research focuses on combining sensing networks, in-field processing units and mesh communication networks to develop data acquisition systems which enable decision support tools for precision agriculture. Human–Machine interaction, namely augmented reality systems, new interfaces and mobile devices, in areas such as education, agriculture and tourism, is also an area of interest. From 2010 onwards he has been involved in management positions at UTAD. He is an Editor-in-Chief of the International Journal of Web Portals (IJWP) since January 2016. As a member of UTAD's Urban Eco-Efficiency Unit, he is currently involved in several research projects mainly related with the application of UAV in agriculture and forestry.

**José Peyroteo** completed a B.Sc. degree in Electrical and Electronics Engineering in 2015, from ISE, University of Algarve. He is a currently a student of the M.Sc. degree of the same institution.

**Patrícia Pinto** holds a PhD in applied statistics to economics and business. She is Professor at the Faculty of Economics, University of Algarve, and an active member of board of the PhD program in Tourism in the same institution. Her current research interests are in applied statistics and modeling in the areas of integrated quality management in tourism destinations, events management, tourism destinations image, branding, marketing strategies, customer satisfaction, and environment behavior. She is a member of the Research Centre for Spatial and Organizational Dynamics (CIEO).

Cristina Portales Ricart is MSc. in Geodesy and Cartography from the Universidad Politécnica de Valencia and MSc. in Surveying and Geoinformation from Technische Universität Wien (2003). Research fellow at the Mixed Reality Laboratory of the University of Nottingham (2005) and at the Interaction and Entertainment Research Centre of the Nanyang University of Singapore (2006), both in the area of computer vision. In 2008 she received the Ph.D. degree by the UPV. From 2011-2012 she worked at AIDO (Industrial Association of Optics, Colour and Imaging), where she was primarily involved in technical management of the FP7 funded project SYDDARTA. In 2012 she received the Juan de la Cierva post-doctoral grant by the Spanish Ministry to join IRTIC (Research Institute of Robotics and Information and Communication Technologies), where she continued researching in accurate 3D reconstruction. Her current research interests are focused on geometric calibration, image processing, 3D reconstruction, multispectral imaging, augmented reality and human-computer-interaction.

**José Rodrigues** is adjunct professor teaching at the Geomatics Node of the Department of Civil Engineering, and he's also a researcher at the Marine and Environmental Research Center (CIMA). He graduated in Mathematics from the University of Lisbon, Portugal, and obtained his PhD degree in Computational Geometry from the University of Seville, Spain. His research interests are in Computational Geometry, Geographic Information Systems and Web3D. Over the years he taught courses in the Geomatics field such as geographical informations systems, mathematical cartography, photogrammetry, GIS programming and 3D Data Acquisition and Modelling, at undergraduate and graduate level. He is board member of the Scientific Commission of the Master in Geomatics. Since 2015, he is Vice Director of the Engineering Institute of the University of Algarve.

**Carlos Santos** concluded the B.Sc. Degree (five-year program) in Mechanical Engineering, from University of Algarve, in 1997. In 2013 he also concluded a Master of Science degree in Energy and Building Air Conditioning, from University of Algarve.

**João D. P. Sardo** was born in Faro, Portugal in 1989. He graduated in electrical and electronics engineering in 2011 and received the post-graduation in 2013, both from the University of the Algarve. In 2011 he co-founded eLab Hackerspace, a community oriented laboratory. An electronics, digital fabrication and open source enthusiast.

**Till Maximilian Schuh** holds a master in Tourism Economics and Regional Development and a bachelor in Sustainable Business Management. His master thesis was developed under the theme of tourism destination image and it was presented in several international conferences. His research and publication interests are in tourism marketing, sustainable tourism and rural tourism. Since August 2016, he works at the Thomas Cook AG in Oberursel, Germany as a business assistant.

**Jorge Semião** received the B.Sc. (five-year program), the M.Sc. and the Ph.D. degrees in electrical engineering from Instituto Superior Técnico, Technical University of Lisbon, Portugal, respectively in 1996, 2001 and 2010, with specialization in Electronic and Computer Systems, and Microelectronics. He joined University of the Algarve in 1997, and he is currently an Assistant Professor and the Director of the Electrical Engineering Department of Instituto Superior de Engenharia, University of the Algarve, in Faro, Portugal. He is also a senior researcher in the Quality, Test and Co-Design of HW/SW Systems research group of Instituto de Engenharia de Sistemas e Computadores R&D in Lisbon (INESC-ID), Portugal. His research interests include CMOS aging, dynamic voltage and frequency scaling, reliability, fault-tolerance and fault-detection. He participated in several research projects, and he is the author and co-author of over 80 scientific articles published, three national patents and three patents still under patent pending.

Caroline Silva is an architecture student at the University of Caxias do Sul.

**Uglješa Stankov** is Associate Professor of Tourism at the University of Novi Sad Faculty of Sciences, Department of Geography, Tourism and Hotel Management. His main research interests are IT applications in tourism, e-marketing and geoinfromation technologies. For this purpose, he actively cooperates with the researchers and professional organizations and participates in several international projects. He has also published various peer-reviewed papers in international journals such as Acta Oeconomica, Open Geosciences, European Journal of Tourism Research and Tourism and Hospitality Management.

**Humberto Thomé-Ortiz** is a doctor in Rural Sciences by the Autonomous University of Chapingo. Full time faculty at the Institute for Agricultural and Rural Sciences (ICAR) of the Autonomous University of the State of Mexico. Member of the National System of Researchers, Level I. His research areas are Local Agro-food Systems, Territorial Development, and Agro-food Tourism. He is principal investigator of the Project "Agro-food tourism in the State of Mexico and its perspectives for rural development" funded by the Mexican National Council for Science and Technology. His most recent publication is "Native maizes in feminised campesino strategies in the face of climate change".

**Michele Trentin** is a Master Student in Tourism and Hospitality from the University of Caxias do Sul (UCS).

**Bruna Tronca** is an architect graduated from the University of Caxias do Sul (UCS), Master Student in Tourism and Hospitality, Scholarships student CAPES.

**Panagiotis Tsalis** is a civil engineer specialized in accessibility issues. His PhD thesis concerns the development of a standard for the assessment and improvement of accessibility for persons with restricted mobility at municipal level. Dr. Tsalis has participated in various European and national research projects concerning accessibility of people with disabilities and people with restricted mobility in general. Furthermore, he has participated in several conferences and scientific publications with papers and presentations on this particular subject, and has given lectures on accessibility issues at the Aristotle University of Thessaloniki and the University of Thessaly. Finally, he has acted as instructor on accessibility issues for the Greek National Confederation of Persons with Disability.

Roberto Ivo Fernandes Vaz graduated in Electronic and Telecommunications Engineering (2012) at University of Aveiro – Portugal, and has an MSc Degree in Interactive Multimedia Communication (2014) by the same University. Currently, he's a Ph.D. student at UT AustinlPortugal Digital Media Doctoral Program. He participated in one international project I&D in Brazil (between the Federal University of Minas Gerais and the MM Gerdau Museum), has experience designing museographic projects and researches in the areas of human-computer interaction, digital media, interaction design, technology in museums and technological accessibility.

Ana Veiga graduated in Architecture and Urbanism (Federal University of Minas Gerais – UFMG, 2001), specialized in Construction Technology and Productivity (UFMG, 2002), M. A. in Architecture and Urbanism (UFMG, 2005) and D.Sc. in Image Art and Technology (UFMG – School of Fine Arts, 2012), with her thesis published as a book: "Gestão de projetos de museus e exposições" (Museums and Exhibits Project Management) by Editora C/Arte, FAPEMIG, 2013. Professor at the Federal University of Minas Gerais (UFMG), Department of Technology at the School of Architecture. Experience, projects and research in the following subject areas: Project Management Software Applications, Telework, Web (web design, web writing, UX, SEO, WordPress), Cultural Writing, Museums and Exhibits, Drawing and Scientific Illustration. Coordinator of the GRAFT research group and its virtual laboratory (LavGRAFT). Member of UNIC (Brazilian National Union of Scientific Illustrators).

María José Viñals is full Professor at the Higher School of Architecture in the Universitat Politècnica de València (UPV), Spain. She took a MA and a PhD in Geography at the Universitat de València-Estudi General (Spain). Dr. Viñals was the Director of the UPV International Office (2005-09) and director of the Spanish Wetland Centre (Fundación Biodiversidad, Spanish Ministry of Environment) during four years. Her field of expertise focuses on Natural and Cultural Heritage Planning and Management, Public Use and Conservation of Natural Protected Areas and Heritage Sites, Tourism and Heritage Interpretation. She usually teaches in undergraduate and graduate Environmental Sciences, Tourism and Architecture studies. Furthermore, she has been the director of several professional masters and courses in Spain and other countries (Italy, Portugal, Morocco, Tunisia, Australia, Nicaragua, Argentina, Uruguay, Jordan, Ecuador, Honduras, etc.). She has mentored and supervised the research work over a hundred of graduate students and has directed eight doctoral theses. Dr. Viñals has led interdisciplinary teams, that have conducted research in various aspects of protected areas and world heritage sites, particularly addressing issues related to the outdoor recreation planning and managing, and natural and cultural tourism resources. She has been also consultant to the Spanish Ministry of Environment, International Spanish Agency for Cooperation and Development (AECID), to the International Ramsar Convention on Wetlands, UNESCO Word Heritage Centre, to the UN World Tourism Organization, and advisor to World Wildlife Fund (WWF). Dr. Viñals has published more than a hundred fifty international research works (papers, books, conference papers, book chapters, etc.) devoted particularly to heritage and tourism management. Currently, she is an evaluator of the National Agency for Assessment and Accreditation (ANECA) of the Spanish Ministry of Education.

## Index

3D Printing 44-45 B Barcelos 120-122, 124-126, 128, 130-132, 135-136, A 139 Accessibility 1-2, 5-6, 10, 13, 21-22, 30-32, 43-44, Beacon 40-41, 45 46, 61, 80, 99, 102, 191-198, 203, 205-211, 214, Biometric Identification Technologies 262 228, 318, 327, 342, 358, 362, 364-365, 367, 369 Biometric technologies 248 Accessibility in Museums 32 Building Information Modeling (BIM) 216, 219, 226 Accessible Tourism 101, 119, 193 Business Intelligence 5, 15, 20-21, 29, 393 Acquisition 2, 5-6, 13, 33, 58, 65, 72-75, 78-79, 81, Byzantine Monuments 191-192, 195-196, 205, 214 84-85, 143, 145, 147, 152, 273, 314, 421, 448 Adaptive User Interfaces 97, 119  $\mathbf{C}$ Altiplano 353 App 36-37, 39-41, 44, 64, 97-98, 100-104, 108-109, Central Archaeological Council 194, 203, 206, 208, 214 113, 126, 161, 223, 240, 243, 250, 401-402, 434 Central Mexico 338-339, 341, 344, 350, 353 Chatbots 314, 319-321, 326-328, 335 Applications 2-3, 6, 8, 10, 12, 20, 22, 32, 36, 40, 45, 54-66, 71-73, 80-85, 100-101, 113, 122-124, 127, CIDOC CRM 403, 405-408, 411, 415, 419 Cloud Computing 16, 221, 377-380, 382-391, 393-139, 141, 147, 149, 183, 195, 199, 211, 221, 241, 246-251, 254-255, 318-320, 326, 340-341, 377, 396, 400 Co-Creation 246, 248, 250, 257, 262-263, 272, 275, 379-380, 384-386, 388-389, 391, 402-403, 408-412, 414-415, 421-422, 430, 433 278, 283 AR Translation 71 Communal Lands 353 Archaeological Parks 174, 190 Computer Vision 57-58, 99, 109, 119 Conceptual Model 42, 59, 406, 415, 419 Archaeological Sites 60, 74, 77-78, 81, 84-86, 140-143, 148, 155-158, 162-164, 173-176, 183, 191-194, Conceptual Reference Model 406, 415, 419 208-211, 214, 216, 218-219, 221-222 Consumer Behavior 2, 8, 22, 29, 286 Archaeological Spaces 222 Creative Experiences 264, 272, 276-278, 283 Archeology 142-143, 148-149, 158, 175-176, 190 Creative Tourism 264-265, 267-269, 272-278, 283 Architecture (Missionary) 243 Creativity 45, 235, 272-275, 278, 283 Ask Vicente 321-322, 326-328, 335 Crosswalk 419 Augment Reality in Tourism Industry 71 Cultural Heritage 1-2, 5, 13, 21, 29, 43-44, 54-56, Augmented Reality 10, 15, 29, 37, 39, 41-42, 53-56, 59-60, 62-64, 71-73, 78-80, 85, 113, 127, 141, 58-66, 71-73, 79, 82, 85, 96-98, 100, 108, 111, 149-154, 158, 160, 162, 175-177, 184, 190, 192, 113, 119-128, 132-134, 139, 157-159, 164, 195, 197, 210-211, 220, 272, 285, 310, 313, 358, 361, 215-216, 221-224, 226, 240, 252, 256-257, 391 406-407, 419 Authentic Illumination 155 Cultural Heritage Management 176, 190

### Index

Cultural Open Data 406, 419
Cultural Resources 64, 79, 174-175, 182-184, 190, 310, 340, 342, 353, 404-405
Cultural Tourism 2-4, 10, 60, 71, 80, 124-125, 128, 130, 173-175, 177, 183-185, 190, 210, 243, 272-273, 275, 312-316, 326-328, 335

## D

Dark Site 356-358, 360, 362, 364, 366-369
Dark Tourism 354-360, 365-369, 376
Death Consumption 376
Decision Support 421-424, 426, 430, 437, 440-443, 448
Decision Support Platform (DSP) 448
Destination Image 183, 284-290, 292-293, 295, 297, 302, 310
Destination Marketing 55, 66, 71, 184, 288, 294, 306
Digital Representation of Data 93
Disability 43-44, 192-194, 208
Disabled Publics 30, 32, 43

## $\mathbf{E}$

EEA Grants 214
Electronic Word of Mouth 190, 264, 313
Emerging Technology 37, 60
Energy efficiency 420-427, 429-430, 437, 441-442
España Verde 284-285, 290-299, 301-306, 310
Event Management 248, 257, 262
e-Word of Mouth 313-315, 355
Experience co-creation 250, 257

## F

Fab Lab 53 Facebook 5, 32, 37, 185, 249-250, 254, 271, 284, 288, 292, 294-295, 304-306, 313, 319 Foodie 335

## G

Gamification 15, 21, 29, 37, 98-99, 101, 223
Gastronomic Recommendations 312, 314, 320, 322-328, 335
Gastronomic Tourism 313-314, 320-321, 327-328, 335
Geographic Information Systems 134, 141, 220, 338-341, 353
German Market 284-285, 294, 304-305

## H

HCI 98-99
Heritage 1-5, 10, 13-16, 21, 29, 31, 35, 39, 43-45, 54-56, 59-60, 62-66, 71-76, 78-86, 113, 124-125, 127, 129, 136, 141-142, 149-154, 158, 160, 162, 174-179, 182-185, 190-192, 194-197, 210-211, 214, 216, 219-220, 222, 224, 228-231, 233-234, 236-238, 241, 243, 272-273, 284-285, 291, 310, 313, 315, 335, 357-358, 361, 366, 405-407, 409-411, 419

Hardware and software optimization 257

Holocaust Tourism 376 Human Senses 43, 80, 85, 96-97, 100, 240 Human-Computer Interaction 42, 98, 119, 122, 248

## I

IaaS 378, 385, 400
ICT 1-3, 5, 10, 63, 124, 162, 177, 184-185, 195, 214, 245-249, 252, 255-257, 263-265, 267-272, 274-278, 359, 421-422, 426
Imaging 77-78
Information Systems 5-6, 16, 19, 29, 124, 134, 141, 220, 255, 338-343, 353, 378, 401
Intelligent software agents 248
Interaction 6, 8, 10, 13, 22, 31, 33-35, 37, 41-45, 57-58, 62, 64, 72-73, 80-86, 93, 98-100, 102, 119, 122, 125, 133, 148-149, 158-159, 177, 195, 199, 229, 238, 246, 248-249, 272, 275, 284-286, 288, 319, 356, 358, 365-367, 379, 389, 391, 400, 413
Interaction Design 98, 102, 119
Internet of Things (IoT) 20, 40, 378, 421-422, 448

## J

Jesuit Missions 231-234, 237, 240-241, 243

## K

Kinect 53, 79, 82

## L

Laser Scan Survey 227 Linked Data 402-403, 405, 409, 411, 419

### P $\mathbf{M}$ Machine Learning (ML) 421-422, 427, 435, 448 PaaS 385-386, 400 Maker Faire 38, 53 Participatory Approach 338-339, 341-342, 350, 353 Management 4-5, 8-9, 19, 21, 62, 79-80, 84, 99, 121, Peer to Peer (MIM) 335 124, 127, 132, 134-135, 141, 159, 173-176, Pilgrim 120-121, 128, 131, 136, 139 183-185, 190, 192, 197, 200, 219-220, 223, 226, Procedural Modeling 144-145, 148-154, 161-162, 164 245-248, 255, 257, 262, 276, 284, 286, 288, 294, PROSPELASIS 191-192, 195-197, 201-203, 205-206, 306, 318, 339-343, 348, 350, 353, 368, 378-380, 209-211 386-394, 396, 400-403, 406-408, 411, 421-422, 424, 426-430 Q Marker-based AR 98 QR Code 53, 250 Marketing Intelligence 12-13, 22 Mass Events 245-248, 250-251, 257, 262 R Metadata 79, 402, 404-406, 409-411, 414-415, 419 Metadata formats 404-405, 414-415 Recommendation Systems 3, 5-6, 10, 12-13, 22, 248 Milreu 215-219, 222, 224, 227 Religion 3-4, 355, 357-358, 360-361, 365-367, 369 Mixed Reality 56, 71, 123, 139, 158 Religious tourism 1, 3-5, 13-16, 21, 122, 128-129 Mobile 2, 6, 8-9, 13, 15, 21-22, 32, 36-37, 40-41, 44-Renewable Energy 421-426, 428-430, 440, 442-443, 45, 53, 55-58, 60-66, 71, 80, 83, 96-103, 108-111, 113-114, 121-124, 127, 132, 139-140, 143, 158-RFID 246-247, 250, 255, 263 161, 163, 185, 195, 200, 215-216, 221, 223, 227, Rural Development 338, 341 234-235, 239, 243, 247-248, 250-251, 254-255, Rural Tourism 339, 350, 353 263, 312, 314, 316, 318-321, 324, 326-328, 335, 378, 391, 393-394, 401-402, 410-411, 433-434 S Mobile Augmented Reality 58, 60, 64-66, 71, 139, 158-159 SaaS 377, 385-386, 400 Mobile Devices 2, 6, 8-9, 13, 22, 40, 53, 55, 57-58, Santiago Way 122, 135-136, 139 61-64, 71, 80, 83, 99, 113, 121, 139-140, 158, Semantic Portal 409, 419 215-216, 221, 223, 247, 255, 411 Semantic web 78-79, 402-403, 409, 414 Mobile Instant Messaging 312, 314, 316, 318-320, Semantics 78, 85-86, 152, 405 324, 326-328, 335 Shape 43, 45, 72-74, 77-78, 83-85, 128, 141, 150-153, Mobile Movement 250, 263 156, 201, 227, 229, 269-270, 274, 342, 355, 369 Multi-Dimensional (mD) Data 73, 93 SLA 380, 384 Multimodal Technology 81, 93 Smart Destination 14, 255, 263 Multi-Sensorial Display 100, 119 Smart Grids 421-422, 429 Multi-touch 32-35, 45, 103 Smart phones 263 Museum 30-46, 60-62, 81-82, 97-106, 108, 111, Smart Tourism 55, 245-247, 255, 257, 376, 378, 389, 113-114, 125, 144, 161, 194-195, 240, 291, 403, 391, 400 405-407, 409, 411, 413, 415 Social Media 15, 20-21, 29, 31-33, 41, 45, 53, 65, 177, Mycological Tourism 338-344, 346, 348-350, 353 184-185, 246, 248-250, 270, 274, 276, 284-285, 288, 292, 294, 304-306, 310, 313-314, 317, 320, N 355-356 Spain Today 285, 299, 310 Natural Heritage 174, 284, 310 Specialized Consumption 266-267, 283 Nearly Zero Energy Building (nZEB) 448 Spirituality 354-355, 357-358, 360-361, 365-367 New business models 29 Supervisory Control and Data Acquisition (SCADA) New Cultural Mediators 264-265, 268, 271-272, 283 448 NFC 40, 53, 250

## $\mathbf{T}$

Tactile Models 201-202, 214 Tangible Assets 72-73, 81-82, 85, 93, 175 Tangible User Interfaces 42, 44-45 Technological Mediated Experiences 283 Technology 2, 5, 7, 13-16, 21-22, 30-32, 36-38, 40-46, 53-63, 65-66, 74, 79, 81-83, 93, 100-101, 114, 119, 121-127, 143, 157, 191, 195, 197, 199, 220-221, 224, 228-231, 237-241, 243, 246-248, 250-251, 255, 258, 264, 268-273, 275-276, 283, 307, 314, 320, 338, 343, 351, 359, 369, 377, 379-381, 383-384, 389-392, 395-396, 403, 422 Territorial Specialization 353 The AR Tourism Guide 62, 71 The AR Tourism Navigation 60, 71 Three-Dimensional (3D) Data 93 Tourism 1-10, 12-16, 19-22, 29, 38, 54-56, 58-60, 62-66, 71-73, 75, 79-80, 83-85, 97, 101, 119, 121-122, 124-125, 127-132, 134, 173-179, 183-185, 190, 193, 198, 208, 210, 215-216, 222, 224, 228-229, 236, 239-241, 243, 245-247, 251, 255-257, 263-269, 271-278, 283-286, 288-293, 295, 299, 302, 304-306, 310, 312-318, 321-322, 326-328, 335, 338-344, 346, 348-350, 353-360, 365-369, 376-378, 389-392, 394-396, 400-401, 410, 414, 421, 423, 425 Tourism AR Games 64, 71 Tourism Experience 1, 3, 13-14, 16, 29, 56, 73, 101, 247, 256, 275, 342, 357 Tourism Sector 1-2, 175, 239, 247, 268, 275, 288, 314, 321, 377-378, 389-392, 394-396, 423 Tourism Supply 285, 290, 292, 299, 305-306, 310 Traditional Ecological Knowledge 338-339, 343, 345, 350, 353 TripAdvisor 5, 313, 317-318, 324, 354, 356, 358, 360-362, 365-366, 369, 376 Trunyan Cemetery 354-355, 357-362, 364-369, 376

## U

UGC - User Generated Content 335 Unesco 3, 35, 59, 124, 129, 174, 191, 197, 208, 210, 214, 229, 233-234, 241, 243, 291 Unesco World Heritage 35, 59, 210, 214, 241 UNESCO World Heritage List 210, 214 Universal Design 99 Unmanned Aerial Vehicle 215-216, 227 User Interfaces 42, 44-45, 97-98, 102, 119

## V

Virtual Reality 38-40, 45, 55-56, 59, 71-72, 79-80, 100, 121, 124, 141, 157, 163, 215, 222-224, 227, 326 Virtual Reconstruction 73-74, 76, 84, 93, 127, 140-142, 144, 147, 152-156, 158, 162-163, 216, 219, 221-224 Virtual Reconstruction Process 141-142, 162 Visualization 35-36, 42, 57, 63, 65, 73, 75, 77, 97, 102, 140-143, 155, 157-160, 163, 200-201, 216,

## W

220, 224, 433

Wearable Technology 41
Web 2.0 184, 221, 313-314, 316-318, 320, 326-328, 335
Web 2.0 (Web Social) 335
Wine Tourism 291, 313, 335
World Heritage 10, 35, 59, 124, 129, 174, 177, 191, 210, 214, 229, 234, 241, 243, 291
Worth of Mouth 190